

Transport coefficients and low energy excitations of a strongly interacting holographic fluid

Poovuttikul, N.

Citation

Poovuttikul, N. (2017, November 16). Transport coefficients and low energy excitations of a strongly interacting holographic fluid. Casimir PhD Series. Retrieved from https://hdl.handle.net/1887/57561

Version:	Not Applicable (or Unknown)		
License:	<u>Licence agreement concerning inclusion of doctoral thesis in the</u> <u>Institutional Repository of the University of Leiden</u>		
Downloaded from:	https://hdl.handle.net/1887/57561		

Note: To cite this publication please use the final published version (if applicable).

Cover Page



Universiteit Leiden



The handle <u>http://hdl.handle.net/1887/57561</u> holds various files of this Leiden University dissertation

Author: Poovuttikul, N. Title: Transport coefficients and low energy excitations of a strongly interacting holographic fluid Date: 2017-11-16

Transport coefficients and low energy excitations of a strongly interacting holographic fluid

Proefschrift

ter verkrijging van de graad van Doctor aan de Universiteit Leiden, op gezag van Rector Magnificus Prof. Mr. C. J. J. M. Stolker, volgens besluit van het College voor Promoties te verdedigen op dinsdag 16 November 2017 te klokke 10:00 uur

DOOR

Napat Poovuttikul

geboren te Bangkok (Thailand) in 1990

Promotiecommissie

Promotor: Prof. dr. J. Zaanen Overige leden: dr. K. Landsteiner (IFT-UAM/CSIS, Spain) dr. A. Parnachev (Trinity College Dublin, Ireland) Prof. dr. S. Vandoren (IFT, Utrecht University) Prof. dr. E. R Eliel Prof. dr. K. E. Schalm Prof. dr. V. Vitelli

Casimir PhD series 2017-35 ISBN 978-90-8593-319-9

The work describe in this thesis has been carried out at the Lorentz Institute, Leiden University. The author was supported by Development and Promotion of Science and Technology (DPST) Scholarship from the Thai government.

The cover shows the private and public transport phenomena of electricity, heat, humans and other pollutants. Incidentally, this photo was taken at Wong-sawang junction (close to the author's house in Bangkok), infamous for its traffic congestion, a few seconds before a waterfall-like fluid (a.k.a. a tropical rain) starts to pour down and almost destroys the author's camera.

Contents

42

	1	
t is QFT?	1	
v duality and effective theory	4	
hunt for universality		
ard hydrodynamics	9	
of hydrodynamics and gauge/gravity duality	12	
etry, conserved current		
nd fields	13	
e neutral relativistic fluid	15	
ing translational symmetry	20	
ucing anomalous $U(1)$ current	22	
alised global symmetry	24	
2-point correlation functions and Kubo formulae		
proach to holographic duality	30	
ring global symmetry	31	
raphic thermal 1-point and 2-point function	35	
raphic RG flow	36	
nembrane paradigm"	38	
r derivative holography	40	
2	er derivative holography	

without translational symmetry

	3.1	Motiva	ation	42
	3.2	Effecti	ve theory for systems with broken translational symmetry	46
		3.2.1	Constructing the constitutive relation	47
		3.2.2	Kubo's formula for η^{\star}	52
	3.3	Holog	raphic computation	54
		3.3.1	Action and Thermodynamics	55
		3.3.2	Coherent regime and constitutive relation from fluid/-	
			gravity correspondence	57
		3.3.3	Fluctuations and violation of the viscosity bound at lead-	
			ing order	59
		3.3.4	Numerical results and beyond the leading order \ldots .	62
	3.4	Discus	sions and outlook	64
	3.5	Appen	ldices	67
		3.5.1	Scalars, vectors and tensors from basic structures	67
4	Univ	versalii	ty of anomalous conductivities in theories with higher-	
	deri	vative	holographic duals	69
	4.1	More I	background materials and motivations	69
	4.2 The holographic setup			
		The ho	olographic setup	78
	4.3	The ho Proof	olographic setup	78 83
	4.3	The ho Proof 4.3.1	olographic setup	78 83 84
	4.3	The ho Proof o 4.3.1 4.3.2	olographic setup	78 83 84 88
	4.34.4	The ho Proof o 4.3.1 4.3.2 Examp	olographic setup	78 83 84 88 93
	4.3 4.4	The ho Proof 6 4.3.1 4.3.2 Examp 4.4.1	olographic setup	78 83 84 88 93 94
	4.3 4.4	The ho Proof o 4.3.1 4.3.2 Examp 4.4.1 4.4.2	olographic setup	78 83 84 88 93 94 95
	4.3	The ho Proof 6 4.3.1 4.3.2 Examp 4.4.1 4.4.2 4.4.3	olographic setup	78 83 84 88 93 94 95
	4.3	The ho Proof 6 4.3.1 4.3.2 Examp 4.4.1 4.4.2 4.4.3	olographic setup	 78 83 84 88 93 94 95 97
	4.3	The ho Proof 6 4.3.1 4.3.2 Examp 4.4.1 4.4.2 4.4.3 4.4.4	olographic setup	78 83 84 88 93 94 95 97
	4.34.44.5	The ho Proof 6 4.3.1 4.3.2 Examp 4.4.1 4.4.2 4.4.3 4.4.4 Discuss	olographic setup	78 83 84 88 93 94 95 97 100 101
	4.34.44.54.6	The ho Proof 6 4.3.1 4.3.2 Examp 4.4.1 4.4.2 4.4.3 4.4.4 Discuss Appen	olographic setup	78 83 84 93 94 95 97 100 101

5	Magnetohydrodynamic waves in a strongly interacting holographic			
	plas	ma		104
	5.1	Introd	uction	104
	5.2	coupled to electromagnetic interactions	110	
		5.2.1	Quantum electrodynamics	110
		5.2.2	Strongly interacting holographic matter coupled to dy-	
			namical electromagnetism	112
	5.3 Holographic analysis: equation of state and transport co			s 118
		5.3.1	Holographic action and the magnetic brane	119
		5.3.2	Holographic renormalisation and the bulk/boundary dic-	
			tionary	121
		5.3.3	Equation of state	127
		5.3.4	Transport coefficients	131
	5.4	Magne	etohydrodynamic waves in a strongly coupled plasma $$.	137
		5.4.1	Speeds and attenuations of MHD waves	143
		5.4.2	MHD modes on a complex frequency plane	146
		5.4.3	Electric charge dependence	148
	5.5 Discussion		152	
	5.6	Appen	ndices	155
 5.6.1 Kubo formulae for first-order transport coefficients 5.7 Further details regarding the derivation of the transport co cients		Kubo formulae for first-order transport coefficients	155	
		er details regarding the derivation of the transport coeffi-		
			157	
	5.8	Disper	sion relations of magnetosonic waves	163
6	Con	clusior	n and outlook	167
Sai	menv	atting		173
Bil	oliogı	raphy		175
Cu	rricu	lum V i	itæ	200
Lis	t of I	Publica	tions	202

Acknowledgement

203

On the other hand, one does not yet have a mathematically complete example of a quantum gauge theory in four-dimensional space-time, nor even a precise definition of quantum gauge theory in four dimensions. Will this change in the 21st century? We hope so!

A. Jaffe and E. Witten on Yang-Mills existence and mass gap [1]

Introduction

1.1 Preface : What is QFT?

Quantum field theory (QFT) has been an active area of research for several decades by now. By all means, it is one of the greatest accomplishments of theoretical physics from the last century, with applications ranging from high energy physics, such as the standard model, to low energy physics, such as our understanding of metals and semiconductors. We are able to predict the magnetic dipole moment of the electron which agrees with the experimental result to eight decimal places: an unprecedented accuracy in the entire history of science! We are able to predict the existence of anti-particles, W^{\pm} , Z bosons, gluons and quarks in the standard model of particle physics. We have a theory of electrons in solids, which enables us to build computers, smartphones, NMR machines, superconductors and many many more applications.

But perhaps we only see the tip of the iceberg. Our achievements from con-

ventional QFT rely on single-particle classical physics intuition: we start from a microscopic Lagrangian description then treat the interaction and the quantum fluctuation as a perturbation. However, we seem to fail miserably when trying to compute physical quantities from the usual QFT formalism when the interactions are strong. One of the most glaring examples is the confinement phenomenon in quantum chromodynamics (QCD), which is responsible for most of the mass of the known matter in the universe. It is known that the interaction strength in QCD (describing physics of quarks and gluons in nuclear matter) is infinitely large at low energy and this dynamically generates a mass gap, resulting in bound states of quarks and gluons, such as protons and neutrons. We know that this is true thanks to the results from lattice simulations, but more evidently, because most of our mass came from protons and neutrons. Still¹, there is no analytic proof why they should form such bound states!



Figure 1.1. A visualisation of a proton in the "vacuum" filled with gluon field. The gluonic field in the proton interior is pushed away by quarks resulting in the flux tube structure confining the quarks together (figure from [2])

Strong interaction are also believed to play a crucial role in materials with high critical temperature superconductivity. There are many intriguing aspects to these materials but I will focus on the strange metallic phase at temperature

¹See e.g. a Millennium Prize problem [1] by Clay Mathematics Institute involving the existence of this mass gap.

above the, unusually high, superconductor critical temperature². As a name suggests, a strange metal has certain properties that are different from a typical metal. For example, the resistivity of a strange metal grows linearly in contrast to the quadratic growth found in a normal metal. This may not seems like a big deal, but the T^2 dependence is extremely robust. It was shown, from renormlisation group analysis, that any small perturbation around normal metal cannot change this property [5, 6] (see also a review [7]). Thus, one may argue that the "perturbation" has to be very strong in order to obtain the linear resistivity and, again, we enter the regime of strongly interacting quantum field theory. To make matters even worse, the brute-force numerical simulation is categorised as NP-hard in computational complexity [8], thus making the classical simulation of strongly coupled fermions in thermodynamic limits unachievable³.

However, it an overstatement to say that we cannot compute anything when the interaction is strong. Numbers of non-perturbative techniques and many toy models where computations are doable even at strong interaction have been developed over the past few years. These non-perturbative results might make one doubt whether the Lagrangian-based formulation of QFT is the best way to understand quantum systems. For example, there are a number of nonperturbative methods which do not rely on microscopic Lagrangian such as conformal bootstrap [13, 14], integrability [15], modern analysis of scattering amplitude [16–18]. Instead they rely heavily on the symmetries of the systems. Many non-perturbative results are also obtained by utilising the property called *duality*. Duality is the phenomenon where by two or more Lagrangians which describe the same physics and, sometimes, enables us to map a difficult problem of strongly interacting QFT to a very simple one. Since this is the most important aspect of this thesis, let me be a little more precise. The strongest form of

 $^{^{2}}$ See [3] and chapter 2-3 of [4], for a brief history and introduction to high temperature superconductor.

³This problem is known as "the sign problem" which, sadly, is largely ignored by most introductions to quantum field theory. However, it is one of the biggest enemies when one tries to directly simulate strongly correlated fermion systems, both in condensed matter and lattice QCD (see e.g. [9, 10]). See also [11] for a discussion of the sign problem in a path integral formalism and the references therein. Solving this problem in polynomial time is equivalent to solving, yet another, Millennium Prize problem by Clay Mathematics Institute [12].

the duality is the following:

If a theory A is dual to a theory B, then for every physical quantity in A there are corresponding physical quantities in B. The theories A and B may have different microscopic constituents or gauge groups or even live in a different dimension.

If A and B describes the same physics, one may ask a philosophical question: which Lagrangian describe the "true nature" of quantum theory. Are they both physical? Is this just a mathematical trick? Or should we think of theory B as an emergent phenomenon in a certain limit of theory A? At this point, we simply do not know.

So what is QFT? The facts that a conventional Lagrangian formalism seems to hit a brick wall, most non-perturbative results rely so little on the microscopic Lagrangian and, sometimes, more than one Lagrangian describe the same physics, all seem to indicate that we should rethink what we know about it. How should QFT be reformulated? I am afraid I do not know. There seems to be a lot of things we need to understand before answering such a question. *The goal of this thesis* is to make a few small steps toward, hopefully, the answer by exploring a peculiar duality between strongly interacting quantum field theory and the theory of general relativity, known as the gauge/gravity duality.

1.2 Gauge/gravity duality and effective theory

The origin of this duality dates back to the study of black hole entropy. One very peculiar property of black holes is that their entropy is proportional to the area of a surface called black hole horizon, not the enclosed volume. This and many other pieces of evidence prompted 't Hooft and Susskind to conjecture that a black hole can be described by a quantum field theory in one lower dimension [19, 20]. In this picture, the statistical physics interpretation of black hole entropy is nothing but the number of microstates of the corresponding quantum field theory. This statement can be confirmed in the string theory setup by

counting the microstates of extended objects called D-brane[21]. Shortly after, it was argued by Maldacena [22] that the large class of quantum field theories constructed in similar manner not only capture the entropy of the black hole but also the dynamics of the spacetime itself! The strongest statement of this duality is:

For a quantum field theory with a holographic dual, there is a theory of (quantum) gravity in higher dimensions in which every physical quantity in the field theory can be mapped into a physical quantity of the dual gravity.

The fact that a certain spacetime in a theory of (super)gravity seems to emerge from quantum field theory is, in itself, a very fascinating phenomenon. In addition, there is a very interesting usage of this duality. It turns out that the dual gravity theory becomes classical when the coupling of the QFT becomes infinitely large. Thus, with this duality, one might hope to use it to compute quantities in the, conventionally, unaccessible regime of quantum field theory.

Despite such a promising aspect, one has to be careful about the applicability of this duality. With this in mind, let me illustrate this caveat with some technical details. In the prototypical example of gauge/gravity duality, the aforementioned quantum field theory is a well-known maximally supersymmetric quantum field theory in 3+1 spacetime dimensions, called $\mathcal{N} = 4$ supersymmetric Yang-Mills theory with SU(N) gauge group. It consist of a non-Abelian SU(N) gauge field, four species of fermions and six scalar fields. All of these transform under the adjoint representation of SU(N). Roughly speaking, this theory is a matrix theory where all the fields are $N \times N$ matrices. The coupling between theses fields is extremely constrained by the supersymmetry, leaving only one free parameter in the Lagrangian: the Yang-Mills coupling constant $g_{\rm YM}$.⁴ At first glance, studying this theory or its deformations is very unlikely

⁴In terms of these parameters, the dual gravity theory only becomes tractable when $g_{\text{YM}}^2 N \gg$ 1 and $N \gg 1$. Otherwise, one needs to include quantum gravity effects in string theory. This issue will be discussed again in chapter 2.

to give us any new insight on the microscopic parameters of a realistic theory (such as hopping parameters, spin orbit coupling etc).

However, it is certainly possible for theories with different microscopic structures to be macroscopically equivalent to one another. This concept, known as *universality*, where many theories with different microscopic constituents become macroscopically indistinguishable, is well-known in the study of critical phenomena. This is the situation we are hoping for: *the infrared physics of the system we are interested in is the same as that of a theory with holographic dual.* If this indeed happens, we should be asking for the macroscopic information rather than the microscopic.



Figure 1.2. A schematic renormalisation group flow from small (UV) to large (IR) length scale where the QFT with holographic dual is in the same universality class as the system we are interested.

Focusing on the macroscopic quantities, such as thermodynamic variables, seems to be a very natural starting point. It is widely known that in strongly interacting quantum systems, thermodynamic quantities exhibit (non-mean field) scaling relations, which are very difficult to achieve using conventional methods. In the contrary, thermodynamic quantities of field theories with holographic dual are almost guaranteed to have such scaling relations. Interestingly, some of these relations are the same as those observed in strongly correlated systems.

Beyond equilibrium analysis, the gauge/gravity duality can be used to study transport phenomena and real-time dynamics of strongly interacting theories at finite temperature. Instead of having a quasi-particle description, a large class of theories with gravity dual exhibit fluid-like excitations. Intuitively, this is due to the fact that a quantum particle lose its individuality because the interaction with its neighbours is infinitely strong. This situation is well-summarised in Fig. 1.3. Such hydrodynamics features have also been observed in the study of the Nernst effect in quantum critical regions [23], graphene near the charge neutrality point [24, 25] and PdCoO₂ [26].



Figure 1.3. (LEFT) The ballistic transport where each quasi-particle barely interacted with each other. (RIGHT) The hydrodynamic transport where quantum particles collide with each other, losing their individuality and becoming a fluid-like entity (figure from [27]).

The fact that hydrodynamics is able to capture the dynamics of such systems may not be very surprising. After all, hydrodynamics is an almost universal effective theory that describes systems with excitation wavelength, λ_{excite} much larger than the mean free path, ℓ_{mfp} , of the constitute particles in the system (which is known to be vanishing as the interaction increases). What sets the strongly interacting quantum fluid apart from a classical fluid, however, is its viscosities. It was found that, in classes of strongly interacting quantum systems, the viscosity is governed by the ratio of Planck's constant and Boltzmann's constant \hbar/k_B where the appearance of \hbar signifies its quantum nature [28] (see also [4, 29] for more stories from the condensed matter viewpoints). In this aspect, gauge/gravity duality not only confirms this observation but also gives the value of the viscosity explicitly. By examining a large class of models with gravity duals, Kovtun, Son and Starinets (KSS) [30] found an interesting pattern and postulated that the ratio of the shear viscosity η and the entropy density s is bounded from below i.e.

$$\frac{\eta}{s} \ge \frac{\hbar}{4\pi k_B},\tag{1.1}$$

where the inequality is saturated when the coupling is infinitely large. Remarkably, this ratio can also be extracted from experimental data in known strongly interacting quantum systems, see Fig. 1.4. Interestingly, the viscosity/entropy density ratio gets closer the value $\hbar/4\pi k_B$ as the interaction becomes stronger near the critical point and so far the bound has yet to be violated experimentally. This also strongly suggests that many strongly interacting quantum field theories may belong to the same universality class with the universal diffusion constant $D := \eta/sT = \hbar/4\pi k_B T$ at infinite coupling.



Figure 1.4. (LEFT) The interpolation between value of η/s in $\mathcal{N} = 4$ supersymmetric Yang-Mills between known results at weak coupling $g_{YM}^2 N \ll 1$ and strong coupling $g_{YM}^2 N \gg 1$. (RIGHT) The experimental data in various materials near the critical point compared to the KSS bound (1.1) (figures from [30, 31]).

I would like to emphasise that, although thinking about quantum theory from hydrodynamic point of view may not give us a straightforward knowledge of microscopic information, it nevertheless allows us to make a link between dissipative nature and quantum nature of many body quantum systems. From this perspective, gauge/gravity duality is a playground where computations involving these aspects can be performed explicitly. Moreover, the lower bound on shear viscosity/entropy density can, in principle, be experimentally tested. A result such as the KSS bound inspired searches for similar universal bounds e.g.[32–42] and hopefully these will help us better understand the nature of quantum field theory at strong coupling.

1.3 This thesis : A hunt for universality beyond standard hydrodynamics

One reason why hydrodynamics is such a universal effective theory is its underlying principle: global symmetry. From a more modern point of view, hydrodynamics is the low energy dynamics of the Noether currents whose time evolutions governed by conservation laws. In the long wavelength limit, the microscopic details are washed out and the effective theory is fully described by the equations of state and a few parameters in the gradient expansion (such as shear and bulk viscosity). For example the relativistic Navier-Stokes equation can be written as

$$\partial_{\mu}T^{\mu\nu} = 0, \qquad T^{\mu\nu} = (\mathcal{E} + \mathcal{P})u^{\mu}u^{\nu} + \mathcal{P}\eta^{\mu\nu} + \sum_{\text{All }\mathcal{O}(\partial^{1}) \text{ terms}} \alpha_{i} \mathbb{T}^{\mu\nu}_{(i)}, \quad (1.2)$$

where $T^{\mu\nu}$ is the stress-energy tensor, $\mathbb{T}_{(i)}$ are all possible symmetric traceless rank–2 tensors with one derivative. The conservation of $T^{\mu\nu}$ is linked to the spacetime translational symmetry of the system. The system of equations (1.2) can be extended to include dynamics of additional conserved currents if the system also possesses more global symmetries. The coefficients α_i are called *transport coefficients*. The behaviour of these transport coefficients in a strongly interacting theories is the main focus of this thesis, with an ultimate goal to explore universal relations analogous to (1.1).

Chapter 3,4,5 of this thesis are based on my original works with Piyabut Burikham [43] and Sašo Grozdanov [44, 45] where we study the interplay of additional (or broken) global symmetries, lower energy excitations and transport coefficients in strongly interacting quantum field theories with holographic dual. Chapter 2 contains a brief introduction and background material to relativistic hydrodynamics and implementations of gauge/gravity duality in this context. In the rest of this thesis, all the results are expressed in units where $\hbar = c = 1$. The outline of each chapter is the following.

In chapter 3, we construct a consistent hydrodynamic model where translational symmetry is broken. This is achieved by coupling the strongly interacting matter to the external spatial dependent scalar fields. We carefully define what is the shear viscosity in this system and study the validity of the KSS bound (1.1) in a hope to apply this bound to condensed matter systems where translational symmetry is broken by a lattice or disorder. We find that the bound is violated when we include the first order correction from the scalar fields. We are also able to identify how this violation occurs from the hydrodynamic point of view.

In chapter 4, the system now contains two additional global U(1) symmetries and is subjected to a small external non-dynamical magnetic field. One Noether current is conserved but the other is broken by a quantum anomaly. This gives rise to a new transport channel called*anomalous transport*. Although quantum anomaly is famous for their non-renormalisability (or 1-loop exactness in the diagrammatic approach), it is not entirely clear why these new conductivities should inherit the same property. We find that the information regarding these conductivities is encoded on the black hole horizon in the dual gravity theory. With these horizon formulae, we show that, for all theories with a holographic dual at finite temperature, the form of these anomalous conductivities is fixed by quantum anomaly coefficients, unaffected by the value of the coupling constant and details of the renormalisation group flow.

In chapter 5, the matter no longer breaks any existing global symmetry but is now coupled to a *dynamical* external gauge field, a setup one usually finds in plasma physics. A standard treatment of this system is called magnetohydrodynamics (MHD), which is a very successful framework in the plasma physics community. The problem is that conventional MHD is constructed with an underlying assumption that the equation of state of the matter is independent of the magnetic field. This assumptions makes perfect sense in a normal earthly circumstances where the temperature is very high and the magnetic field is very small. However, there are examples in nature where this is not true. We use a novel concept of generalised global symmetry, put forth by Gaiotto, Kapustin, Seiberg and Willet [46], to identify the true global symmetry of this system. Together with the new formalism of hydrodynamics with generalised global symmetry, recently constructed by Grozdanov, Hofman and Iqbal [47], we implement this new concept in gauge/gravity duality. As a result, for the first time, we are able to investigate low energy dynamics of this system at arbitrary strength of magnetic field.

Lastly, in chapter 6, I summarise the results in this thesis, its contribution to our understanding of quantum field theory at strong coupling and possible future research directions. You need a different way of looking at them than starting from single particle descriptions. You don't try to explain the ocean in terms of individual water molecules.

Sean Hartnoll, in Quanta Magazine

2

Lightning review of hydrodynamics and gauge/gravity duality

In this chapter, I briefly review some basic concepts of hydrodynamics and gauge/gravity duality. I will start by constructing the simplest relativistic hydrodynamics from global symmetry and gradually introduce more elements which are essential for chapters 3,4 and 5. The following section review more technical details on how to compute 2-point correlation function and extract transport coefficients from hydrodynamic equations. Then I present a very short introduction to the holographic duality with the minimum technical details possible. I will also restrict all of the discussion to relativistic hydrodynamics, ignoring non-relativistic model.

None of the material presented in this chapter is new. Hydrodynamics is, strictly speaking, a millennia-old subject. Many aspects of relativistic hydrodynamics can be found in a classic textbook by Landau and Lifshitz [48] or, for a more modern notation and applications, by Rezzolla and Zanotti [49]. The discussion in section 2.1-2.2 is inspired by [50, 51] complemented by the discussion about global symmetry in [46]. The procedures presented in section 2.2 are widely used in the holography community to compute correlation function and the review of the method itself (without reference to holography) is neatly summarised in [51]. The basic principle of gauge/gravity duality has been very well documented over the last few decades from both string theory [52, 53] and applications to condensed matter or quark-gluon plasma perspectives(see e.g. reviews, lecture notes [54–57]) and books [4, 58–60]). Therefore I will spare the reader the details and only cover the portion of the story that is most relevant to this thesis.

2.1 Global symmetry, conserved current and background fields

As mentioned in the previous chapter, one way to think about hydrodynamics as the effective theory describing the low-energy dynamics of many-body systems where the excitation wave length is much longer than the mean free path. Thess low-energy dynamics are governed only by the conservation laws of the system. Thus, one may say that hydrodynamics owes its existence to continuous global symmetries, which are related to conserved currents by the Noether theorem.

Usually, we derive the Noether current from a certain global symmetry transformation of fields in the microscopic theory. However, there is another way to look at it. A more Lagrangian-free way to access a global symmetry is to couple the system to a non-dynamical background field. For example, to introduce the chemical potential in the grand canonical ensemble, we deform the Hamiltonian in the following way

$$H \to H - \mu Q$$
, (2.1)

where Q is the number of particles. Now, we know that particle number is the

Noether charge of a conserved U(1). In relativistic notation, this deformation can instead be written as

$$H \to H - \int d^{d-1}x J^{\mu}A_{\mu}, \quad \text{where} \quad \partial_{\mu}J^{\mu} = 0, \quad (2.2)$$

where the U(1) charge is obtained by integrating the current J^{μ} over the spatial volume $Q = \int d^{d-1}S_{\mu} J^{\mu}$. The vector field A_{μ} plays the role of the non-dynamical *background field* we discussed earlier. This background field can also transform in the same way as the U(1) gauge field: $A_{\mu} \rightarrow A_{\mu} + \partial_{\mu}\Lambda$. Thus we should think about this background field as a *non-dynamical background gauge field*.

Interestingly, the background gauge field with the gauge group \mathfrak{G} can be used to singled out the Noether current associated to the global symmetry group \mathfrak{G} . This can be seen due to the fact that the minimum coupling between the matter and the background gauge field, $\int d^d x J^{\mu} A_{\mu}$, will not be invariant under the gauge transformation, unless J^{μ} is a conserved current.

The global symmetry also plays a crucial role in quantum physics. The state of a quantum system can be characterised by the eigenvalues of the conserved charge operators. In the study of the hydrodynamic limit of quantum systems, the quantity we are interested in is therefore the expectation of the operator \hat{J}^{μ} acting on the thermal state $\langle \hat{J}^{\mu} \rangle_{\text{thermal}}$. To access these quantities without directly specifying the Lagrangian, we put the theory on a curved manifold \mathcal{M} with metric $g_{\mu\nu}$ and, if the theory possesses the U(1) global symmetry, we couple it to the background U(1) gauge field A_{μ} . The background metric is introduced to capture the stress-energy tensor $\langle T^{\mu\nu} \rangle_{\text{thermal}}$. With these ingredients, we can write down the partition function, which has the background fields as its arguments. For example, the partition function of the relativistic charged fluid in the presence of a background metric $g_{\mu\nu}$ and gauge field A_{μ} can be schematically written as

$$Z[g_{\mu\nu}, A_{\mu}] = \left\langle \exp\left[i\int d^d x \sqrt{-g}\left(\frac{1}{2}T^{\mu\nu}g_{\mu\nu} + J^{\mu}A_{\mu}\right)\right]\right\rangle$$
(2.3)

The background fields can now be used as a source for the conserved currents.

The variational derivative of the partition function with respect to the background fields gives

$$\langle T^{\mu\nu} \rangle = \frac{-2i}{\sqrt{-g}} \frac{\delta \log Z}{\delta g_{\mu\nu}}, \qquad \langle J^{\mu} \rangle = \frac{-i}{\sqrt{-g}} \frac{\delta \log Z}{\delta A_{\mu}}$$
(2.4)

The next step is to to express the conserved currents in the long wavelength limit. This can be done by gradient expanding the conserved currents and writing down all possible terms allowed by symmetry at a given order. This is not yet a physical hydrodynamics as the theory has to be augmented by additional physical requirements such as the extensivity condition and the positivity of the local entropy production near equilibrium.¹

To make things more explicit, I first illustrate how this procedure works in a typical charge neutral fluid in section 2.1.1. The relevant global symmetries of this section are only spacetime translational symmetries (equivalently, $T^{\mu\nu}$ is the only conserved current). Then I will briefly review the hydrodynamic constructions relevant to the work of chapter 3-5, where some global symmetry groups are added or broken.

2.1.1 Charge neutral relativistic fluid

First of all, one needs to express $T^{\mu\nu}(x)$ in terms of the local macroscopic variables in order to solve the conservation equation

$$\nabla_{\nu}T^{\mu\nu} = 0 \tag{2.5}$$

which is a requirement that the system is invariant under diffeomorphism. To do this, we split the system into small pieces and assume that each piece, called fluid elements, which occupy an infinitesimal volume at position x^{μ} . This allows one to define local thermodynamics variables: the temperature T(x), the entropy s(x), the energy density $\varepsilon(x)$ and the pressure p(x), which are small variations of their global equilibrium values. Each fluid element is also allowed

¹Alternatively, one can also attempt to gradient expand the effective action. It turns out that this approach gives us more insight on the emergent symmetry of the system once dissipative effects are properly taken into account properly. We will come back to this approach in chapter 6.

to move around with relativistic velocity $u^{\mu}(x)$, which satisfied the condition $u^{\mu}u_{\mu} = -1$ (see Fig 2.1).



Figure 2.1. An illustration of a fluid living on the manifold \mathcal{M} with metric $g_{\mu\nu}$. The red box denotes a fluid element at point x^{μ} which moves with a velocity u^{μ} .

We are now ready to construct the stress-energy tensor. Since the stressenergy tensor is a rank-2 symmetric tensor, the possible 0th order in derivative expansion is

$$T^{\mu\nu} = \mathcal{E}u^{\mu}u^{\nu} + \mathcal{P}\Delta^{\mu\nu} + \mathcal{O}(\partial^{1}), \qquad \Delta^{\mu\nu} = g^{\mu\nu} + u^{\mu}u^{\nu}.$$
(2.6)

where $\Delta^{\mu\nu}$ is acting as a projector to a d-1 dimensional plane perpendicular to u^{μ} i.e. $u_{\mu}\Delta^{\mu\nu} = 0$ and $\Delta^{\mu\nu}g_{\mu\nu} = d-1$. The reader might notice that this form of the stress-energy tensor is similar to the one describing an ideal fluid if $\mathcal{E} = \varepsilon$ and $\mathcal{P} = p$. However, at this level, \mathcal{E} and \mathcal{P} are just arbitrary functions of thermodynamics variables. The expression of the conserved currents in term of these variables is called *the constitutive relation*.

To rigorously give a physical meaning to these two coefficients, we will follow a beautiful analysis of the equilibrium partition function in the presence of a background fields, introduced in [61, 62]. In this setup, the finite temperature can be incorporated by putting the system in the cylinder $\tilde{\mathcal{M}}_{d-1} \times S^1$ (or equivalently by performing the Wick rotation which transforms the original manifold \mathcal{M} to $\tilde{\mathcal{M}}_{d-1} \times S^1$), where we denote the radius of S^1 as L_0 . The fluid velocity u^{μ} in equilibrium is proportional to the Killing vector ξ^{μ} of the system². In this language, the temperature and fluid velocity can be expressed

²This is a more refined version of saying that if $\xi^{\mu} = (1, 0, ..., 0)$, then $\xi^{\mu} \partial_{\mu}(\text{Lagrangian}) = \partial_t(\text{Lagrangian}) = 0$

as

$$T = \frac{1}{L_0 \sqrt{-\xi^{\mu} \xi_{\mu}}}, \qquad u^{\mu} = \frac{\xi^{\mu}}{\sqrt{-\xi^{\lambda} \xi_{\lambda}}}.$$
(2.7)

In the equilibrium configuration, the free energy can be expressible in terms of these two quantities. In the Ginzburg-Landau style, the only term at 0th order in derivative expansion is

$$F = -\log Z = \int d^d x \sqrt{-g} \, p(T) \tag{2.8}$$

where F is the Gibbs free energy and p(T) is the pressure. Now, varying the above partition function with respect to the metric $g_{\mu\nu}$ and substituting it in the definition of the stress-energy tensor in (2.4), we immediately find that

$$T_{\text{equilbrium}}^{\mu\nu} = \left(T\frac{\partial p}{\partial T} - p\right)u^{\mu}u^{\nu} + p\Delta^{\mu\nu} + \mathcal{O}(\partial^{1}).$$
(2.9)

The first term is nothing but the energy density obtained by the Euler relation $\varepsilon + p = sT$. With this, we obtain the stress-energy tensor for an ideal fluid as promised. The above method may seem like overkill but its true power will be manifest when the global symmetry structure is more complicated or when higher order derivative terms are involved.

Next, we move to add terms at first order in the derivative expansion. This gives the following stress-energy tensor

$$T^{\mu\nu} = \mathbb{E} u^{\mu}u^{\nu} + \mathbb{P}\Delta^{\mu\nu} + \sum_{\text{all vectors}} \alpha_{(i)}(T) \left(\mathbb{V}^{\mu}_{(i)}u^{\nu} + u^{\mu}\mathbb{V}^{\nu}_{(i)} \right) + \sum_{\text{all tensors}} \beta_{(i)}\mathbb{T}^{\mu\nu}$$
(2.10)

Here, we denote

$$\mathbb{E} = \varepsilon + \sum_{\text{all scalars}} \gamma_{(i)} \mathbb{S}_{(i)}, \qquad \mathbb{P} = p + \sum_{\text{all scalars}} \delta(i) \mathbb{S}_{(i)}$$
(2.11)

The structures $\mathbb{S}, \mathbb{V}, \mathbb{T}$ are respectively all possible independent scalar, vector, tensor constructed from u^{μ} and T at first order in the derivative expansions.

The list of all possible structures is

scalar :
$$\nabla_{\mu}u^{\mu}, u^{\mu}\nabla_{\mu}T,$$

vector : $\Delta^{\mu\nu}\nabla_{\nu}T, u^{\nu}\nabla_{\nu}u^{\mu},$
tensor : $\sigma^{\mu\nu} := \Delta^{\mu\rho}\Delta^{\nu\sigma}\left(\nabla_{\rho}u_{\sigma} + \nabla_{\sigma}u_{\rho} - \frac{2}{d-1}g_{\rho\sigma}\nabla_{\lambda}u^{\lambda}\right),$
(2.12)

and $\{\alpha_{(i)}, \beta_{(i)}, \gamma_{(i)}, \delta_{(i)}\}\$ are unknown functions of thermodynamic variables, which are referred to as *transport coefficients*. However, a straightforward investigation will reveal that not all structures listed above are linearly independent. First of all, the variables $u^{\mu}(x), T(x)$ must satisfy the conservation law $\nabla_{\nu}T^{\mu\nu} = 0$ order by order. By substituting in the 0th order terms in $T^{\mu\nu}$, one finds relations between the following quantities

$$u^{\mu}\nabla_{\mu}T \sim \nabla_{\mu}u^{\mu}, \qquad u^{\nu}\nabla_{\nu}u^{\mu} \sim \Delta^{\mu\nu}\nabla_{\nu}T.$$
 (2.13)

Thus, one can see that $u^{\mu}\nabla_{\mu}T$ and $u^{\nu}\nabla_{\nu}u^{\mu}$ are not independent and can be remove from the list (2.12). The other type of redundancy is due to the fact that the temperature field T(x) and fluid velocity u^{μ} have no microscopic specification out of equilibrium. This allows us to shift the velocity u^{μ} and temperature field T(x) by terms subleading in the gradient expansion i.e.

$$u^{\mu}(x) \to u^{\mu}(x) + \mathcal{V}^{\mu}(\partial T, \partial u), \qquad T(x) \to T(x) + \mathcal{T}(\partial T, \partial u)$$
 (2.14)

where the vector \mathcal{V}^{μ} and the scalar \mathcal{T} only contain terms at first order in the gradient expansion. This redundancy is known in the literature as *frame choices*. For chapter 3 and 5, we will use the hydrodynamics construction in the *Landau frame* where \mathcal{V}^{μ} and \mathcal{T} are chosen such that

$$u_{\mu}T^{\mu\nu} = -\varepsilon u^{\nu}, \qquad (2.15)$$

which essentially set $\alpha_{(i)}$ and $\gamma_{(i)}$ to zero. Imposing these constrains, one finds that there are only two the remaining terms at 1^{st} order :

$$T_{\text{first derivative}}^{\mu\nu} = -\eta \sigma^{\mu\nu} - \zeta \Delta^{\mu\nu} (\partial_{\lambda} u^{\lambda}), \qquad (2.16)$$

where η and ζ are the shear and bulk viscosity respectively.

The last constraint to be imposed in this system is the positivity of local entropy production near equilibrium [48]. This condition is widely believed to be related by the second law of thermodynamics. This is, however, not entirely true since there is no such thing as the second law for local entropy (for example, refrigerators do exist). However, this condition gives a sensible constraint in most cases and, only recently, has finally been shown to emerge from consistency conditions of effective action for dissipative systems close to equilibrium [63].

This constraint demands the existence of a local entropy current J_S^{μ} such that $u^{\mu}J_S^{\mu} = s$ and $\nabla_{\mu}J_S^{\mu} \ge 0$. In practice, we can express J_S^{μ} in the gradient expansion as

$$J_{S}^{\mu} = su^{\mu} + \sum_{\text{all vectors}} \tilde{\alpha}_{(i)}(T) \mathbb{V}_{(i)}^{\mu} + \sum_{\text{all scalars}} \tilde{\gamma}_{(i)} \mathbb{S}_{(i)} u^{\mu}$$
(2.17)

where $\tilde{\alpha}_{(i)}, \tilde{\gamma}_{(i)}$ are some unknown coefficients. Then, one can proceed by substituting the equation of motion $\nabla_{\nu}T^{\mu\nu} = 0$ in $\nabla_{\mu}J^{\mu}_{S}$. In many cases, the coefficients $\tilde{\alpha}, \tilde{\gamma}$ can be chosen to eliminate terms that are not positive definite³, resulting in the desired structure

$$\nabla_{\mu}J_{S}^{\mu} = \sum F_{j}\left(\alpha_{(i)}, \beta_{(i)}, \gamma_{(i)}, \delta_{(i)}\right) \text{(positive definite combinations of } \partial u, \partial T)^{2}$$

where F_j are some linear combinations of the transport coefficients. The positivity of $\nabla_{\mu}J_{S}^{\mu}$ implies that $F_j \geq 0$. In this setup, fortunately, the entropy current is well-defined and one finds that

$$J_{S}^{\mu} = T^{-1} \left(p \, u^{\mu} - T^{\mu\nu} u_{\nu} \right), \qquad T \, \nabla_{\mu} J_{S}^{\mu} = \frac{1}{2} \eta \, \sigma_{\mu\nu} \sigma^{\mu\nu} + \zeta (\partial_{\lambda} u^{\lambda})^{2} \quad (2.18)$$

which implies that $\eta \ge 0$ and $\zeta \ge 0$. In this case, the positivity of the local entropy production does not put a lot of constraint in the system but as we crank up the complexity of the system, its consequences will be more dramatic.

³Unfortunately, this does not always happen, causing the definition of entropy current to be ambiguous as it depends on arbitrary constants $\{\tilde{\alpha}, ...\}$, see e.g. [64].

Again, this condition seems very simple but the constraints it impose will become more dramatic as we crank up the complexity of our system.

If the system also possesses a U(1) global symmetry, we can also couple it to a background U(1) gauge field A_{μ} and write down the constitutive relation. The stress energy tensor $T^{\mu\nu}$ and the conserved current J^{μ} can be expressed in terms of the same hydrodynamic quantities and background field. The caveat is that, since $T^{\mu\nu}$, J^{μ} are physical quantities and therefore cannot depend on the gauge choice, they can only depend on the field strength $F_{\mu\nu}$. In this case the constitutive relation for J^{μ} is

$$J^{\mu} = nu^{\mu} - \sigma \,\Delta^{\mu\nu} \left(F_{\nu\lambda} u^{\lambda} - T \nabla_{\nu} (\mu/T) \right) + \chi_E \Delta^{\mu\nu} \left(F_{\nu\lambda} u^{\lambda} \right) + \chi_T \left(\Delta^{\mu\nu} \nabla_{\nu} T \right)$$
(2.19)

The parameter *n* is the density of the U(1) charge and $\{\sigma, \chi_E, \chi_T\}$ are transport coefficients. In this case, the coefficients χ_E and χ_T will generate a non-positive definite term in $\nabla_{\mu} J_S^{\mu}$ thus forcing them to be zero.

2.1.2 Breaking translational symmetry

Translational symmetry breaking is one of the defining properties of solid state physics. The lack of translational symmetry is caused by the presence of a lattice or disorder, which is responsible for the finite conductivity of the system. Typically, the lattice/disorder spacing $\ell_{\rm dis}$ is very short compared to the mean free path $\ell_{\rm mfp}$, causing the hydrodynamic gradient expansions to breakdown as $\ell_{\rm dis} \ll \ell_{\rm mfp}$. However, we can still study hydrodynamic properties in the opposite limit, where the translation symmetry is only weakly broken such that the mean free path is still much shorter than the lattice/disorder spacing $\ell_{\rm dis} \gg \ell_{\rm mfp}$. Arguably, the quantum system will still behave like a fluid at a certain scale (see e.g. illustration in Fig. 1.3)

The simplest way to break translational symmetries in this system is to add, by hand, a term analogous to the Drude model. The Ward identity for $T^{\mu\nu}$ in

flat space is modified to

$$\partial_{\mu}T^{\mu t} = 0, \qquad \partial_{\mu}T^{\mu i} = \frac{1}{\tau_{\rm imp}}T^{ti}$$
 (2.20)

where T^{ti} is the momentum in x^i direction and $1/\tau_{imp}$ is momentum relaxation rate [23, 65]. The second equation breaks the spatial translational symmetry in all directions. One can then proceed to compute hydrodynamic quantities by assuming that the stress-energy tensor still retain the original form in (2.6) and (2.16). This approach is a good approximation for conductivities, which are mostly governed by the zeroth order terms in $T^{\mu\nu}$ and continues to give interesting results [65–68]. However, first order hydrodynamics in (2.16) turns out to be inconsistent with the modified conservation equation (2.20).

This formalism can be made more systematic. Instead of adding the term $1/\tau_{imp}$ by hand, the translational symmetry can be incorporated by making the background metric or the background gauge field spatially dependent, similar to suspended graphene or an optical lattice in cold atom experiments. This program has been put forth [69–71] and found applications in a clean graphene experiment where hydrodynamic signatures have been found [24, 25], see e.g. figure 2.2.



Figure 2.2. An illustration of a negative local resistance caused by viscous electron backflow in graphene experiment in [25]. The vortices, which are signatures of hydrodynamics are apparent in sub-panel A and B.

In chapter 3 of this thesis, we proceed with the same background field approach but instead of using the background metric to break translational symmetry, we introduce additional scalar fields to do the job. We couple the system to the background metric and spatially dependent scalar fields ϕ_i such that the new generating function is

$$Z[g_{\mu\nu},\phi_i] = \left\langle \exp\left[i\int d^d x \sqrt{-g}\left(\frac{1}{2}T^{\mu\nu}g_{\mu\nu} + \mathcal{O}_i\phi_i\right)\right]\right\rangle$$
(2.21)

where $\langle O_i \rangle$ is the operator that corresponds to the scalar field source ϕ_i . The stress-energy tensor of this theory obeys the following modified conservation law

$$\nabla_{\nu} T^{\mu\nu} = \langle \mathcal{O}_i \rangle \nabla^{\nu} \phi_i \tag{2.22}$$

The key advantage of this theory is that it has a much simpler holographic dual compared to the previous case, allowing us to compute transport coefficients explicitly. The constitutive relations can be derived systematically in terms of thermodynamic quantities, fluid velocity u^{μ} , background metric $g_{\mu\nu}$ and the scalar fields ϕ_i . We then explore this system and the fate of the KSS bound when the translational symmetry is broken in this particular way.

2.1.3 Introducing anomalous U(1) current

Quantum anomalies are one of the most beautiful and genuine quantum effects. They are phenomena where the classical theory is invariant under a certain global symmetry but this symmetry does not survive when the theory transitions to the quantum regime. The most well-known anomaly is the chiral anomaly. An illustrative example is the massless fermion in even spacetime dimensions. For example, in 3+1 dimensions, the massless Dirac Lagrangian in the presence of a background gauge field is invariant under two U(1) global symmetries called $U(1)_V$ and $U(1)_A$, which transform the fermion field as

$$\psi \to \psi \exp(i\theta), \quad \text{and} \quad \psi \to \psi \exp(i\gamma^5\theta)$$
 (2.23)

However, only one current is conserved. The conservation of current in this setup can be written in the following ways

$$\partial_{\mu}J_{V}^{\mu} = 0, \qquad \partial_{\mu}J_{A}^{\mu} = \kappa \,\epsilon^{\mu\nu\rho\sigma}F_{\mu\nu}F_{\rho\sigma}. \tag{2.24}$$

where the coefficient κ is called the *anomaly coefficient*. Similarly, one observes that putting the theory in curved space also has a similar effect but, this time, with an additional term associated to *the gravitational anomaly*

$$\nabla_{\mu}J^{\mu}_{A} = \epsilon^{\mu\nu\rho\sigma} \left(\kappa F_{\mu\nu}F_{\rho\sigma} + \lambda R_{\mu\nu\alpha\beta}R_{\rho\sigma}^{\ \alpha\beta}\right) \tag{2.25}$$

The quantum anomaly has been studied intensively over the last few decades since the study of pions in the pre-QCD era. Since then, it has been well understood and play an important role in our understanding of particle physics, string theory and condensed matter (see e.g. [72–75] for reviews). The quantum anomaly claim to fame came from its non-renormalisation nature: the diagrammatic computation of the anomaly coefficient is 1-loop exact [76–78], which originates from a topological quantity in the index theorem [79]. Despite all their theoretical success, these beautiful anomaly coefficients had never been measured in nature until the recent developments in the last few years.

The mentioned development materialised from the realisation that a quantum system with anomaly gives rise to a new kind of transport phenomena. We learned from high school physics that a charged particle will circulate around the magnetic field line as it is subjected to an external magnetic field. In a system with an anomaly, however, there exists an unusual current which flows along the magnetic field line and the magnitude of the current is proportional to the magnetic field itself! This phenomenon is dubbed *chiral magnetic effect* [80] and from known microscopic theories, the conductivity associated to this *anomalous current* is indeed fixed by the anomaly coefficients (see e.g. [80– 86]). Subsequently, the strong interaction computation has also been done using gauge/gravity duality and the same relations between anomaly coefficients and anomalous conductivity are also found in simple models.

We are interested to see whether there exists a version of the non-renormalisation

theorem for anomalous conductivities, especially at finite temperature and in the presence of the gravitational anomaly⁴. We investigate this possibility in a large class of strongly interacting quantum field theories with holographic duals and present the result in chapter 4. The theory we are interested in has two U(1) global symmetries i.e. $U(1)_V \times U(1)_A$ and we couple them to two background gauge fields, V_{μ} and A_{μ} respectively. The partition function of this theory is

$$Z[g_{\mu\nu}, V_{\mu}, A_{\nu}] = \left\langle \exp\left[i\int d^4x \sqrt{-g}\left(\frac{1}{2}T^{\mu\nu}g_{\mu\nu} + J^{\mu}_V V_{\mu} + J^{\mu}_A A_{\mu}\right)\right]\right\rangle.$$

We construct the theory such that one of the global symmetries, namely $U(1)_A$ is anomalous. The Ward identities for the currents in this theory are

$$\nabla_{\mu}J_{V}^{\mu} = 0,$$

$$\nabla_{\mu}J_{A}^{\mu} = \epsilon^{\mu\nu\rho\sigma} \left(\kappa F_{A,\mu\nu}F_{A,\rho\sigma} + \gamma F_{V,\mu\nu}F_{V,\rho\sigma} + \lambda R^{\alpha}{}_{\beta\mu\nu}R^{\beta}{}_{\alpha\rho\sigma}\right)$$
(2.26)

where $F_{V,\mu\nu}$, $F_{A,\mu\nu}$ are the field strengths associated to the background field V_{μ} , A_{μ} respectively.

2.1.4 Generalised global symmetry

The global symmetries mentioned earlier are the one that people have been familiar with since the nineteenth or early twentieth centuries. As we mentioned in section 2.1, these symmetries (Poincaré symmetry and U(1) global symmetry) are often associated with transformations of point-like objects (such as a particles) which leave the theory unchanged. Consequently, the conserved charges in d-dimensional systems are obtained by integrating the conserved currents over a spatial volume, e.g. for the U(1) charge

$$Q = \int d \text{ (volume) } J^t = \int d^{d-1} S_\mu J^\mu.$$
(2.27)

where the spatial volume S_{μ} is a vector pointing in the time direction. We can also couple the system to the vector gauge field A_{μ} as we did in the previous

⁴See recents review about this issue in e.g. [44, 87–89]

section. In the language of differential geometry, the conserved current and the gauge field can be classified as a 1–form object (since it contains one index).

But of course, this is not the only global symmetry in quantum field theory, especially when they system consists of extended objects such as superfluid vortices, strings, domain walls, membranes etc. The study of such extended objects plays a crucial role in many areas of physics, particularly in string theory [90]. In these examples, there is also a notion of global symmetry and conserved charge. However, in contrast to the typical conserved charge (2.27) where we integrate the conserved current over the spatial volume, the conserved charge for extended objects is obtained by integrating over a surface (see figure 2.3).



Figure 2.3. An illustration of a conserved charge for a system with string-like objects. In this case, the conserved charge is a flux obtained by integrating the number of a string/field line passing through a 2–dimensional surface (figure from [47]).

For an extended q - 1 dimensional object (q=1 for a particle and q=2 for a string), the integration is done over a p := d - q dimensional surface. Mathematically, the conserved charge for such an object is

$$Q_{\text{gen}} = \int dS_{\mu_1 \mu_2 \dots \mu_p} J^{\mu_1 \mu_2 \dots \mu_p}_{(p)}, \qquad (2.28)$$

where $S_{\mu_1\mu_2...\mu_p}$ is a totally antisymmetric tensor of rank p or a p-form object (and so is $J_{(p)}^{\mu_1...\mu_p}$). Similar to the conventional conserved current (2.27), we can couple this p-form current $J_{(p)}$ to the background gauge field. However, instead of the 1-form gauge field, it will couple to the q-form gauge field $A_{\mu_1\mu_2...\mu_q}$ which is a totally antisymmetric tensor of rank q. Recently, higher-form symmetry has been systematically categorised in [46], although it has been a recurring theme in various areas such as symmetry protected topological phase and topological order e.g. [46, 91, 92], phenomenological models for su-

perfluid vortices [93–95], dislocation/disclination in liquid crystals [96, 97].

How does this have anything to do with plasma physics in 3+1 dimensions? Naively, one may argue that the plasma, which is charged matter coupled to an electromagnetic field, has a U(1) symmetry associated to the conserved current j^{μ} in the Maxwell equation

$$\nabla_{\mu}F^{\mu\nu} = j^{\nu}, \qquad \nabla_{\nu}\left(\epsilon^{\mu\nu\rho\sigma}F_{\rho\sigma}\right) = 0.$$
(2.29)

Here, we can see from the first equation that $\nabla_{\mu}j^{\mu} = 0$ by definition. The second equation is usually thought of as a constraint and trivially vanishes when one expresses the field strength $F_{\mu\nu}$ in terms of the gauge field A_{μ} i.e. $F_{\mu\nu} = \partial_{\mu}A_{\nu} - \partial_{\nu}A_{\mu}$. However, this current j^{μ} is not associated with a global symmetry but a gauge symmetry. It is not a true symmetry but rather a redundancy of the description. The other subtle differences between global and gauge symmetry can be found in the table below

global symmetry	Gauge symmetry
Intrinsic property of the system	a redundancy rather than real symmetry
can be spontaneously broken	cannot be spontaneously broken
classifies physical states	all physical states are gauge invariant
can be anomalous	cannot have anomaly
can couple to background gauge fields	gauge field is dynamical

So, is there any global symmetry in a system described by (2.29) ? It was argued by [46] that the true global symmetry is encoded in the second set of Maxwell equation (2.29). It is the 2-form global symmetry with the current $J^{\mu\nu}$ defined as

$$J^{\mu\nu} = \frac{1}{2} \epsilon^{\mu\nu\rho\sigma} F_{\rho\sigma}.$$
 (2.30)

The conserved charge $Q = \int dS_{\mu\nu} J^{\mu\nu}$ is nothing but the magnetic flux passing though the system as depicted in Fig. 2.3.

With the knowledge of global symmetries at our disposal, we can play the same game we did in the previous section. First we define a partition function
of the theory coupled to the background metric $g_{\mu\nu}$ and the 2-form gauge field $b_{\mu\nu}$ which sources the 2-form current $J^{\mu\nu}$ in the following way :

$$Z[g_{\mu\nu}, b_{\mu\nu}] = \left\langle \exp\left[i\int d^4x \sqrt{-g}\left(\frac{1}{2}T^{\mu\nu}g_{\mu\nu} + J^{\mu\nu}b_{\mu\nu}\right)\right]\right\rangle$$

The conserved currents can be obtained by varying the partition function

$$\langle T^{\mu\nu} \rangle = \frac{-2i}{\sqrt{-g}} \frac{\delta \log Z}{\delta g_{\mu\nu}}, \qquad \langle J^{\mu\nu} \rangle = \frac{-i}{\sqrt{-g}} \frac{\delta \log Z}{\delta b_{\mu\nu}}.$$
 (2.31)

The low energy effective theory with these global symmetries can then be obtained by expressing $T^{\mu\nu}$ and $J^{\mu\nu}$ in terms of hydrodynamic variables. However, with this system, one has to take the direction of the magnetic field lines into account. This amounts to introducing a vector h^{μ} , pointing along the magnetic field lines, as a dynamical variable. After analysing the equilibrium partition function similar to those in section 2.1.1, the conserved currents at zeroth order in the gradient expansion are [47]

$$T^{\mu\nu} = (\varepsilon + p)u^{\mu}u^{\nu} + pg^{\mu\nu} - \mu\rho h^{\mu}h^{\nu} + \mathcal{O}(\partial^{1}), J^{\mu\nu} = \rho (u^{\mu}h^{\nu} - u^{\nu}h^{\mu}) + \mathcal{O}(\partial^{1}).$$
(2.32)

Here ρ denotes the magnetic flux density and μ is the chemical potential associated with the 2-form charge (or equivalently, the rate of change of the free energy with respect to the number of magnetic field line). The details of this derivation and the constitutive relations at first order in derivative expansion can be found in [47] and in chapter 5 of this thesis.

In the context of plasma physics, the situation where the matter (which obeys the Navier-Stoke equation) coupled to the dynamical electromagnetic field has been studied in the framework called *magnetohydrodynamics* (MHD). And while it is a successful self-consistent description, it has one unsatisfying underlying assumption, namely the equation of state is independent of the magnetic field. On the other hand, the framework of [47], discussed in this section, is obtained by utilising only the global symmetry of the system. Hence, the equation of state in this approach is free from the assumptions of the standard MHD formulation. This new framework allows us to explore a larger parameter space than that of the standard formulation. We apply this framework to a strongly interacting quantum field theory with holographic dual coupled to the dynamical U(1) gauge field and report our findings in chapter 5.

2.2 2-point correlation functions and Kubo formulae

The focus of this section is the retarded 2-point correlation function, G^R , of conserved currents. The introduction of a background field allows us to compute this quantity in a very natural way. For example, the the stress-energy tensor in the almost flat metric $g_{\mu\nu} = \eta_{\mu\nu} + h_{\mu\nu}$ (where $\eta_{\mu\nu}$ is the Minkowski metric and $h_{\mu\nu}$ is a small perturbation) can be written as⁵

$$\langle T^{\mu\nu}(x)\rangle_g = \langle T^{\mu\nu}(x)\rangle_{h=0} - \frac{1}{2}\int d^4y \, G^R_{T^{\mu\nu}T^{\rho\sigma}}(x,y)h_{\rho\sigma}(y) + \mathcal{O}(h^2)$$
 (2.33)

where the retarded 2-point function is defined in the operator language as

$$G^R_{T^{\mu\nu}T^{\rho\sigma}}(x,y) := -\Theta(x^0 - y^0) \left\langle [T^{\mu\nu}(x), T^{\rho\sigma}(y)] \right\rangle$$
(2.34)

This method can be applied for all the conserved currents and background fields in previous sections.

The retarded 2-point function play a central role in this thesis for two reasons which are detailed below.

• Firstly, the pole of the 2-point function in Fourier space captures the dispersion relation of the low energy excitations. For hydrodynamics, there are two types of low energy excitations: diffusive mode and sound mode

$$G^{R}(\omega,k)^{-1} = 0 \quad \Rightarrow \begin{cases} \omega = -i\mathcal{D}_{p}k^{2}, & \text{diffusive mode} \\ \omega = \pm c_{s}k - i\mathcal{D}_{l}k^{2} & \text{sound mode} \end{cases}$$

where c_s is the speed of sound, D_p is the diffusion constant and D_l is the sound attenuation. The speed of sound only depends on the thermody-

⁵The computation for the real-time correlation function in the presence of the background field is best treated using Schwinger-Keldysh (or Closed-Time-Path) formalism. See, for example, [50, 98, 99] for the derivation.

namic quantities while the $\mathcal{D}_{p,l}$ are linear combinations of the first order transport coefficients and thermodynamics quantity.

 This bring us to the second point. The residue of some of the retarded functions G^R(ω, k) is proportional to the first-order transport coefficients. This allows us to express transport coefficients as linear combinations of correlation functions. Such expressions are referred to as *Kubo* formulae.

For the reader's convenience, I will use the charge neutral fluid as an example although the extension to a more complicated system is straightforward. The procedure to obtain (2.33) in hydrodynamic models is the following:

- 1) First, we perturb the background fields, such as $g_{\mu\nu}$, around their means values $g_{\mu\nu} = g^{(0)}_{\mu\nu} + h_{\mu\nu}$. For the metric, the mean value $g^{(0)}_{\mu\nu}$ is usually the flat space metric as in (2.33).
- Then, we solve the conservation equation e.g. ∇_μT^{μν} = 0 for dynamical variables (such as the temperature field T(x) and fluid velocity u^μ). As a result, these dynamical variables will be expressed in terms of the background field perturbation h_{μν}.
- Lastly, one can substitute the perturbed background fields and the solution of {T, u^μ}, expressed in terms of h_{μν}, into the constitutive relations. The conserved currents can be expanded to linear order in h_{μν} and by comparing this result to (2.33), we obtain the correlation function.

As an example, let me present a result for a few interesting correlation function in the charged neutral fluid. Firstly, the energy density correlation function is

$$G_{T^{tt}T^{tt}}^{R} = \frac{(\varepsilon+p)|k|^{2}}{\omega^{2} - \left(\frac{\partial p}{\partial \varepsilon}\right)|k|^{2} + i\frac{\omega|k|^{2}}{\varepsilon+p}\left(\zeta + \frac{2d-2}{d}\zeta\right)}.$$
(2.35)

The pole of this 2-point function indicates that the energy in the system is carried by the sound mode with the speed $c_s = (\partial p / \partial \varepsilon)^{1/2}$ and the sound attenuation $\mathcal{D}_l = (\zeta + (2d-2)\eta/d) / 2(\varepsilon + p)$. On the other hand, the correlation

function of momentum density $T^{t\perp}$, orthogonal to the momentum k_i , is governed by the diffusive mode

$$G_{T^{t\perp}T^{t\perp}}^{R} = \frac{\eta |k|^2}{i\omega - \frac{\eta}{\varepsilon + p}|k|^2} + \text{contact term}$$
(2.36)

with the diffusion constant $D_p = \eta/(\varepsilon + p)$. The remaining correlation functions can be obtained by the Ward identities

$$\partial_{\mu}T^{\mu\nu} = 0 \quad \Rightarrow \quad k^{\mu}G^{R}_{T^{\mu\nu}T^{\rho\sigma}} = 0.$$
(2.37)

Finally, one can find linear combinations for these correlations functions and obtain Kubo formulae for the shear and bulk viscosity

$$\eta = -\lim_{\omega \to 0} \operatorname{Im} \partial_{\omega} G^{R}_{T^{xy}T^{xy}}(\omega, k = 0),$$

$$\zeta + \frac{2d-2}{d} \eta = -\lim_{\omega \to 0} \operatorname{Im} \partial_{\omega} G^{R}_{T^{xx}T^{xx}}(\omega, k = 0).$$
(2.38)

It is important to notice that while these formulae do not depend on the microscopic information, they are direct consequences of the conservation law. Therefore they will be modified once we consider systems with different global symmetries e.g. those in section 2.1.1-2.1.3. Nevertheless, the procedure outlined here can still be applied, as we show explicitly in chapter 3-5.

2.3 Bottom-up approach to holographic duality

Obviously, there are a lot more details from string theory and supersymmetric gauge theory which play an important role in the fully fledged gauge/gravity duality. Nevertheless, the following discussion will be about a few essential aspects of the duality which allow us to show that the dual QFT exhibits hydrodynamic behaviour.

In the following sections, we present the gauge/gravity duality as a set of dictionary rules relating QFT physical quantities such as global symmetries, "source" background fields and conserved currents in section 2.1 to quantities in the gravity dual. We then continue by briefly review features in gravity which

are relevant to chapters 3-5 of this thesis.

2.3.1 Capturing global symmetry

The precise form of the duality is obtained by stating that the partition function of a strongly interacting QFT discussed in section 2.1 is equal to the semiclassical partition function of a certain gravity theory with at least one dimension higher. Both partition function are also functions of the same arguments, namely the background metric and the other background fields, which act as sources. This is the celebrated Gubser-Klebanov-Polyakov-Witten (GKPW) relation [100, 101]

$$Z_{\text{QFT}}[g_{\mu\nu},...] = Z_{\text{gravity}}[g_{\mu\nu},...].$$
(2.39)

But what is the generating function of the gravitational theory? How can it depend on the background fields in the QFT? In order to answer these question, let us analyse this dual gravity theory in more details to see how the global symmetry and background fields are manifested.

First of all, we need to recall that this gravity theory is, in fact, a certain low energy limit of string theory. It does not consist of only dynamical graviton but also all sort of scalar fields Φ_i and higher form gauge fields $A^{(p)}_{\mu_1...\mu_p}$. Schematically, the action of the "gravity dual" in d + 1 dimensions is

$$\mathcal{L} = \mathcal{R} - \sum_{\text{all scalars}} Z_n(\Phi_i) (\partial \Phi_i)^2 - V(\Phi_i) + \sum_{\text{all gauge fields}} Y_m(\Phi_i) (F_{(p)})^2 + (\text{even more complicated terms involving } \mathcal{R}, \Phi_i, A^{(p)}, F_{(p)})$$
(2.40)

where \mathcal{R} is the Ricci scalar in d + 1 dimensions. The functions $\{Z_m, Y_m\}$ can be either constant or non-trivial functions of scalar fields Φ_i . The detail of these functions and more complicated terms can be fixed by knowing their string theory origin. For example, the gravity dual to d = 4, $SU(N_c) \mathcal{N} = 4$ supersymmetric Yang-Mills at strong coupling is governed by following action

$$S = \frac{N_c^2}{8\pi^2} \int d^5 X \sqrt{-G} \left(\mathcal{R} + 12\right)$$
 (2.41)

Alternatively, one can treat $\{Z_m, Y_m\}$ as phenomenological parameters to "cook up" gravity duals with desirable properties. I will follow the second approach in this thesis.

Examples where we have evidence that the duality exists are when the solution of the action (2.40) has an asymptotic region which is described by the Anti-de Sitter (AdS) metric

$$ds^{2} = G_{ab}dX^{a}dX^{b} = \frac{du^{2}}{4u^{2}} + \frac{L}{u}\left(g_{\mu\nu}(u)dx^{\mu}dx^{\nu}\right)$$
(2.42)

where X^a describes d + 1 dimensional coordinate system $(x^{\mu}, u), x^{\mu}$ describes the coordinates on a surface where u = constant and L is the characteristic length scale of the AdS space, called the AdS radius. The aforementioned asymptotic region is when $u \to 0$, referred to as the boundary of the AdS space. Here, the first dictionary rule can be introduced to relate the metric $g_{\mu\nu}(u)$ to the source metric $g_{\mu\nu}$ in the QFT.

• The background metric $g_{\mu\nu}$ coupled to the QFT stress-energy tensor $T^{\mu\nu}$ is the asymptotic value of the metric $g_{\mu\nu}(u)$ describing the geometry of a d dimensions surface at $u \to 0$.

$$g_{\mu\nu}|_{\text{QFT}} = \lim_{u \to 0} u \, G_{\mu\nu} = g_{\mu\nu}(u \to 0).$$
 (2.43)

In fact, a similar procedure can be applied to other fields in the gravity dual. The small u expansion in the asymptotic region of all the fields $\mathcal{F} = \{g_{\mu\nu}, \Phi_i, ...\}$ in (2.40) can be written as

$$\mathcal{F}(u,x) = u^m \Big[f^{(0)}(x) + f^{(1)}(x)u + \dots + u^n f^{(n)}(x) + \dots \\ + \tilde{f}^{(n)}(x)u^n \log u \Big], \quad \text{where } \tilde{f}^{(n)} = 0 \text{ for } d \text{ odd.}$$
(2.44)

Solving this expansion using the equation of motion, one finds that all the functions $f^{(i)}$ can be written in terms of two independent functions $f^{(0)}(x)$ and

 $f^{(n)}(x)$. A standard procedure [101, 102] indicates that the leading term in the expansion, $f^{(0)}$ should be interpreted as a source in the QFT picture. This leads us to a more general dictionary rule:

• The non-dynamical background field $f^{(0)}(x) = \{g_{\mu\nu}(x), A_{\mu}(x), ...\}$, which couple to the operator in QFT in d dimensions sets the asymptotic value of the classical dynamical fields $\mathcal{F}(u, x)$ propagating in the d + 1 dimensional spacetime, i.e.

$$f^{(0)}(x)|_{\text{QFT}} = \lim_{u \to 0} u^{-m} \mathcal{F}(u, x).$$
(2.45)

Consequently, the global symmetry \mathfrak{G} of the QFT is translated to a gauge symmetry \mathfrak{G} on the dual gravity side. This is due to the fact that the gauge redundancy of the background fields $f^{(0)}$ is transferred to the dynamical fields $\mathcal{F}(u, x)$ via the expansion (2.44). As a result, the gravity action defined in (2.40) is gauge invariant under the gauge transformations of \mathcal{F} if the QFT operator sourced by $f^{(0)}$ is a conserved current.

The other crucial point is how to incorporate finite temperature in the gravity dual. One can start by putting a QFT on a cylinder as in section 2.1.1. According to the dictionary rule (2.43), this introduces a time circle, set by the QFT temperature, at AdS boundary. One find that the thermal cycle smoothly shrinks to a point as we move further in the bulk of the AdS space, leaving the geometry in a cigar shape depicted in Fig. 2.4. Interestingly, this geometry is nothing but the Euclideanised black hole geometry. By undoing the Wick rotation, we find that the temperature effect can be obtained by introducing a black hole horizon with Hawking temperature equal to the temperature of the dual QFT.

The black hole also plays another important role in determining the unknown function coefficients $f^{(n)}(x)$ in the near boundary expansion (2.44). Essentially, the boundary condition at the black hole horizon (e.g. ingoing boundary condition, regularity condition) allows one to determine $f^{(n)}(x)$, thus the profile of $\mathcal{F}(u, x)$ is uniquely determined by $f^{(0)}$.



Figure 2.4. The evolution of the thermal cycle from the boundary $u \rightarrow 0$ into the bulk region u > 0. The circle shrinks smoothly to a point at distance $u = u_h$. The geometry is terminated at this point and forms a cigar-like structure.

By piecing together all this information, we now have the ingredients in the gravity dual that correspond to the physical quantities we are interested in, namely the global symmetries, background fields that source the conserved currents and finite temperature. With this, we will be able to construct a strongly interacting quantum field theory from a gravity dual that has a desirable global symmetry structure. To be more explicit, a few examples of bottom-up holographic models and their corresponding QFT properties are listed below.

• Conformal field theory in *d* dimensions with no additional global symmetries at finite temperature corresponds to the black hole geometry in the Einstein-Hilbert action

$$S = \frac{1}{2\kappa} \int d^{d+1}x \sqrt{-G} \left(\mathcal{R} + d(d-1) \right).$$
 (2.46)

This can be seen from the fact that the only relevant background field and conserved charges are the background metric $g_{\mu\nu}$ and the stress-energy tensor $T^{\mu\nu}$. The cosmological constant originates from the string theory setup. For our purposes, we can view it as a term added to ensure that the geometry on the gravity dual side has an asymptotic AdS region.

- To add additional global symmetries, such as a global U(1) symmetry on

the QFT side, we can add a Maxwell term to the action:

$$S = \frac{1}{2\kappa} \int d^{d+1}x \sqrt{-G} \left(\mathcal{R} + d(d-1) - \frac{1}{4e^2} F_{ab} F^{ab} \right).$$
(2.47)

where $F_{ab} = \partial_a A_b - \partial_b A_a$ where $A_\mu(u \to 0, x)$ is the QFT background gauge field that sources the U(1) conserved current J^μ . If the global symmetry happens to be a non-Abelian group, such as SU(2), we simply change the gauge field A_a in the gravity dual to a non-Ablelian gauge field A_a^i , where *i* runs over the group index.

2.3.2 Holographic thermal 1-point and 2-point function

We are now ready to use the GKPW to obtain expectation values and correlation functions.

In the semi-classical approximation, the partition function of the gravity theory is determined by the onshell action, i.e.

$$Z[g_{\mu\nu},...] = \exp\left[iS_{\text{gravity}}\right] = \exp\left[\frac{i}{2\kappa}\int d^{d+1}\sqrt{-G}\left(\mathcal{R}+...\right)\right]$$
(2.48)

where fields $\mathcal{F}(u, x)$ in S_{gravity} are determined by the sources $f^{(0)}(x) = \{g_{\mu\nu}(x), ...\}$ and the boundary condition at the horizon. I would like to emphasise that, by keeping track of the background fields, we can conveniently use the same definition of the stress-energy tensor and the other conserved currents as in section 2.1, such as (2.4) and (2.31). Moreover, the retarded 2-point functions of these operators can also be obtained straightforwardly using the prescription in (2.33). For example, the stress-energy tensor 1-point and 2-point function are

$$\langle T^{\mu\nu} \rangle = \frac{2}{\sqrt{-g}} \frac{\delta S_{\text{gravity}}}{\delta g_{\mu\nu}}, \quad G^R_{T^{\mu\nu}T^{\rho\sigma}} = -2 \frac{\delta}{\delta g_{\mu\nu}} \left(\sqrt{-g} \left\langle T^{\mu\nu} \right\rangle \right)$$
(2.49)

One can clearly see that this way of computing 1-point and 2-point function is a lot easier compared to the conventional QFT method beyond a few loops correction. Practically, one only needs to solve a differential equation for $\mathcal{F}(u, x)$ with the boundary condition determined by $f^{(0)}$, insert solutions in the gravity action S_{gravity} and apply the formulae (2.49). A similar procedure can be used to obtain correlation functions of the other conserved currents.

Historically, this procedure made contact with hydrodynamics in a series of heroic works by Policastro, Son and Starinets [103–105] for the gravity theory dual to $\mathcal{N} = 4$ supersymmetric Yang-Mills. They found that the 2-point functions that agree with those predicted by hydrodynamics (2.35)-(2.36) with the equation of state

$$p = \frac{1}{3}\varepsilon \sim T^4, \quad \eta = \frac{s}{4\pi}, \quad \zeta = 0$$
(2.50)

Soon after, this computation was extended to holographic system with different symmetries and equations of state (see e.g. [106-109] for early work on this construction). Later on, the way to extract the constitutive relation (in section 2.1.1) was developed [110-113], allowing one to link properties of the black hole dynamics and hydrodyhamics (this prescription is known under the name *fluid/gravity correspondence*). These works have been inspirations as well as the foundation for numerous works in the past few years, including those in this thesis.

2.3.3 Holographic RG flow

It is natural to ask what the physical meaning of the radial direction labelled by the coordinate u in the previous section is. This is a long-standing question which still remains unanswered. Earlier work [22, 114, 115] strongly suggested that it is related to the energy scale of the theory and that the gauge/gravity duality is the geometric realisation of the Wilsonian renormalisation group flow (see Fig. 2.5), where the region near the boundary u = 0 is associated to the UV fix point.

Although the precise relation between the radial direction and Wilsonian RG flow is still unclear, there are evidences to suggest that the above statement is not completely wrong. For example, the action of simple holographic models such as those in (2.46)–(2.47) diverges as $u \rightarrow 0$, reminiscent of a UV divergence in conventional QFT. Thus, for the holographic generating function to



Figure 2.5. The comparison between the AdS space and series of block spin transformation along the Wilsonian RG flow.

be physical, one needs to regularise the gravity action by introducing a "cutoff surface" at $u = \epsilon \ll 1$ then add a local counter-term to S_{gravity} to subtract the divergence [101, 102]. One may also extract the Callan-Symanzik equation by integrating out the near boundary region $u < 1/\Lambda$, analogous to momentum shell renormalisation [116, 117] (see also [118] for a review of development before 2011 and references).

If we adopt this notion of RG flow, one can drive the theory away from its UV fixed point by turning on a relevant deformation. The simplest way to do this is to add a scalar field ϕ , which sources a relevant operator $\langle \mathcal{O}_{\phi} \rangle$ with scaling dimensions $\Delta_{\mathcal{O}} < d$. In holography, this can be done by introducing a scalar field Φ , whose boundary condition set by the source ϕ , in the gravity action i.e.

$$S_{\text{gravity}} = \int d^{d+1} X \sqrt{-G} \left(\mathcal{R} - (\partial \Phi)^2 - V(\Phi) \right), \qquad (2.51)$$

where the potential $V(\Phi)$ triggers the deformation away from the AdS boundary. It can also be tuned to obtained a desirable geometry in the bulk (or IR QFT) [119, 120]. This procedure is also extendable to the holographic RG flow for a QFT with finite density [121, 122] by studying the action

$$S_{\text{gravity}} = \int d^{d+1} X \sqrt{-G} \left(\mathcal{R} - (\partial \Phi)^2 - V(\Phi) - \frac{Z(\Phi)}{4} F_{ab} F^{ab} \right), \quad (2.52)$$

where the function $Z(\Phi)$ sets the relevance of the U(1) charge at different stage along the RG flow. This ingredient of the holographic construction allows us to explore large classes of theories with the same global symmetry but different equations of state and thus allows us to check the validity of universal statements such as the KSS bound (1.1) and those in section 1.2. Interestingly, one can engineer holographic theories with the same scaling relations as those found in high temperature superconductor materials [123, 124].

2.3.4 The "membrane paradigm"

If we take seriously the AdS radius u as an energy scale in the Wilsonian renormalisation scheme, it is tempting to say that the low energy dynamics, such as DC conductivity and viscosities are captured by the deep interior of the AdS spacetime (see e.g. Fig. 2.5). In systems with finite temperature, the "IR cutoff" is generated by the black hole horizon, see Fig. 2.6. Since we know that the QFT dual to a black hole is governed by hydrodynamic principles, one might be tempted to say that the region near the black hole horizon (called the stretch horizon) can be thought of as some sort of fluid and might even be the same fluid in dual QFT. The idea that the black hole horizon behaves as a fictitious fluid is not new. In fact, it dates back to work on black hole in 70's-80's known as *the black hole membrane paradigm* [125].



Figure 2.6. An illustration of AdS space with black hole. The horizon is located at $u = u_h > 0$ where the scale u_h is set by the temperature. This geometry can be obtained by Wick rotating the cigar geometry in Fig. 2.4. The excitations with low energy probe deeper in the interior near the horizon, suggesting that the low energy dynamics is governed by the physics near the black hole horizon.



cates that the fictitious fluid at the horizon and the fluid described by the dual QFT are not the same in general [110, 116, 126–130]. This is due to the fact that the hydrodynamic data of the dual QFT contain not only a contribution from the stretched horizon but also from the bulk of AdS spacetime. Nevertheless, some hydrodynamic data of the fictitious fluid at the horizon can still be mapped to that of the dual QFT [126]. This is due to the fact that there exist "conserved currents" along the AdS radial direction (u-direction) which carry the hydrodynamic data from the stretched horizon over to the boundary, allowing one to map them to the dual QFT.

Let us briefly outline how this works for the computation of the shear viscosity. Typically, if one wants to compute $G_{T^{xy}T^{xy}}^{R}(\omega, 0)$ to feed into the Kubo formula (2.38), it is required to solve Einstein's equation, which is a second order differential equation. However, for the shear viscosity, one can arrange the equation of motion in such a way that it becomes a total derivative along the u-direction i.e.

Second order DiffEqn
$$\Big|_{\text{low energy}} \Rightarrow \frac{d}{du} \Big[\mathcal{J}(u) \Big] = 0.$$
 (2.53)

In this case, the radially conserved current \mathcal{J} contains the information about the shear viscosity of the stretched horizon and the dual QFT when evaluate it at the horizon $u = u_h$ and at the AdS boundary $u \to 0$ respectively. Once the explicit form of \mathcal{J} is known, one can use (2.53) to extract the shear viscosity of the dual QFT purely from the near-horizon data.

The existence of this radially conserved current not only simplifier the computation of the transport coefficients but is also a smoking gun for a universality relation. To be more precise, the physical quantities contained in the radial conserved currents are fully captured by the deep IR information (i.e. at the stretched horizon) and do not depend on the specific details of the "RG flow" along the u-direction. This concept plays a crucial role in identifying which quantities we should expect to display universal behaviour.

2.3.5 Higher derivative holography

From a string theory point of view, the fact that some QFTs have a simply gravity dual such as those dual to $\mathcal{N} = 4$ supersymmetric Yang-Mills in (2.41) is not a coincidence. String theory is a quantum theory of gravity with two fundamental parameters: the string characteristic length ℓ_s and the string coupling g_s , which controls the quantum fluctuations in the dual gravity theory. For QFTs such as SU(N) $\mathcal{N} = 4$ supersymmetric Yang-Mills to have a simple gravity dual, both of these parameters have to be small. This limit corresponds to sending the rank N of the gauge group SU(N) and the 't Hooft coupling $\lambda := g_{\rm YM}^2 N$ in the dual QFT to infinity. To be more precise, one can relate the string parameters ℓ_s, g_s to the QFT parameters λ, g_s, N as

$$\frac{L}{\ell_s} = \lambda \gg 1$$
 and $4\pi g_s = g_{\rm YM}^2 = \frac{\lambda}{N} \ll 1.$ (2.54)

Similar limits can also be found for the other supersymmetric theory with holographic dual, see e.g. [52].

While the large N limit is somewhat pathological, the large 't Hooft coupling limit represents the fact that the QFT we are interested in has infinitely strong coupling. The combination of the two limits greatly simplifies the computations in the gravity dual. However, this is nothing but one corner in the parameter space. To explore different regions of parameter space and make a universal statement such as the bound (1.1), one needs to find a way to extend our computation beyond large N, λ limit.

While it is very difficult to move away from the large N limit due to the more problematic nature of the string coupling g_s corrections⁶, the large N limit with $1/\lambda$ correction is much more tame and very well-documented [135–141]. It turns out that, in order to include finite λ corrections in the gravity dual, one has to add higher-derivative terms. For example, in $\mathcal{N} = 4$ SYM, the first 't Hooft coupling correction comes from the following higher derivative

 $^{^6} See$ recent developments on 1/N corrections in the large 't Hooft coupling limit in e.g. [131–134].

term [141]

$$\Delta S = \int d^5 X \sqrt{-G} \left(\gamma W \right) \tag{2.55}$$

where W is a linear combination of forth powers of the Weyl tensor C_{abcd} (c.f. [141]).

$$W = C^{abcd} C_{mbcn} C_a^{\ rsm} C_{\ rsc}^n + \frac{1}{2} C^{adbd} C_{mnbc} C_a^{\ rsm} C_{\ rsd}^n.$$
(2.56)

Note also that, in the string theory construction, the constant γ would also depends on scalar fields in the dual gravity side.

The above finite 't Hooft coupling correction is specific to $SU(N) \mathcal{N} = 4$ supersymmetric Yang-Mills. Given the vast landscape of string theory, it is reasonable to expect that generic higher derivative corrections could occur for the other QFTs with holographic duals. The higher derivative couplings e.g. γ could also depend on the scalar fields as in the case of $\mathcal{N} = 4$ supersymmetric Yang-Mills but they are constrained by consistency conditions such as unitarity, causality, etc., as illustrated in [142–147].

5

3.1 Motivation

In recent years, numerous developments in relativistic strongly interacting quantum field theory at finite temperature have been made using the guage/gravity duality [22, 100, 101], which reduces the computations of 2-point functions to solving certain differential equations in the classical general relativity. In the IR limit, if the theory remains translational invariant, many theories of this type can be described using macroscopic variables governed by the conservation of energy-momentum : the hydrodynamic theory. Equipped with this description, the Green's functions obtained from gauge/gravity duality can be interpreted in terms of the language of relativistic hydrodynamics [103, 110] and allow us to predict universal bound for transport coefficients [30, 32–35, 148], defined by hydrodynamics constitutive relations. One of the most interesting bounds is the shear viscosity/entropy density, $\eta/s \ge 1/4\pi$ [30], which has been conjectured to be related to the minimum entropy production of the black hole in the dual gravity theory [149, 150].

Interesting applications of the gauge/gravity duality and relativistic hydrodynamics have also been found in the condensed matter systems [23, 31, 151, 152]. Despite the fact that the translational symmetry in such systems is broken due to lattice/disorder, the transport properties derived in holographic models [65, 69, 124, 153–174] fit surprisingly well with the hydrodynamic prescriptions. Moreover, the universal bounds, similar to those mentioned earlier, have been proposed [36] and some of them can also be demonstrated explicitly [38, 39]. Recently [129, 175, 176] also demonstrate that the DC transport coefficients can be extracted from the forced Navier-Stokes equations. Evidences from the work mentioned above hint that there should be a hydrodynamics-like description for the disordered theory.

If there is indeed a hydrodynamics-like description for theory without translational symmetry, one would naturally ask the following : how would such description differ from the standard relativistic hydrodynamics ? Which of the intuitions and universal results in the hydrodynamics are still applicable¹? In this work, despite there are potentially interesting physics to be explored at strong disordered theory, we focus on the hydrodynamics-like theory when translational symmetry is weakly broken as it should be more closely related to the standard hydrodynamics. We also restrict ourselves to the type of models where translational symmetry breaking is the one in simple holographic models described below.

In ref [23], the effective theory motivated by hydrodynamics was proposed to describe the quantum critical transport where the translational symmetry is weakly broken. The dynamics of this theory is governed by the following

¹Some aspect of this question has already been explored in [69]

equation of motion

$$\nabla_{\mu}T^{\mu 0} = 0, \qquad \nabla_{\mu}T^{\mu i} = -\Gamma T^{0i},$$
(3.1)

where the index i = 1, 2, d - 1 denotes the spatial dimensions. The dimensionful quantity Γ sets the scale for the broken translational symmetry and corresponds to the width of the Drude peak (see e.g. [65]). The stress-energy tensor is assumed to have the standard relativistic hydrodynamics form

$$T^{\mu\nu} = \epsilon \, u^{\mu} u^{\nu} + p \, \Delta^{\mu\nu} - \eta \, \sigma^{\mu\nu}, \qquad (3.2)$$

where the notation can be found in e.g. [51] and in the Appendix 3.5.1. The model successfully captures, in particular, thermo-electric conductivity and seems to be consistent with holographic computations mentioned above, see also [69] and references therein.

However, the theory described by (3.1)-(3.2) has a few draw back. As pointed out in [65, 177, 178], the above model's predictions do not agree with those from simple holographic model of [179, 180] beyond the leading order in the derivative expansion. Moreover, the correlation functions are not correctly related by the Ward identity derived from (3.1).

Alternatively, we use insight from holographic models [65, 158, 163, 168, 173, 180, 181]. In these models the translational symmetry is broken by the massive graviton or spatial dependent massless scalar fields in the dual gravity theory.² We following the terminology of [69] and refer to these models as theories with *mean field disorder*. From the dual theory point of view of the holographic theory with massless scalar fields, the source ϕ_i breaks the translational symmetry explicitly and the conservation of stress-energy tensor is modified to be

$$\nabla_{\mu}T^{\mu\nu} = \langle \mathcal{O}_i \rangle \nabla^{\nu} \phi_i \tag{3.3}$$

where $\langle O_i \rangle$ is the expectation value of the operator sourced by ϕ_i . From the point of view of hydrodynamics, the above setup is equivalent to putting the

²Relations between classes of massive gravity and models with scalar fields are discussed in [173].

fluid in the manifold with background metric $g_{\mu\nu}$ and background source fields ϕ_i which breaks translational symmetry. At the equilibrium, the metric is set to be flat and the scalar sources have the profile $\phi_i = mx^i$. Taking the scalar field ϕ_i into account, the constitutive relation will also depend on the scalar fields, unlike (3.1). This coupling between fluid and spatial dependent scalar fields has already been explored earlier in [182] and more recently in [177, 178]. The modified constitutive relation for $T^{\mu\nu}$ generally has more terms than those in (3.2). The coefficients in front of independent structures in the modified constitutive relations in [177, 178, 182] are obtained by fluid/gravity method[110] for certain gravity dual theories. However, there should be general relations between the Green's function and the coefficients in the constitutive relations, which may differ from those in the standard hydrodynamics³.

The purpose of this work is to find a systematic way of constructing the constitutive relations that also include the spatially dependent scalar fields and try to answer the questions mentioned earlier. We pay special attention to the shear viscosity and the viscosity/entropy density bound. One of our key result is that the shear viscosity η defined as coefficients of the shear tensor $\sigma^{\mu\nu}$, beyond the leading order in gradient expansion, differs from the value η^* extracted from standard definition $\eta^* = -\lim_{\omega \to 0} (1/\omega) \operatorname{Im} G^R_{TxyTxy}(\omega, k = 0)$. This can be seen both from the constitutive relation, where we see that η^* is polluted by the additional terms due to the scalar fields, and from holographic computation, where η is extracted using fluid/gravity method [177, 178, 182] while η^* is obtained by directly computing the retarded Green's function.

The body of this work is consist of two main parts. In section 3.2, we focus on the constitutive relation of the effective hydrodynamics theory while the holographic computations are discussed in section 3.3. To be more precise, in section 3.2.1, we build up the constitutive relation of $T^{\mu\nu}$ and $\langle O_i \rangle$ in terms of hydrodynamics variables and $\nabla \phi_i$, up to the second order in the derivative expansions. The gradient expansion in this work is organised using the anisotropic scaling of [177, 178]. This procedure is inspired by the construction of higher order hydrodynamics [50, 64, 110, 184]. In section 3.2.2, we outline

³The readers can find modern reviews of the subjects in e.g. [51, 183]

a consistent method to extract the retarded Green's function and show that η^* also include the other transport coefficients, not only the shear viscosity η . We then move on to the holographic computation, where the action and thermodynamics quantities are summarised in 3.3.1. We then compute η/s using the result from fluid/gravity [177, 178] and show that the KSS bound is violated in section 3.3.2. The computation of η^*/s at the leading order can be found in 3.3.3, which are differ from the expression of η/s in the previous section. The numerical profile of η^*/s and η/s at arbitrary value of disorder strength m/T are shown in 3.3.4. We discuss the results of this work and open questions in 4.5. An appendix contain structures in the constitutive relation.

Note added : Near the final stage of this work, we learned that [185] found the same result for η^*/s . While the manuscript is in the preparation stage, [40] appears and has overlaps with our computations in section 3.3 but with different interpretation.

3.2 Effective theory for systems with broken translational symmetry

In this section, we first outline the procedure of how to construct the constitutive relation when the zero density fluid is coupled to the background metric $g_{\mu\nu}$ and the scalar field ϕ_i . Our expressions valid only in 2 + 1 dimensions fluid but it would be straightforward to extend them to arbitrary dimensions. Our notation is closely related to those in [64] and are explained in Appendix 3.5.1. We make a small comment regarding how the role of shear viscosity, η , in the entropy production rate compared to the conformal fluid. Next, we describe the procedure to extract Green's function from the constitutive relation and the equation of motion. We show that G_{TxyTxy}^R also contains higher derivative terms even at linear order in ω .

3.2.1 Constructing the constitutive relation

Just as in the construction of the standard hydrodynamics (those with translational symmetry), we expand $T^{\mu\nu}$, J^{μ} , $\langle \mathcal{O}_i \rangle$ in terms of the macroscopic variables $\{\mathcal{E}, u^{\mu}\}$ and background fields $\{g_{\mu\nu}, \phi_i\}$ order by order in the derivative expansion along x^{μ} direction. Since the scalar field, ϕ_i is explicitly proportional to x^i , Instead of the usual gradient expansion, we also set the momentum relaxation scale to be a small parameter as in [177, 178]. Let us call this small parameter δ , the magnitude of the gradient of the fluid variables $\{T, u^{\mu}, g_{\mu\nu}\}$ and the momentum relaxation scale m have the following scaling

$$\partial T \sim \delta, \qquad \partial u \sim \delta, \qquad \partial g \sim \delta, \qquad m \sim \delta^{1/2}.$$
 (3.4)

This is done according to the previous study that the momentum relaxation rate $\Gamma \sim m^2$ e.g. [65]. Therefore, the frequency ω of the fluid is of the same scale as Γ .

To systematically construct the constitutive relation, it is convenient to decompose the stress energy tensor into the following form

$$T^{\mu\nu} = \mathcal{E}u^{\mu}u^{\nu} + \mathcal{P}\Delta^{\mu\nu} + t^{\mu\nu}, \qquad (3.5)$$

where we choose to work with the Landau frame i.e. $u_{\mu}t^{\mu\nu} = 0$. Note that the above assumption might not be applicable for the theory without translational symmetry in general. In this work, we assume that the fluid remains translational invariant at equilibrium as this also happens in the holographic models with mean field disorder. Consequently, around the equilibrium, one can choose terms \mathcal{E}, \mathcal{P} such that they contain no derivative in $\{u^{\mu}, \mathcal{E}\}$ and the scalar fields ϕ_i only enters $t^{\mu\nu}$ as $\nabla \phi_i$. Thus, the nontrivial task is reduced to building the transverse symmetric tensor out of the macroscopic variables $\{T(x), u^{\mu}, g_{\mu\nu}, \partial_{\mu}\phi_i\}$ and their derivatives up order δ^2 . Note that constitutive relation in (3.5) must also satisfy the equation of motion (3.3). In other words, the modified Ward identity (3.3) implies that the constitutive relations

must satify one scalar and one vector equation

$$0 = -D\mathcal{E} - (\mathcal{E} + \mathcal{P})\nabla_{\mu}u^{\mu} + u_{\nu}\nabla_{\mu}t^{\mu\nu} - \langle \mathcal{O}_{i}\rangle D\phi_{i},$$

$$0 = (\mathcal{E} + \mathcal{P})Du^{\mu} + \nabla_{\perp}^{\mu}\mathcal{P} + \Delta^{\mu}{}_{\nu}\nabla_{\rho}t^{\rho\nu} - \langle \mathcal{O}_{i}\rangle\nabla_{\perp}^{\mu}\phi_{i}.$$
(3.6)

Here, we define the derivative $D \equiv u^{\mu} \nabla_{\mu}$ and $\nabla^{\mu}_{\perp} \equiv \Delta^{\mu\nu} \nabla_{\nu}$. The above equations put constraints on all scalars and vectors one can put into the constitutive relation. Using the first constraint, one may choose to write down a scalar in terms of the other scalars at the same order. The second constraint can be used in the same way to eliminate one vector. We follow the convention of [51] to eliminate $D\mathcal{E}$ and Du^{μ} so that the derivatives of T(x) and u^{μ} only enter the constitutive relation as $\nabla^{\mu}_{\perp}T$ and $\nabla^{\mu}_{\perp}u^{\nu}$. The scalar fields, ϕ_i , however, contain both derivatives. Nevertheless, it is still convenient to decompose them into $D\phi_i$ and $\nabla^{\mu}_{\perp}\phi_i$ as the former vanishes at equilibrium $u^{\mu} = (1, 0, 0)$.

The procedure described so far is almost identical to the construction of the standard relativistic hydrodynamic constitutive relation. However, we would like to point out a few caveats in the above construction. First of all, despite the similarity of the notation, the parameters \mathcal{E} is the energy density but \mathcal{P} is not the pressure. Under our assumption, the energy density, $\epsilon \equiv T^{00} = \mathcal{E}$, as $t^{\mu\nu}$ is chosen in the Landau frame. At order δ^1 , the spatial diagonal parts are $T^{xx} = T^{yy} = \mathcal{P}$. However, terms such as $\Delta^{\mu\nu} \nabla(\phi)^{2N}$ with N = 1, 2, ...may also be part of $t^{\mu\nu}$ at higher order in δ due to the fact that they are not ruled out by the frame choice. Nevertheless, the correction terms to \mathcal{P} will be vanishes in the traceless case $T^{\mu}_{\ \mu} = 0$. Regardless of the ambiguity, the spatial components T^{ii} of the stress-energy tensor is still not the pressure in the simple holographic theory [180]. There, the pressure, p, is obtained from the thermodynamics relation $\epsilon + p = sT$. Lastly, the scaling scheme (3.4), implies that the scalar expectation value \mathcal{O}_i must be expanded up to order $\delta^{5/2}$ so that equation of motion (3.3) can be solved consistently order by order. We would also like to emphasize that it is not necessary to set the scaling such that $\omega \sim m^2$ as in (3.4)The constitutive relation for the fluid coupled to the scalar field with spatial dependence has already been considered in [182]. There, the consitutive relations are expanded with the scaling scheme $\partial u \sim \partial T \sim \partial g \sim \partial \phi$ up to the second order in the derivative expansion. The scaling scheme is indeed convenient to incorporate the effect of broken translational symmetry into the first order hydrodynamics. However, it should also be possible to take $\omega \sim m^N$ (with N > 2) to take into account the higher order effect of the translational symmetry breaking scale m. We will come back to comment on this point later in this section.

We list all possible independent scalars, vectors and transverse symmetric tensors, which we used to construct the consitutive relation up to order δ^1 in Appendix 3.5.1. The structures of higher order than δ^1 can be consistently built up but the number of independent terms grows very quickly. For the purpose of our work, we only list the tensors that would enter the stress-energy tensor.

The most general tensor $t^{\mu\nu}$ in (3.5), expanded up to order δ^2 can be written as

$$t^{\mu\nu} = -\eta \sigma^{\mu\nu} - \eta_{\phi} \Phi^{\mu\nu} + t^{\mu\nu}_{(2)} - \Delta^{\mu\nu} \left(\zeta \nabla_{\mu} u^{\mu} + \zeta_1 D \phi_i D \phi_i + \zeta_2 \nabla_{\perp\mu} \phi_i \nabla^{\mu}_{\perp} \phi_i - P_{(2)} \right).$$
(3.7)

The scalar, $P_{(2)},$ and orthogonal tensor, $t^{\mu\nu}_{(2)},$ of order δ^2 terms can be written explicitly as^4

$$P_{(2)} = \zeta \tau_{\pi} D(\nabla_{\mu} u^{\mu}) + \xi_{1} \sigma^{\mu\nu} \sigma_{\mu\nu} + \xi_{2} (\nabla_{\mu} u^{\mu})^{2} + \xi_{3} \Omega^{\mu\nu} \Omega_{\mu\nu} + \tilde{\xi}_{4} \nabla_{\perp\mu} \mathcal{E} \nabla^{\mu}_{\perp} \mathcal{E}$$

$$+ \xi_{5} R + \xi_{6} u^{\mu} u^{\nu} R_{\mu\nu} + \xi_{7} (\nabla_{\perp\mu} \phi \cdot \nabla^{\mu}_{\perp} \phi)^{2} + \xi_{8} (D\phi \cdot D\phi)^{2}$$

$$+ \xi_{9} (\nabla_{\perp\mu} \phi \cdot \nabla^{\mu}_{\perp} \phi) (D\phi \cdot D\phi) + \xi_{10} (\nabla^{\mu}_{\perp} \phi \cdot D\phi) (\nabla_{\perp\mu} \phi \cdot D\phi)$$

$$+ \xi_{11} (\nabla_{\perp\mu} \phi \cdot D\phi) \nabla^{\mu}_{\perp} \mathcal{E} + \xi_{12} (\nabla_{\perp\mu} \phi \cdot \nabla^{\mu}_{\perp} \phi) (\nabla_{\lambda} u^{\lambda})$$

$$+ \xi_{13} (D\phi \cdot D\phi) (\nabla_{\lambda} u^{\lambda}) + \xi_{14} \sigma^{\mu\nu} (\nabla_{\perp\mu} \phi \cdot \nabla^{\mu}_{\perp} \phi),$$
(3.8)

⁴The notation of the first seven terms of $P_{(2)}$ and first eight terms of $t_{(2)}^{\mu\nu}$ are adopted from second order hydrodynamics constitutive relation of [50, 64, 184] where they write down the constitutive relation in terms of $\{u^{\mu}, \ln s\}$. We convert derivative of $\ln s$ into \mathcal{E} using the thermodynamics relation, $d\mathcal{E} = T ds$. The coefficient $\tilde{a} \equiv a/(sT)^2$ where $a = \xi_4, \lambda_4$ in [64, 184]

and

$$t_{(2)}^{\mu\nu} = \eta\tau_{\pi} \left[\left\langle D\sigma^{\mu\nu} + \frac{1}{2}\sigma^{\mu\nu}\nabla_{\lambda}u^{\lambda} \right] + \kappa \left[R^{\langle\mu\nu\rangle} - u_{\rho}u_{\sigma}R^{\rho\langle\mu\nu\rangle\sigma} \right] + \frac{1}{3}\eta\tau_{\pi}^{\star}\sigma^{\mu\nu}(\nabla_{\lambda}u^{\lambda}) + 2\kappa^{\star}u_{\rho}u_{\sigma}R^{\rho\langle\mu\nu\rangle\sigma} + \lambda_{1}\sigma^{\rho\langle\mu}\sigma^{\nu\rangle}_{\rho} + \lambda_{2}\sigma^{\rho\langle\mu}\Omega^{\nu\rangle}_{\rho} + \lambda_{3}\Omega^{\rho\langle\mu}\Omega^{\nu\rangle}_{\rho} + \tilde{\lambda}_{4}\nabla_{\perp}^{\langle\mu}\mathcal{E}\nabla_{\perp}^{\nu\rangle}\mathcal{E} + \lambda_{5}\sigma^{\mu\nu}(D\phi \cdot D\phi) + \lambda_{6}\Phi^{\mu\nu}(D\phi \cdot D\phi) + \lambda_{7}\sigma^{\mu\nu}(\nabla_{\perp}^{\lambda}\phi \cdot \nabla_{\perp\lambda}\phi) + \lambda_{8}\Phi^{\mu\nu}(\nabla_{\lambda}u^{\lambda}) + \lambda_{9}\Phi^{\mu\nu}_{ij}D\phi_{i}D\phi_{j} + \lambda_{10}\Phi^{\mu\nu}(\nabla_{\perp\lambda}\phi \cdot \nabla_{\perp}^{\mu}\phi) + \lambda_{11}\Phi^{\mu\nu}_{ij}\nabla_{\perp\lambda}\phi_{i}\nabla_{\perp}^{\lambda}\phi_{j}.$$
(3.9)

Similarly, the scalar fields expectation value $\langle O_i \rangle$ can be written in terms of linear combination of independent scalars with index *i* of the scalar fields, ϕ_i , namely

$$\langle \mathcal{O}_i \rangle = c_0 D \phi_i + c_1 (\nabla_\mu u^\mu) D \phi_i + c_2 (\nabla^\mu_\perp \mathcal{E}) \nabla_\mu \phi + c_3 (D \phi \cdot D \phi) D \phi_i + c_4 (\nabla_{\perp\mu} \phi \cdot \nabla^\mu_\perp \phi) D \phi_i + c_5 (D \phi \cdot \nabla_{\perp\mu} \phi) \nabla^\mu_\perp \phi_i + \mathcal{S}_i (\delta^{3/2}, \delta^2, \delta^{5/2}).$$

$$(3.10)$$

where S_i is a linear combination of scalar of order $\delta^{3/2}$, δ^2 , $\delta^{5/2}$ that transforms in the same way as \mathcal{O}_i . The explicit form of S_i is omitted as they are not relevant for the discussion in this work. In the holographic theory described by Einstein-Maxwell-scalar fields action in e.g.[180], the stress-energy tensor is traceless, $T^{\mu}_{\ \mu} = 0$. Such condition imposed on $t^{\mu\nu}$ implies that

$$\zeta = 0, \qquad \zeta_1 = 0, \qquad \zeta_2 = 0, \qquad P_{(2)} = 0.$$
 (3.11)

Note that, even if $T^{\mu}_{\ \mu} = 0$ resembles the conformal field theory, this theory is not conformal due to the presence of nonzero expectation value $\langle O_i \rangle$. Moreover, in the computation involving 2-point function, one can also perturb the fluid velocity as an additional small parameter. This allows one to ignore the term proportional to c_3 and terms with higher order of $D\phi$ in (3.7)-(3.10).

Before moving on, let us comments on the above form of $T^{\mu\nu}$ and \mathcal{O}_i , which are the result of the gradient expansions to the higher order while keeping the anisotropic scaling $\omega \sim m^2 \sim \delta$. The main reason which cause these expressions to be so lenghtly is the fact that that the tensors and scalars structures built from ∂u and ∂g at higher order in δ . Keeping the same scaling and going beyond order δ^2 is simply overkill since most of the terms in the expressions similar to those in (3.8)-(3.10) are not even entering the 2-point functions' computations. It would be interesting to find the constitutive relation for theory with anisotropic scaling $\omega \sim m^N \sim \delta$ where N is a big number. This way, the constitutive relation will be able to capture more terms due to scalar fields.

We end this section by commenting on the entropy current. Demanding that the entropy production is positive locally implies that some of the coefficients in $t^{\mu\nu}$ and \mathcal{O}_i are constrained [64, 186, 187]. In the case where the scalar field is not present, the entropy current is assumed to have the a canonical form [51]

$$TS^{\mu} = p \, u^{\mu} - T^{\mu\nu} u_{\nu} \tag{3.12}$$

which is reduced to the Smarr-like relation, $\epsilon + p = Ts$, when $u^{\mu} = \delta^{\mu t}$. Upon substituting the equation of motion and the constitutive relation for the conformal fluid at zero density, one will find that $\nabla_{\mu}S^{\mu} = \eta \sigma^{\mu\nu}\sigma_{\mu\nu} \ge 0$. Consequently, this inspired the origin of the bound on η to the minimum entropy production rate of the black hole [149, 150]. It turns out that the entropy production for the theory with broken translational symmetry is not as straightforward as in the standard conformal hydrodynamics. Let us demonstrate by consider the theory at order δ and assume that the entropy current take the canonical form(3.12), the entropy production rate contain three additional terms

$$T\nabla_{\mu}S^{\mu} = (sT - \mathcal{E} - \mathcal{P}) D \ln T - \langle \mathcal{O}_i \rangle D\phi_i + \eta_{\phi} \Phi^{\mu\nu} \sigma_{\mu\nu} + \eta \sigma^{\mu\nu} \sigma_{\mu\nu}$$
(3.13)

where we use the thermodynamics relation, dp = sdT to eliminate $\nabla_{\mu}p$. The first three terms vanish in the absence of the scalar field but it is not so straightforward to eliminate or rearrange them to the positive definite structures. To be more precise, let us expand \mathcal{O}_i at order $\delta^{3/2}$ (to make (3.3) consistent at order δ^3). One finds that

$$\langle \mathcal{O}_i \rangle D\phi_i = c_0 (D\phi \cdot D\phi) + c_1 (D\phi \cdot D\phi) \nabla_\mu u^\mu + c_2 (\nabla_{\perp\mu} \cdot D\phi_i) \nabla^\mu_\perp \mathcal{E} + c_3 (D\phi \cdot D\phi)^2 + c_4 (\nabla_{\perp\mu} \phi \cdot \nabla^\mu_\perp \phi) (D\phi \cdot D\phi) + c_5 (D\phi \cdot \nabla^\mu_\perp \phi) (D\phi \cdot \nabla_{\perp\mu} \phi).$$

$$(3.14)$$

It is likely that one can add vectors that vanish at equilibrium to the canonical entropy current (3.12) to eliminate terms that contains $D \ln T$, $\nabla_{\perp} \mathcal{E}$, $\nabla_{\mu} u^{\mu}$, $\sigma^{\mu\nu}$. However, we can see that the term proportional to the coefficients of c_0 , c_3 , c_4 , c_5 are already positive definite. Given a more complicated structure of the entropy current, it is possible that the entropy could also be produced by terms other than $\eta \sigma^{\mu\nu} \sigma_{\mu\nu}$. It would be very interesting to carefully analyse the entropy production in this type of models but we leave the complete analysis of the entropy current in the future work.

3.2.2 Kubo's formula for η^{\star}

In this section, we discuss the way to consistently extract the retarded Green's function. This method is slightly modified from variational method in [51] and is closely related to holographic computation. Extracting the Green's function in this way is also proven to be useful in deriving Kubo's formula for higher order hydrodynamics, see e.g. [99, 188]

The procedure for the variational method can be explained as the following. Firstly, one put the system in the manifold \mathcal{M} with metric $g_{\mu\nu}$ and background scalar fields ϕ_i . We write down these background fields as their equilibrium value + small perturbations, namely

$$g_{\mu\nu} = \eta_{\mu\nu} + h_{\mu\nu}, \qquad \phi_i = mx^i + \delta\phi_i \tag{3.15}$$

where $\{h_{\mu\nu}, \delta\phi_i\}$ are small perturbations. At the same time, we perturb the energy density \mathcal{E} and fluid velocity to linear order $\{\delta\mathcal{E}, \delta\rho, v^{\mu}\}$, which are also small perturbations. Then, we use the equation of motion (3.3) to solve for $\{\delta\mathcal{E}, \delta\rho, v^{\mu}\}$ in terms of $\{h_{\mu\nu}, a_{\mu}, \delta\phi_i\}$. After solving, substitute the solution for $\{\delta\mathcal{E}, \delta\rho, v^{\mu}\}$ into the constitutive relation (3.5).

We denoted the stress-energy tensor, where $\{\delta \mathcal{E}, \delta \rho, v^{\mu}\}$ are written in terms of $\{h_{\mu\nu}, a_{\mu}, \delta \phi_i\}$, as $\langle T^{\mu\nu} \rangle$. This is precisely the 1-point function from the field theory point of view. The retarded Green's function, G_{AB}^R of operator φ_A and φ_B where $\varphi_A = \{T^{\mu\nu}, J^{\mu}, \mathcal{O}_i\}, \varphi_B = \{h_{\mu\nu}, a_{\mu}, \delta \phi_i\}$ can be written as

$$G^{R}_{\mathcal{O}_{i}\mathcal{O}_{j}}(x) = -\frac{\delta\sqrt{-g}\langle\mathcal{O}_{i}(x)\rangle}{\delta\phi_{j}(0)}, \qquad (3.16)$$

$$G^{R}_{\mathcal{O}_{i}T^{\mu\nu}}(x) = -2\frac{\delta\sqrt{-g}\langle\mathcal{O}_{i}(x)\rangle}{\delta h_{\mu\nu}(0)}, \qquad (3.17)$$

$$G^{R}_{T^{\mu\nu}\mathcal{O}_{i}}(x) = -2\frac{\delta\sqrt{-g}\langle T^{\mu\nu}(x)\rangle}{\delta\phi_{i}(0)}, \qquad (3.18)$$

$$G^{R}_{T^{\sigma\rho}T^{\mu\nu}}(x) = -2\frac{\delta\sqrt{-g}\langle T^{\sigma\rho}(x)\rangle}{\delta T_{\mu\nu}(0)}, \qquad (3.19)$$

where all variations are performed with subsequent $\phi_i = h = 0$ insertion. Note that these 2-point functions are not entirely independent. They are related by the 2-point function's Ward's identity derived from (3.3).

To compute the shear viscosity, it is convenient to start from known result in translational invariant theory. In that case, the shear viscosity can be extracted from the retarded Greens' function of T^{xy} operator. Let us emphasize here again that, a priori, the relation between shear viscosity η and the 2-point functions is not necessary the same as in the usual hydrodynamics. For simplicity, we first study the perturbation that only depends on time. It turns out that one can bypass many steps in the above procedure as the stress-energy tensor δT^{xy} can be written in terms of the $\{h_{\mu\nu}, v^{\mu}, \delta\phi_i, \delta\mathcal{E}\}$ as

$$\delta T^{xy} = \frac{1}{2} \mathcal{P} h_{xy} + \frac{1}{2} \eta_{\phi} m^2 h_{xy} - \frac{1}{2} (\eta - m^2 \lambda_7) \partial_t h_{xy} + \mathcal{O}(h^2)$$
(3.20)

where $\mathcal{O}(h^2)$ denotes the terms that are products of perturbations $\{h_{\mu\nu}, v^{\mu}, \delta\phi_i, \delta\mathcal{E}\}$. We can see that this component of the stress-energy tensor is independent of the primary variables i.e. $\{v^{\mu}, \delta\mathcal{E}\}$. Thus, by Fourier transform $h_{xy}(t) \sim$

 $\int d\omega \, e^{i\omega t} h_{xy}(\omega)$, we immediately arrive at the 2-point function for $G^R_{T^{xy}T^{xy}}$,

$$G^{R}_{T^{xy}T^{xy}} = \left(\mathcal{P} + \eta_{\phi}m^{2}\right) - i\omega\left(\eta - m^{2}\lambda_{7}\right) + \mathcal{O}(h^{2}),$$

$$\Rightarrow \quad \eta^{\star} = \eta - \lambda_{7}m^{2}$$
(3.21)

This implies that $-\omega^{-1} \text{Im} G^R_{T^{xy}T^{xy}}$ are polluted by the terms proportional to m^2 and, unless one only consider $T^{\mu\nu}$ at order δ^1 , the above Kubo formula is not the same as η in the constitutive relation. Note also that

$$\eta^{\star} = -\lim_{\omega \to 0} (1/\omega) \operatorname{Im} G^R_{T^{xy}T^{xy}}$$
(3.22)

is also bound from below at zero, for $\omega \ge 0$ because of the Hermitian property of T^{xy} . The relation between this lower bound of η^* and the entropy production is still unclear at this stage.

3.3 Holographic computation

If we use the effective "hydrodynamics" framework outlined in section 3.2 as a basis to define transport (or hydrodynamic) coefficients in arbitrary systems, it is then natural to expect that η and η^* are not identical evan at the leading order in δ expansion. However, from the hydrodynamics point of view, we do not known whether the quantites η/s and η^*/s violate the KSS bound or not. Moreover, as the coefficient λ_7 and possible higher order corrections are yet to be determined, we do not have an insight of how η and η^* are different before computing them explicitly.

To investigate these problems, we compute both η/s and η^*/s in a simple holographic model and shows that both of them violate the KSS bound. The ratio of η/s can be computed analytically using the results from fluid/gravity from [178]. The ratio η^*/s can also be computed analytically at small m and ω and are found be identical to η/s at the same order of m. Beyond the leading order, they start to deviate from each other.

To perform a holographic calculation of the shear viscosity and other thermodynamic quantities, we use a 3 + 1 dimensional Einstein-Maxwell-Scalar

action with a charged black brane solution ansatz. The scalar fields are assumed to have a fixed profile that explicitly breaks the translational symmetry. Thermodynamic quantities of the black hole are identified with those of the corresponding fluid. In Section 3.3.1, we specify the model and compute thermodynamic quantities. The fluid/gravity calculations are discussed in Section 3.3.2, demonstrating the violation of the KSS bound. Section 3.3.3 shows the perturbative calculation of the shear viscosity/entropy density ratio by the Kubo's formula method. The results of Section 3.3.2 and 3.3.3 shows that the η/s and η^*/s are not identical even at small m, as expected. Numerical calculations of η^*/s are in Section 3.3.4. Notably, Fig. 3.1 shows that the values of shear viscosity/entropy density ratio calculated by the two methods deviate more from one another as m increases.

3.3.1 Action and Thermodynamics

Let us start by specifying the action for the holographic model where the translational symmetry of the boundary theory is broken by the massless bulk scalar fields

$$S = \int_{M} d^{d+1}x \sqrt{-g} \left(R - 2\Lambda - \frac{1}{2} \sum_{i=1}^{d-1} (\partial \phi_i)^2 - \frac{1}{4} F^2 \right) + S_{\text{bnd}}$$
(3.23)

with appropriate boundary and counter terms S_{bnd} . This action exhibits a simple planar charged black hole solution where the translational symmetry of the boundary theory is broken explicitly by the scalar fields. For this solution, the background metric, gauge field and scalar fields can be written as the follow-

Chapter 3. Shear viscosity in holography and effective theory of transport without translational symmetry

ing [180]

$$ds^{2} = -r^{2}f(r)dt^{2} + r^{2}dx_{i}dx^{i} + \frac{dr^{2}}{r^{2}f(r)}, \quad A = A_{t}(r)dt, \quad \phi_{i} = mx^{i},$$

$$f(r) = 1 - \frac{m^{2}}{2(d-2)r^{2}} - \left(1 - \frac{m^{2}}{2(d-2)r_{h}^{2}} + \frac{(d-2)\mu^{2}}{2(d-1)r_{h}^{2}}\right)\left(\frac{r_{h}}{r}\right)^{d} + \frac{(d-2)\mu^{2}}{2(d-1)r_{h}^{2}}\left(\frac{r_{h}}{r}\right)^{2(d-1)},$$

$$A_{t} = \mu\left(1 - \left(\frac{r_{h}}{r}\right)^{d-2}\right),$$
(3.24)

where i = 1, 2, ..., d - 1. We denote the chemical potential by μ . For concreteness, we will focus on the theory with d = 3, which is an arena for many condensed matter systems. The temperature, entropy density, energy density and charge density can be written as

$$T = \frac{r_h}{4\pi} \left(3 - \frac{m^2}{2r_h^2} - \frac{\mu^2}{4r_h^2} \right), \quad s = 4\pi r_h^2,$$

$$\epsilon = 2r_h^3 \left(1 - \frac{m^2}{2r_h^2} + \frac{\mu^2}{4r_h^2} \right), \quad \rho = \mu r_h.$$
(3.25)

Finally, the pressure can be computed using the renormalised Euclidean action [180].

$$p = \langle T^{xx} \rangle + m^2 r_h = \frac{\epsilon}{2} + m^2 r_h = sT + \mu \rho - \epsilon.$$
(3.26)

As mentioned earlier, the pressure here is not the same as the expectation value $\langle T^{ii} \rangle$.

In [65], the value of parameter m is restricted to be $0 < m < r_h\sqrt{6}$ so that the temperature remains non-negative for $\mu = 0$. Once the density of turned on, the allowed range of m becomes $0 < m < \sqrt{6r_h^2 - \mu^2/2}$.

3.3.2 Coherent regime and constitutive relation from fluid/gravity correspondence

The background parametrisation where we keep the entropy density fixed is suitable to find the numerical solution. However, it is more convenient to fix the energy density in order to compare with the result from fluid/gravity [177, 178] and the constitutive relation constructed in section 3.2.1.

We will work on zero density case for simplicity. It is also convenient to introduce a scale r_0 related to the energy density as $\epsilon = 2r_0^3$. In the absence of the scalar field, the position of the horizon in the gravity dual theory is precisely $r_h = r_0$. The relation between r_0 and r_h can be found by the following relation [178]

$$0 = 1 - \left(\frac{r_0}{r_h}\right)^3 - \frac{m^2}{2r_h^2}.$$
(3.27)

This relation can be found by equating the energy density where $m = 0, r = r_0$ and the case where m is nonzero given in Eqn. (3.25). The coefficients in the constitutive relation of $T^{\mu\nu}$ for theory with zero density were found using the fluid/gravity computation [178], where $T^{\mu\nu}$ is expanded up to order δ in the anisotropic scaling (3.4), to be

$$\mathcal{E} = 2r_0^3, \qquad \mathcal{P} = r_0^3, \qquad \eta = r_0^2, \qquad \eta_\phi = r_0.$$
 (3.28)

Interestingly, if one fix the energy density and start to slightly break the translational symmetry, the shear viscosity remains unchanged. Now, the entropy density can be found, in terms of r_0 , using (3.25) and (3.27) as

$$s = 4\pi r_h^2 = 4\pi \left(r_0^2 + \frac{m^2}{3} + \mathcal{O}(m^4) \right).$$
(3.29)

Note that the full expression of r_h is given by

$$r_h = \frac{\left(\sqrt{6}\sqrt{54r_0^6 - m^6} + 18r_0^3\right)^{2/3} + 6^{1/3}m^2}{6^{2/3}(\sqrt{6}\sqrt{54r_0^6 - m^6} + 18r_0^3)^{1/3}}.$$
(3.30)

This immediately implies the violation of the KSS bound [30] as

$$\frac{\eta}{s} = \frac{1}{4\pi} \left(1 - \frac{1}{3} \left(\frac{m}{r_0} \right)^2 + \mathcal{O}(m^4) \right), \qquad r_h = r_0 + \frac{m^2}{6r_0} + \mathcal{O}(m^4).$$
(3.31)

For completeness, we write down the coefficients c_i in the constitutive relation of $\langle O_i \rangle$ obtained from fluid/gravity [178] i.e.

$$c_0 = -r_0^2$$
, $c_1 = r_0(1-\lambda)$, $c_2 = -\frac{(1+\lambda)}{2r_0^3}$, $c_4 = -\frac{1}{6}$, $c_5 = \frac{2}{3}$.
(3.32)

where λ can be found analytically for $\mu = 0$ to be

$$\lambda = -\frac{1}{2} \left(\frac{\pi}{3\sqrt{3}} - \log 3 \right). \tag{3.33}$$

The coefficient c_3 is not specified as it depends on $(D\phi)^3$ and is subleading in the expansions $u^{\mu} = \delta^{0\mu} + v^{\mu}$ mentioned in section 3.2.1. It is interesting to observe that the value of $-2\lambda = \pi/3\sqrt{3} - \ln 3$ is identical to the coefficient of m^2 in Eqn. (3.48) of η^*/s calculated to $\omega m^2 \sim \delta^2$ order. Incidentally, λ appears in the two terms of order δ^2 in Eqn. (3.10) of $\langle \mathcal{O}_i \rangle$. It is possible that this is not a coincidence and the two quantities are actually the same.

We will not discuss the details of the transport coefficient at finite density, $\rho \neq 0$, but would like to mention that the relation between r_h and r_0 in that case can be found by solving

$$0 = 1 - \left(\frac{r_0}{r_h}\right)^3 - \frac{m^2}{2r_h^2} + \frac{\rho^2}{4r_h^4}.$$
(3.34)

The ratio between the entropies when m = 0 and nonzero value of m at the fixed energy density, in this case, at the leading order, is found to be

$$\frac{\eta}{s} = \frac{1}{4\pi} \left(1 - \frac{(2m/r_0)^2}{12 - \rho^2} \right) + \text{higher order terms.}$$
(3.35)

The above relation indicates that the shear viscosity/entropy density decreases more rapidly with the density.

3.3.3 Fluctuations and violation of the viscosity bound at leading order

Let us focus on the computation in the asymptotic AdS_4 space. We will choose the direction of the metric fluctuations to propagate in the x direction, i.e. $\vec{k} \cdot \hat{x} = k$ and consider the shear viscosity with respect to the perpendicular directions. In asymptotic AdS_4 , the metric fluctuation can be split into those with odd and even parity under $y \leftrightarrow -y$. We are interested in odd parity modes namely $\{h_x^y, h_r^y, h_t^y\}$. In the presence of the two massless scalar fields, ϕ_1, ϕ_2 , in AdS_4 , only the fluctuation $\delta\phi_2$ couples to the odd parity channel. The full equations of motion of the relevant modes are

$$\frac{d}{dr}\left[r^4f(h_x^{y\prime}-ikh_r^y)\right] + \frac{\omega}{f}(\omega h_x^y + kh_t^y) - m^2h_x^y + ikm\delta\phi_2 = 0,$$

$$\frac{d}{dr}\left[r^4(h_t^{y\prime} + i\omega h_r^y)\right] - \frac{k}{f}(\omega h_x^y + kh_t^y) - \frac{m^2}{f}h_t^y - \frac{i\omega m}{f}\delta\phi_2 + r^2a_y'A_t' = 0,$$

$$\frac{d}{dr}\left[r^4f(\delta\phi_2'-mh_r^y)\right] + \frac{1}{f}(\omega^2-k^2f)\delta\phi_2 - \frac{m}{f}(i\omega h_t^y+ikfh_x^y) = 0,$$

$$i\omega h_t^{y'} + ikfh_x^{y'} - (\omega^2 - m^2 f - k^2 f)h_r^y - mf\delta\phi_2' + \frac{i\omega}{r^2}a_y A_t' = 0.$$

The combination of the first and the third equations gives

$$\frac{d}{dr}\left(r^4f\Psi'\right) + \frac{\omega^2 - (k^2 + m^2)f}{f}\Psi = 0$$
(3.36)

where $\Psi(r) = \Psi_y \equiv h_x^y - i(k/m)\delta\phi_2$. The scalar field generates mass term for the metric perturbation h_x^y proportional to its profile parameter m^2 . It also breaks the translational invariance with respect to the infinitesimal shift in ydirection.

To find the shear viscosity, we study the near boundary behaviour of $\Psi(r) = \Psi^{(0)} + r^{-3}\Psi^{(3)}$, which is equivalent to $h_x^y(r) = h_x^{y(0)} + r^{-3}h_x^{y(3)}$ in $k \to 0$ limit. Plugging this into the onshell action [65]

$$S = \int \frac{d\omega dk}{(2\pi)^2} \frac{3}{2(k^2 + m^2 - \omega^2)} \bigg[h_x^{y(0)} \bigg\{ (m^2 - \omega^2) h_x^{y(3)} - imk \delta \phi_2^{(3)} \bigg\} \\ + \delta \phi_2^{(0)} \bigg\{ imk h_x^{y(3)} + (k^2 - \omega^2) \delta \phi_2^{(3)} \bigg\} \bigg]$$

and then apply the formula for the "shear viscosity" i.e.

$$\eta^* \equiv -\lim_{\omega \to 0} \frac{1}{\omega} \operatorname{Im} G^R_{T^{xy}T^{xy}}(\omega, k=0) = \frac{3}{\omega} \operatorname{Im} \left(\frac{\Psi^{(3)}}{\Psi^{(0)}} \right) \bigg|_{\omega \to 0}.$$
 (3.37)

.

The equation of motion (3.36) can be solved analytically for small ω , m limit. However, for the large m limit, one is required to solve it numerically. The numerical procedure to find η^* is straightforward as one only need to impose the ingoing boundary condition to in the region region close to the horizon, namely

$$\Psi_{\text{inner}} = \alpha_{+} f(z)^{\left[-i\omega/(3-\frac{m^{2}}{2}-\frac{\mu^{2}}{4})\right]} \left(1+a(1-z)+b(1-z)^{2}+c(1-z)^{3}\right),$$

where we define the new coordinate to be $z = r_h/r$. We present the numerical results in Section 3.3.4.

Let us proceed by solving (3.36) analytically at the leading order in m^2 . In the following calculation, the dimensionful parameters, ω, m, μ are rescaled by the horizon radius r_h to make them dimensionless. For simplicity, let us focus on the case where $\mu = 0, k = 0$. The gauge invariant field Ψ is assumed, consistently, to have the following expansion in m^2

$$\Psi = f(z)^{i\omega/f'(1)}S(z),$$

$$S(z) = A(z) + m^2B(z) + \mathcal{O}(m^4),$$
(3.38)

where at each m order we expand with respect to ω ,

$$A(z) = A_0(z) + \omega A_1(z) + \omega^2 A_2(z) + \mathcal{O}(\omega^3), \qquad (3.39)$$

$$B(z) = B_0(z) + \omega B_1(z) + \omega^2 B_2(z) + \mathcal{O}(\omega^3).$$
(3.40)

The equation of motion at $\mathcal{O}(m^0)$ order after substituting (3.38) into Eqn. (3.36) when $k\to 0$ is

$$0 = A''(z) - \frac{2 + (1 - 2i\omega)z^3}{z(1 - z^3)}A'(z) + \frac{\omega^2(1 + z + z^2 + z^3)}{(1 - z)(1 + z + z^2)^2}A(z).$$

This equation can be solved perturbatively by substituting (3.39) and solve order

by order in ω . Once we obtain the solution satisfying the appropriate boundary condition, it can be used to solve for the solution at the higher order in m.

The equation of motion at $\mathcal{O}(m^2)$ order (the coefficient of m^2 in (3.36)) in $k\to 0$ limit is given by

$$0 = \frac{z \left(4i\omega + 2i\omega z^3 + 3z^2 + 3z - 6\right) A'(z)}{6(1-z) \left(z^2 + z + 1\right)^2} + \frac{g(z)A(z)}{3(1-z) \left(z^2 + z + 1\right)^3} + B''(z) - \frac{\left(2 + (1-2i\omega)z^3\right) B'(z)}{z \left(1-z^3\right)} + \frac{\omega^2 \left(z^3 + z^2 + z + 1\right) B(z)}{\left(1-z\right) \left(z^2 + z + 1\right)^2},$$
(3.41)

where

$$g(z) \equiv \left(-i\omega + \omega^2 z^5 + (\omega^2 - i\omega - 3) z^4 + (\omega^2 - 2i\omega - 6) z^3 + 3(\omega^2 - i\omega - 3) z^2 + (-6 - 2i\omega)z - 3 \right).$$
(3.42)

The boundary conditions of $A_0(z), A_1(z), A_2(z)$ are set as the following

$$A_0(0) = 1, |A_0(1)| < \infty; A_1(z = 0, 1) = A_2(z = 0, 1) = 0.$$
 (3.43)

We can solve to obtain $A_0(z) = 1$, $A_1(z) = 0$ so that $A(z) = 1 + \omega^2 A_2(z)$. The full expression of $A_2(z)$ is lengthy but since we are interested in its behaviour near z = 0, we can Taylor expand A(z) giving

$$A(z) = 1 + \omega^2 \left(\frac{z^2}{2} - \frac{z^3}{54} (18 + \sqrt{3}\pi - 9\ln 3) \right) + \mathcal{O}(z^4). \quad (3.44)$$

The function B(z) can also be straightforwardly solved in a perturbative way by substituting A(z) into (3.41) and solve order by order in ω . Requiring the boundary condition $B_0(0) = 0$, $|B_0(1)| < \infty$, the leading order solution is

$$B_0(z) = \frac{1}{\sqrt{3}} \left[\arctan\left(\frac{1+2z}{\sqrt{3}}\right) - \frac{\pi}{6} \right] - \ln\left(\sqrt{\frac{3}{4} + (\frac{1}{2}+z)^2}\right).$$
(3.45)

The resulting functional form is a lengthy expression satisfying boundary conditionnext to leading order soluton, B_1 , can be obtained in a similar way by requiring $B_1(z = 0) = B_1(z = 1) = 0$. Again, since we are interested in the

behaviour of B(z) near z = 0, we can Taylor expand to get

$$B(z) = -\frac{1}{6}(3+i\omega)z^2 + \frac{z^3}{3}\left(1 + \frac{i\omega}{9}(3+\sqrt{3}\pi - 9\ln 3)\right) + \mathcal{O}(z^4).$$
 (3.46)

The perturbative solution is thus

$$\Psi(z) = 1 - \frac{z^2}{6} \left(3(m^2 - \omega^2) + \frac{i\omega m^4}{m^2 - 6} \right) + z^3 \left(\frac{i\omega (m^2 - 2)}{m^2 - 6} + \frac{m^2}{27} [9 + i\omega (3 + \sqrt{3}\pi - 9\ln 3)] - \frac{\omega^2}{54} (18 + \sqrt{3}\pi - 9\ln 3) \right) + \mathcal{O}(z^4).$$
(3.47)

Then the shear viscosity can be calculated by the usual relation

$$\eta^{\star} = \lim_{\omega \to 0} \frac{3}{\omega} \operatorname{Im}\left(\frac{\Psi^{(3)}(0)}{\Psi^{(0)}(0)}\right) \simeq 1 - m^2 \left(\ln 3 - \frac{\pi}{3\sqrt{3}}\right), \qquad (3.48)$$

where we expand $\Psi=\Psi^{(0)}+\Psi^{(1)}z+\Psi^{(2)}z^2+\Psi^{(3)}z^3+....$

Interestingly, the coefficient of m^2 , $\pi/3\sqrt{3} - \ln 3$, is identical to the value of -2λ in (3.33) calculated from the fluid/gravity approach. We speculate that the two quantities could actually be related despite being at different order in the derivative expansion. ⁵

3.3.4 Numerical results and beyond the leading order

In this section, we solve the equation for Ψ numerically with fixed $r_h = 1$, using the procedures outlined in the previous section. The purpose of these numerical computation is two-fold. First of all, we would like to check the validity of the analytic computation and the prediction from fluid/gravity when the disorder strength is small. Secondly, it would be interesting to see the pattern of how the retarded correlation G_{TxyTxy}^R behave at higher order. The main point of the latter part is to emphasize that, when the higher order in δ is included, the quantity $\eta^* = -\omega^{-1} \text{Im} G_{\Psi\Psi}^R|_{\omega \to 0}$ is *not* the value of η in the constitutive relation. This is due to the fact that the 2-point function is polluted by the term

⁵Note added : We would like to mention that the expression for η^* / s here agrees with those presented in [40, 185].
of the form (scalars) $\sigma^{\mu\nu}$ e.g. $\lambda_7 \sigma^{\mu\nu} (\nabla_{\perp} \phi)^2$ in (3.8) and (3.8).



Figure 3.1. Numerical value of viscosity ratio $4\pi\eta^*/s$ at zero chemical potential compared with $4\pi\eta/s$ in the fluid/gravity calculation as a function of m/T. The dotted curve is the ratio $4\pi\eta^*/s$ computed using Kubo's formula for η^* as described in section 3.3.3. The solid curve (fluid/gravity) is computed from η/s where $s = 4\pi r_h^2$ and r_h is given by the full expression in (3.30). We refer to this curve as entropy ratio since the value of η is proportional to the entropy density when m = 0 with the same energy density. It is clear that there is a large deviation between the numerical η^* and the fluid/gravity η .

In figure 3.1, we demonstrate that both η/s and η^*/s violate the KSS bound. The violation of KSS bound for η/s can be understood as η is only sensitive to r_0 as we pointed out in section 3.3.2. On the other hand, the violation of η^*/s comes from the change in entropy and the higher order terms in δ expansion. Interestingly, our numerical result indicates that the differences $\eta - \eta^*$ is monotonically increasing as m/T grows.

We can also consider what happens in the finite chemical potential case. In figure 3.2, we can see that the ratio η^*/s violate the bound for even small value of m. The numerical value of η^*/s decrease more rapidly as one increase the chemical potential. Although we don't have an analytic expression to see the explicit μ/r_h dependence, this feature can already be observed at a small value of m. In the regime where the difference between η^* and η is small, the above feature agrees with the prediction from (3.35).

A simple Mathematica code used to produced plots in this section is available upon request.

Chapter 3. Shear viscosity in holography and effective theory of transport without translational symmetry



Figure 3.2. The numerical profile of $4\pi\eta^*/s$ with respect to the m/r_h at various μ/r_h , where $\eta^* = -\omega^{-1} \text{Im} G^R_{\Psi\Psi}|_{\omega \to 0}$ for different chemical potentials. Each curve truncates at zero temperature where $m/r_h = \sqrt{6 - \mu^2/2r_h^2}$.

3.4 Discussions and outlook

We follow up on the insight from [177, 178], which suggest that coupled the fluid to the background spatially dependent scalar fields ϕ_i is an accurate and consistent framework to study the hydrodynamics behaviour of the theory with broken translational symmetry. We construct the constitutive relation to order δ^2 and shows that the standard hydrodynamic formula we used to extract the usual shear viscosity, η , is no longer applicable when the scalar fields are included in the constitutive relation. With the modified constitutive relation, we speculate that the shear viscosity may not be the only channel to produce the entropy. However, the correct form of the entropy current has yet to be found. Thus, our constitutive relation should be considered as the worse case scenario, where no hydrodynamics coefficient is constrained by the positivity of local entropy production and we cannot make a clear statement on the minimum entropy production conjecture of [149, 150]. It would be very interesting to make the entropy production rate argument more precise in this class of theories and study the manifestation of the minimum entropy production conjecture in this class of theory, particularly, possible connection between the conjecture and the universal bound in disordered systems [36, 38, 39].

Regarding the holographic computation, we have analytically and numer-

ically computed the "shear viscosity" per entropy density ratio, η^*/s , in the finite-density holographic models with translational symmetry breaking for an asymptotically AdS_4 spacetime. The analytic computation has been done using a perturbative method order by order in m^2 and ω . The ratio is found to violate the KSS bound $\eta/s = 1/4\pi$ for arbitrary translational symmetry breaking parameter m. In 4 (d = 3) dimensions for small m, the ratio is

$$\frac{4\pi\eta^*}{s} \simeq 1 - \frac{m^2}{r_h^2} \left(\log 3 - \frac{\pi}{3\sqrt{3}} \right) + \mathcal{O}(m^4).$$

At larger m, the deviation of η^*/s and η/s grows as we can see from Fig. 3.1. Incidentally, the difference $\eta - \eta^*$ is monotonically increasing. As we saw that the difference is caused by the higher order terms e.g. λ_7 , it would be interesting to understand whether the coefficient λ_7 and other terms participate in η^* are constrained by some underlying principles or not.

A simple explanation of the violation of KSS bound is the entropy contribution from the scalar fields. In the presence of the translational symmetry breaking scalar field profile, the entropy is increased as we can see from the enlarged horizon in Eqn. (3.29). On the other hand, the shear viscosity remains insensitive to m at the leading order. The η/s ratio thus becomes smaller than the KSS bound for any m. Remarkably, the violation persists even in the zero temperature limit the degree of violation depends on the chemical potential μ through dependency on m. Inspired by the viscosity bound violation, it is interesting to investigate other hydrodynamic bounds in the translational symmetry breaking axion-gravity model. First, let us consider the sound speed bound $c_s^2 \leq 1/2$ [148]. From Eqn. (3.25), we might think that the sound speed c_s should be calculated from $p = m^2 r_0 + \epsilon/2$ by the quantity $(\partial p/\partial \epsilon)$. But if we choose to fix m, μ

$$\frac{\partial p}{\partial \epsilon}\Big|_{m,\mu} = \frac{1}{2} + m^2 \frac{\partial r_0}{\partial \epsilon}\Big|_{m,\mu} = \frac{1}{2} + \frac{2m^2}{\mu^2 + 2(6r_0^2 - m^2)} \ge \frac{1}{2}, \tag{3.49}$$

For m = 0, this quantity saturates the bound $(\partial p / \partial \epsilon) \le 1/2$. However, when m is turned on, the above definition of the speed of sound violates the sound-speed bound. A more consistent candidate for c_s^2 is the quantity $(\partial P / \partial \mathcal{E})$ as

Chapter 3. Shear viscosity in holography and effective theory of transport without translational symmetry

the modified constitutive relation has the following sound pole

$$\omega^2 - \left(\frac{\partial \mathcal{P}}{\partial \mathcal{E}}\right)\Big|_{\mu,\mu} k^2 + \dots = 0, \qquad (3.50)$$

instead of the physical pressure p in the standard hydrodynamics. Using (3.28), the speed of sound bound is trivially satisfied.

$$c_s^2 \equiv \frac{\partial \mathcal{P}}{\partial \mathcal{E}} = \frac{1}{2},\tag{3.51}$$

saturating the sound-speed bound regardless of the translational symmetry breaking. The other interesting bound related to the sound speed is the bulk viscosity bound [32] for d = 3,

$$\frac{\zeta}{\eta} \ge 2\left(\frac{1}{2} - c_s^2\right). \tag{3.52}$$

Since in our model the fluid is traceless so that the bulk viscosity $\zeta = 0$ [180], the bulk viscosity bound is trivially saturated.

One obvious next goal is also to find an effective hydrodynamic framework for a theory with strong disorder. As we also mentioned earlier, the main obstacle for the current framework is due to the complexity when one includes higher order terms in gradient expansions. It would be interesting to find a constituent way to incorporate terms higher order in $\nabla \phi_i$ without including higher order hydrodynamic terms containing ∂u and ∂g . In fact, the formalism to extract DC conductivities from forced Navier-Stokes equation has been recently developed in [129, 175, 176] without invoking the derivative expansions. The connection between this method and the one studied in this work has been discussed in [178]. It would be interesting to see how robust the connection between the two frameworks is when one includes higher order terms in $\nabla \phi$. Chapter 3. Shear viscosity in holography and effective theory of transport without translational symmetry

3.5 Appendices

3.5.1 Scalars, vectors and tensors from basic structures

The constitutive relation of the "hydrodynamics" effective theory in this work are constructed from the following local macroscopic variables $\mathcal{E}(x)$, $u^{\mu}(x)$ and the background fields $g_{\mu\nu}(x)$, $\phi_i(x)$ For simplicity, let us work on zero density. To find the structures that enter the constitutive relation, we organise the scalar, vector and tensor at each order in the expansion in δ .

• Structures of order δ^0 : For the system where the low energy limit is homogenous, as considered in this work, the zeroth order term cannot explicitly contain the scalar field $\phi_i = mx^i$. The objects at this order are

Scalar :
$$\mathcal{E}(x)$$

Vector : $u^{\mu}(x)$ (3.53)
Tensor : $u^{\mu}u^{\nu}, \Delta^{\mu\nu}$

The projector, $\Delta^{\mu\nu} = g^{\mu\nu} + u^{\mu}u^{\nu}$ is orthogonal to the 4-velocity i.e. $\Delta_{\mu\nu}u^{\mu} = 0.$

• Structures of order $\delta^{1/2}$: Terms at this order can only be linear in the derivative of ϕ_i as the expansion in δ is organised using anisotropic sclaing

Scalar :
$$D\phi_i$$

Vector : $u^{\mu}D\phi_i, \nabla^{\mu}_{\perp}\phi_i$ (3.54)

where we introduce the notation for the directional derivative along the direction of the 4-velocity as $D = u^{\mu} \nabla_{\mu}$ and the derivative perpendicular to u^{μ} as $\nabla^{\mu}_{\perp} = \Delta^{\mu\nu} \nabla_{\nu}$

• Structures of order δ^1 : The basic structure at this order can be constructed from $\nabla \mathcal{E}, \nabla u$ and $(\nabla \phi_i)^2$. We only construct the tensors orthogonal to u^{μ} the Landau frame $u^{\mu}t_{\mu\nu}$ is chosen. Combining these objects together, we obtain

Scalar :
$$\nabla_{\mu}u^{\mu}, (D\phi_i)(D\phi_j), \nabla^{\perp}_{\mu}\phi_i\nabla^{\mu}_{\perp}\phi_j$$

Vector : $u^{\mu}D\phi_i D\phi_j, \nabla^{\mu}_{\perp}\mathcal{E}, \nabla^{\mu}_{\perp}\phi_i(D\phi_j),$ (3.55)
Tensor : $\sigma^{\mu\nu}, \Phi^{\mu\nu}_{ij}$

where $\sigma^{\mu\nu}$ and $\Phi^{\mu\nu}_{ij}$ are defined as

$$\sigma^{\mu\nu} = 2\Delta^{\mu\alpha}\Delta^{\nu\beta}\nabla_{(\alpha}u_{\beta)} - \Delta^{\mu\nu}(\nabla_{\lambda}u^{\lambda}),$$

$$\Phi^{\mu\nu}_{ij} = \nabla^{\mu}_{\perp}\phi_i\nabla^{\nu}_{\perp}\phi_j - \frac{1}{2}\Delta^{\mu\nu}(\nabla_{\perp\lambda}\phi_i\nabla^{\lambda}_{\perp}\phi_j)$$
(3.56)

The trace of tensor $\Phi_{ij}^{\mu\nu}$ over the index i, j is denoted by $\Phi^{\mu\nu} = \sum_{i=1}^{3} \Phi_{ii}^{\mu\nu}$. To avoid the cluttering of indices, we denote, $\phi \cdot \phi = \sum_{i} \phi_{i} \phi_{i}$ and $\Phi_{ij} \phi_{i} \phi_{j} = \sum_{i,j} \Phi_{ij} \phi_{i} \phi_{j}$. Note also that, the divergent of the fluid velocity $\nabla_{\mu} u^{\mu}$ is equivalent to $\nabla_{\perp\mu} u^{\mu}$ since $u_{\mu} D u^{\mu} = 0$.

• Structures of order $\delta^{3/2}$: Only relevant part in the constitutive relation that requires structure at this order is $\langle \mathcal{O}_i \rangle$. Thus, we need to construct scalar objects under spacetime transformation which contain the index iof the scalar fields ϕ_i . All possible combination of objects that satisfy the above requirements are listed below

mixed term :
$$(\nabla_{\mu}u^{\mu})D\phi, \nabla_{\perp\mu}\phi_i\nabla^{\mu}_{\perp}\mathcal{E},$$

pure ϕ_i terms : $D\phi_i(D\phi_jD\phi_j), (D\phi_i)(\Delta^{\mu\nu}\nabla_{\mu}\phi_j\nabla_{\nu}\phi_j),$ (3.57)
 $(D\phi_j)(\Delta^{\mu\nu}\nabla_{\mu}\phi_i\nabla_{\nu}\phi_j)$

4

Universality of anomalous conductivities in theories with higher-derivative holographic duals

4.1 More background materials and motivations

Anomalies

An anomaly is a quantum effect whereby a classically conserved current J^{μ} ceases to enjoy its conservation, $\nabla_{\mu}\langle J^{\mu}\rangle \neq 0$ [72–74, 189]. To date, a multitude of different anomalies have been discovered that can be classified into two main categories: local (gauge) and global anomalies. A gauge anomaly corresponds to a gauged symmetry (and current) and the consistency of a quantum field theory requires this anomaly to vanish. While global anomalies are permitted, their existence still imposes stringent conditions on the structure of quantum field

theories due to the anomaly matching condition discovered by 't Hooft [190]. The condition states that a result of an anomaly calculation must be invariant under the renormalisation group flow and is thus independent of whether it is computed in the UV microscopic theory or an IR effective theory.

Of particular importance to quantum field theory have been the chiral anomalies, which are present in theories with massless fermions. The values of the current divergences resulting from these anomalies are known to be one-loop exact. From the point of view of the topological structure of gauge theories, one can suspect that this should be true very generically due to the fact that the anomaly is related to the topologically protected index of the Dirac operator. Perturbatively, non-renormalisation of the one-loop anomalies was established in [76–78]. In a typical four dimensional chiral theory, there are two classically conserved currents: the axial J_5^{μ} (associated with the γ_5 Dirac matrix) and the vector current J^{μ} . By including quantum corrections, their Ward identities can be written as

$$\nabla_{\mu} \langle J_{5}^{\mu} \rangle = \epsilon^{\mu\nu\rho\sigma} \left(\kappa F_{A,\mu\nu} F_{A,\rho\sigma} + \gamma F_{V,\mu\nu} F_{V,\rho\sigma} + \lambda R^{\alpha_{1}}_{\ \alpha_{2}\mu\nu} R^{\alpha_{2}}_{\ \alpha_{1}\rho\sigma} \right),$$

$$\nabla_{\mu} \langle J^{\mu} \rangle = 0,$$
(4.1)

where $F_{A,\mu\nu}$, $F_{V,\mu\nu}$ are the field strengths associated with the axial and the vector gauge fields. $R^{\alpha}_{\ \beta\mu\nu}$ is the Riemann curvature tensor of the curved manifold on which the four dimensional field theory is defined, and κ , γ and λ are the three Chern-Simons coupling constants. While the axial current conservation is violated by quantum effects, the vector current remains conserved. Among other works, various arguments in favour of non-renormalisation of one-loop anomalies have been presented in [62, 80, 83–85, 191–193]. The situation is much less clear when, as in [194], one considers the contributions of mixed, gauge-global anomalies. In such cases, it was shown in [194] that one should expect anomalous currents to receive radiative corrections at higher loops. The connection between this work and mixed, gauge-global anomalies will be elaborated upon below. A further set of open questions related to the non-renormalisation of anomalies enters the stage from the possibility of considering non-perturbative effects in QFT. From a historically more unconventional point of view, anomalies have recently also been studied through the (macroscopic) hydrodynamic entropy current analysis [195, 196].¹ The effects of gravitational anomalies on the hydrodynamic gradient expansion were then studied by using the Euclidean partition function on a cone in [199]. Macroscopic transport properties associated with anomalous conservation laws have now been analysed in detail (at least theoretically) both at non-zero temperature and density. To date, the most prominent and well-understood anomaly-induced transport phenomena have been associated with the chiral magnetic effect [80, 84, 200] and the chiral vortical effect [195, 201].

Chiral conductivities in field theory

In the low-energy hydrodynamic limit, we expect that to leading order in the gradient expansion of relevant fields, the expectation values of these currents can be expressed in the form of Ohm's law. The corresponding conductivities can then be defined in the following way: If a chiral system is perturbed by a small external magnetic field $B^{\mu} = (1/2)\epsilon^{\mu\nu\rho\sigma}u_{\nu}F_{\rho\sigma}$ and a spacetime vortex $\omega^{\mu} = \epsilon^{\mu\nu\rho\sigma}u_{\nu}\nabla_{\rho}u_{\sigma}$, where u^{μ} is the fluid velocity vector in the laboratory frame, then the expectation values of the two currents change by $\langle \delta J^{\mu} \rangle$ and $\langle \delta J_5^{\mu} \rangle$. Note that unlike in Eq. (4.1), both the axial and vector current conservation are now broken by the induced anomalies. To leading (dissipationless) order, the change can be expressed in terms of the conductivity matrix

$$\begin{pmatrix} \langle \delta J^{\mu} \rangle \\ \langle \delta J_{5}^{\mu} \rangle \end{pmatrix} = \begin{pmatrix} \sigma_{JB} & \sigma_{J\omega} \\ \sigma_{J_{5}B} & \sigma_{J_{5}\omega} \end{pmatrix} \begin{pmatrix} B^{\mu} \\ \omega^{\mu} \end{pmatrix},$$
(4.2)

where σ_{JB} is known as the *chiral magnetic conductivity*, $\sigma_{J\omega}$ as the *chiral vortical conductivity* and σ_{J_5B} as the *chiral separation conductivity*. The signature of anomalies can thus be traced all the way to the extreme IR physics and analysed by the linear response theory. This will be the subject studied in this work.

By following a set of rules postulated in [202] (see also [203]), a conve-

¹For a recent discussion of anomalies from the point of view of UV divergences in classical physics and its connection to the breakdown of the time reversal symmetry, see [197, 198].

nient way to express the anomalous conductivities is in terms of the anomaly polynomials. We briefly review these rules in Appendix 4.6.1. They allow one to compute the anomalous conductivities from the structure of the anomaly polynomials in arbitrary (even) dimensions, independently of the value of the coupling constant [199, 202–204].

In the IR limit, we may assume that the stress-energy tensor and the charge current can be expressed in a hydrodynamic gradient expansion [50, 51, 64, 184]. The constitutive relations for a fluid with broken parity, in the Landau frame, are [112, 113, 195, 205]

$$T^{\mu\nu} = \varepsilon u^{\mu} u^{\nu} + P \Delta^{\mu\nu} - \eta \sigma^{\mu\nu} - \zeta \Delta^{\mu\nu} \nabla_{\lambda} u^{\lambda} + \mathcal{O}\left(\partial^{2}\right),$$

$$J_{I}^{\mu} = n_{I} u^{\mu} + \sigma_{I} \Delta^{\mu\nu} \left(u^{\rho} F_{I,\rho\nu} - T \nabla_{\nu} \left(\frac{\mu_{I}}{T}\right)\right) + \xi_{I,B} B_{I}^{\mu} + \xi_{I,\omega} \omega^{\mu} + \mathcal{O}\left(\partial^{2}\right),$$
(4.3)

where the index $I = \{A, V\}$ labels the axial and the vector currents $(J_5^{\mu} = J_A^{\mu})$, $J^{\mu} = J_V^{\mu}$) and their respective transport coefficients. In the stress-energy tensor, ε , P, η and ζ are the energy density, pressure, shear viscosity and bulk viscosity. Furthermore, n, σ , T, μ and $F_{\mu\nu}$ are the charge density, charge conductivity, temperature, chemical potential and the gauge field strength tensor. The vector field u^{μ} is the velocity field of the fluid, the transverse projector (to the fluid flow) $\Delta^{\mu\nu}$ is defined as $\Delta^{\mu\nu} = u^{\mu}u^{\nu} + G^{\mu\nu}$, with $G^{\mu\nu}$ the metric tensor and $\sigma^{\mu\nu}$ the symmetric, transverse and traceless relativistic shear tensor composed of $\nabla_{\mu}u_{\nu}$. Plugging the above constitutive relations into the anomalous Ward identities, one can show that the anomalous conductivities are controlled by the transport coefficients ξ_B and ξ_{ω} (see e.g. [206]). It was shown in [195, 196] that by demanding the non-negativity of local entropy production (and similarly, by using a Euclidean effective action in [61, 62, 199])², the anomalous chiral separation conductivity σ_{J_5B} and the chiral magnetic conductivity σ_{JB} become fixed by the anomaly coefficient γ :

$$\sigma_{J_5B} = -2\gamma\mu, \qquad \qquad \sigma_{JB} = -2\gamma\mu_5. \tag{4.4}$$

²Note that the analysis in [61, 62, 195] only involves the axial gauge field. However, it is straightforward to generalise their results to the case with both the axial and the vector current.

On the other hand, the transport coefficient $\sigma_{J_5\omega}$ could not be completely determined by the anomaly and thermodynamic quantities. Its form contains an additional constant term,

$$\sigma_{J_5\omega} = \kappa \mu^2 + \tilde{c}T^2, \tag{4.5}$$

where \tilde{c} is some yet-undetermined constant, which could run along the renormalisation group flow. By using perturbative field theory methods [86, 207] and simple holographic models [204, 206], it was then suggested that \tilde{c} could be fixed by the gravitational anomaly coefficients, λ . However, the gravitational anomaly enters the equations of motion (4.1) with terms at fourth order in the derivative expansion while ξ_{ω} and ξ_{B} enter the equation of motion at second order. Thus, if one analysed the hydrodynamic expansion in terms of the naïve gradient expansion with all fluctuations of the same order, it would seem to be impossible to express \tilde{c} in terms of the gravitational anomaly. The above paradox was resolved in [199]. There, the theory was placed on a product space of a cone and a two dimensional manifold. The deficit angle δ was defined along the thermal cycle, β , as $\beta \sim \beta + 2\pi(1+\delta)$. Demanding continuity of onepoint functions in the vicinity of $\delta = 0$ then fixed the unknown coefficient \tilde{c} in terms of the gravitational anomaly coefficient λ (the gradient expansion breaks down). The above construction can be extended to theories outside the hydrodynamic regime in arbitrary even dimensions and in the presence of other types of anomalies, so long as the theories only involve background gauge fields and a background metric [203].

In the presence of dynamical gauge fields, the anomalous transport coefficients do not seem to remain protected from radiative corrections. This is consistent with the fact that the chiral vortical conductivity $\sigma_{J\omega}$, given otherwise by the thermal field theory result

$$\sigma_{J\omega} = 2\gamma\mu_5\mu,\tag{4.6}$$

was also argued to get renormalised in theories with dynamical gauge fields by [208–210]. Furthermore, these various pieces of information regarding the

renormalisation of the chiral conductivities are consistent with the findings of [194] (already noted above) and lattice results [211–214]: In theories with dynamical gauge fields and mixed, gauge-global anomalies, chiral conductivities renormalise.

Holography and universality of transport coefficients

Certain classes of strongly interacting theories at finite temperature and chemical potential can be formulated using gauge-gravity (holographic) duality. Thus, in comparison with the weakly coupled regime accessible to perturbative field theory calculations, holography can be seen as a convenient tool to investigate chiral transport properties at the opposite end of the coupling constant scale. Within holography, anomalous hydrodynamic transport was first studied in the context of fluid-gravity correspondence [110] by [112, 113] who added the Chern-Simons gauge field to the bulk. The two DC conductivities associated specifically with chiral magnetic and chiral vortical effects were then computed in the five-dimensional anti-de Sitter Reissner-Nördstrom black brane background in [206, 215, 216]. The results were extended to arbitrary dimensions in [204]. The work of [204] showed that these transport coefficients could be extracted from first-order differential equations (as opposed to the usual second-order wave equations in the bulk) due to the existence of a conserved current along the holographic radial direction. In a similar manner, this occurs in computations of the shear viscosity [30, 103] and other DC conductivities [126, 175]. We will refer to this situation as the case when the membrane paradigm is applicable (see Fig. 4.1). The existence of the membrane paradigm makes the calculation of chiral conductivities significantly simpler. Reassuringly, the holographic results for the chiral conductivities agree with the results obtained from conventional QFT methods described above and stated in Eqs. (4.4), (4.5) and (4.6) [86, 202, 207]. More recently, these calculations were generalised to cases of non-conformal holography (in which $T^{\mu}_{\ \mu} \neq 0$), giving the same results [88, 217]. A way to think of such holographic setups is as of geometric realisations of the renormalisation group flows.



Figure 4.1. A schematic representation of the membrane paradigm: The image on the left-hand-side corresponds to a holographic calculation (without the membrane paradigm) in which one has to solve for the bulk fields all along the D dimensional bulk. On the right-hand-side (the membrane paradigm case), the field theory observable of interest can be read off from a conserved current (along the radial coordinate). Hence, we only need information about its dynamics at the horizon and the AdS boundary. The membrane paradigm enables us to consider independent *effective theories* at the two surfaces with (D-1) dimensions. While the UV effective theory directly sources the dual field theory, it is the IR theory on the horizon that fixes the values of dual correlators in terms of the bulk black hole parameters. As in this chapter, such a structure may enable us to make much more general (universal) claims about field theory observables then if the calculation depended on the details of the full D-dimensional dynamics.

Universal holographic statements, most prominent among them being the ratio of shear viscosity to entropy density, $\eta/s = \hbar/(4\pi k_B)$ [30, 103, 126], can normally be reduced to an analysis of the dynamics of a minimally-coupled massless scalar mode and the existence of the membrane paradigm. The fact that the membrane paradigm exists in some theories for anomalous chiral conductivities thus naturally leads to the possibility of universality of these transport coefficients in holography. Motivated by this fact, in this work, we study whether and when non-renormalisation theorems for anomalous transport can be established in holography.

Recently, a work by Gürsoy and Tarrío [217] made the first step in this direction by proving the universality of chiral magnetic conductivity σ_{JB} in a two-derivative Einstein-Maxwell-dilaton theory with an arbitrary scalar field potential and anomaly-inducing Chern-Simons terms. The only necessary assumptions were that the bulk geometry is asymptotically anti-de Sitter (AdS) and that the Ricci scalar at the horizon must be regular. Because this statement is valid for two-derivative theories, it applies to duals at infinitely strong ('t Hooft) coupling λ and infinite number of adjoint colours, N. In this sense, it is applicable within the same class of theories as the statement of universality for η/s .

Higher-derivative corrections to supergravity actions arise when α' corrections are computed from string theory. Usually, this is done by either computing loop corrections to the β -functions of the sigma model or by computing string scattering amplitudes and guessing the effective supergravity action that could result in the same amplitudes (see e.g. [137, 218, 219]). Via the holographic dictionary, these higher-derivative corrections translate into (perturbative) coupling constant corrections in powers of the inverse coupling constant $(1/\lambda)$ expanded around $\lambda \to \infty$ [139]. The result of $\eta/s = 1/(4\pi)$ (having set $\hbar = k_B = 1$) is not protected from higher-derivative bulk corrections; it receives coupling constant corrections both in four-derivative theories (curvature-squared) [220-223] and in the presence of the leading-order top-down corrections to the $\mathcal{N} = 4$ supersymmetric Yang-Mills theory with an infinite number of colours (these R^4 corrections are proportional to $\alpha'^3 \sim$ $1/\lambda^{3/2}$) [140]. An equivalent statement exists also in second-order hydrodynamics [50, 110]. There, a particular linear combination of three transport coefficients, $2\eta \tau_{\Pi} - 4\lambda_1 - \lambda_2$, was shown to vanish for the same class of twoderivative theories as those that exhibit universality of η/s . It was then shown that the same linear combination of second-order transport coefficient vanishes to leading order in the coupling constant corrections even when curvaturesquared terms [149, 224] and R^4 terms dual to the $\mathcal{N} = 4$ 't Hooft coupling corrections are included in the bulk action [149]. However, by using the nonperturbative results for these transport coefficients in Gauss-Bonnet theory [225], one finds that the universal relation is violated non-perturbatively (or at second order in the perturbative coupling constant expansion) [149].³

³The violation of universality in second-order hydrodynamics was later also verified in [226]

Our goal in this work is to study the universality of the four anomalous conductivities σ_{JB} , $\sigma_{J\omega}$, σ_{J_5B} and $\sigma_{J_5\omega}$ in general higher-derivative theories, thereby incorporating an infinite series of coupling constant corrections to results at infinite coupling (from two-derivative bulk theories). What we will show is that the expressions (4.4), (4.5) and (4.6) remain universal in any higherderivative theory so long as the action (excluding the Chern-Simons terms) is gauge- and diffeomorphism-invariant.⁴ All we will assume, in analogy with [217], is that the bulk theory is asymptotically AdS (it has a UV conformal fixed point) and that it permits a black brane solution with a regular, non-extremal horizon. In its essence, the proof will reduce to showing the validity of the membrane paradigm and then a study of a generic, higher-derivative effective theories (all possible terms present in the conserved current) at the horizon and the boundary (as depicted in Fig. 4.1). The condition of regularity of these constructions at the horizon will play a crucial role in the proof. By studying cases of theories for which the membrane paradigm fails, one can then find theories in which universality may be violated.

Our findings can be seen as a test of holography in reproducing the correct Ward identities for the anomalous currents. The fact that we find universality of chiral conductivities with an infinite series of coupling constant corrections (albeit expanded around infinite coupling) is an embodiment of the fact that when only global anomalies are present, anomalous transport is protected from radiative corrections. An example related to the presence of mixed, gauge-global anomalies, which will invalidate the membrane paradigm, will be studied in Section 4.4. Again, as expected from field theory arguments, a case like that will naturally be able to violate the universality (or non-renormalisation) of chiral conductivities.

The chapter is organised as follows: In Section 4.2, we describe the holographic theory at finite temperature and chemical potential that is studied in

by using fluid-gravity methods in Gauss-Bonnet theory.

⁴As we are mainly interested in theories in which the anomalous Ward identity retains the form of Eq. (4.1), the conditions of gauge- and diffeomorphism- invariance are imposed to avoid explicit violation of Eq. (4.1) by the bulk matter content (see Section 4.4.4 for a discussion of such an example that includes massive vector fields).

the main part of this work. We then turn to the proof of the universality of chiral conductivities in Section 4.3. First, in Section 4.3.1, we show how to compute anomalous conductivities by using the membrane paradigm and specify the conditions that must be obeyed in order for the membrane paradigm to be valid. In Section 4.3.2, we then prove that a gauge- and diffeomorphism- invariant action indeed satisfies those conditions and thus always gives the same anomalous conductivities. In Section 4.4, we study examples that obey and violate the conditions required for universality. In particular, those that violate the universality include either massive gauge fields or naked singularities in the bulk. We proceeds with a discussion of results and future directions in Section 4.5. Finally, Appendix 4.6.1 includes a discussion of anomaly polynomials and the replacement rule.

4.2 The holographic setup

In this work, we consider five dimensional bulk actions with a dynamical metric G_{ab} , two massless gauge fields A_a and V_a that are dual to the axial and the vector current in the boundary theory, respectively, and a set of scalar (dilaton) fields, ϕ_I :

$$S = \int d^5 x \sqrt{-G} \left\{ \mathcal{L} \left[A_a, V_a, G_{ab}, \phi_I \right] + \mathcal{L}_{CS} \left[A_a, V_a, G_{ab} \right] \right\}.$$
(4.7)

The Lagrangian density \mathcal{L} should be thought of as a general, diffeomorphismand gauge-invariant action that may include arbitrary higher-derivative terms of the fields. Since we are interested in anomalous transport, (4.7) must include the Chern-Simons terms, \mathcal{L}_{CS} , that source global chiral anomalies in the boundary theory. In holography, higher-than-second-derivative bulk terms correspond to the ('t Hooft) coupling corrections to otherwise infinitely strongly coupled states ($\lambda \to \infty$). Since \mathcal{L} may include operators with arbitrary orders of derivatives (and corresponding bulk coupling constants), holographically computed quantities describing a hypothetical dual of (4.7) are able to incorporate an infinite series of coupling constant corrections to observables at infinite coupling.⁵ However, one should still think of these corrections as perturbative in powers of $1/\lambda$ due to various potential problems that may arise in theories with higher derivatives, such as the Ostrogradsky instability [227, 228], see also [146] for a recent discussion of causality violation in theories with higher-derivative bulk actions, in particular with four-derivative, curvature-squared actions.

The second source of corrections are the quantum gravity corrections that need to be computed in order to find the 1/N-corrections in field theory. If we consider S in Eq. (4.7) to be a *local* quantum effective action, expanded in a gradient expansion, we may also claim that our holographic results incorporate certain types of (perturbative) 1/N corrections, included in \mathcal{L} . What is important is the expectation (or the condition) that the anomalous Chern-Simons terms in \mathcal{L}_{CS} do not renormalise under quantum bulk corrections.

It will prove convenient to write the action (4.7) as

$$\mathcal{L}\left[A_a, V_a, G_{ab}, \phi_I\right] \equiv \mathcal{L}_G + \mathcal{L}_\phi + \mathcal{L}_A + \mathcal{L}_V, \tag{4.8}$$

where \mathcal{L}_G now contains the Einstein-Hilbert term (along with the cosmological constant) and higher-derivative terms of the metric, expressed in terms various contractions and derivatives of the Riemann curvature R_{abcd} . \mathcal{L}_{ϕ} contains kinetic and potential terms of a set of neutral scalar fields, ϕ_I . By $F_{A,ab}$ and $F_{V,ab}$, we denote the field strengths corresponding to A_a and V_a , respectively. Arbitrary derivatives of $F_{A,ab}$ and $F_{V,ab}$ may enter into \mathcal{L}_A and \mathcal{L}_V , and along with the Chern-Simons terms,

$$\mathcal{L}_{A} = \mathcal{L}_{A} \left[F_{A,ab}, \nabla_{a} F_{A,bc}, \dots, R_{abcd}, \nabla_{a} R_{bcde}, \dots, \phi_{I}, \partial_{a} \phi_{I}, \dots \right],$$

$$\mathcal{L}_{V} = \mathcal{L}_{V} \left[F_{V,ab}, \nabla_{a} F_{V,bc}, \dots, R_{abcd}, \nabla_{a} R_{bcde}, \dots, \phi_{I}, \partial_{a} \phi_{I}, \dots \right], \quad (4.9)$$

$$\mathcal{L}_{CS} = \epsilon^{abcde} A_{a} \left(\frac{\kappa}{3} F_{A,bc} F_{A,de} + \gamma F_{V,bc} F_{V,de} + \lambda R^{p}_{\ qbc} R^{q}_{\ pde} \right).$$

The ellipses '...' stand for higher-derivative terms built from $F_{A,ab}$, $F_{V,ab}$, R, R_{ab} , R_{abcd} and ϕ_I .⁶ Note also that we have chosen \mathcal{L}_A and \mathcal{L}_V so as not to

⁵In type IIB theory, higher-derivative bulk terms and corrections to infinitely coupled results in $\mathcal{N}=4$ theory are proportional to powers of $\alpha'\propto 1/\lambda^{1/2}$. See e.g. [139] and numerous subsequent works.

 $^{^{6}\}text{Latin letters}$ $\{a,b,c,\ldots\}$ are used to label the spacetime indices in the five-dimensional bulk

mix the two gauge fields. If there were mixing terms like $F_{A,ab}F_V^{ab}$ in the Lagrangian, then the anomalous Ward identities would no longer be those from Eq. (4.1) and additional complications regarding operator mixing would have to be dealt with. We note that the normalisation of the Levi-Civita tensor is chosen to be $\epsilon_{trxyz} = \sqrt{-G}$.

Our goal is to study coupling constant corrections to the anomalous conductivities that arise from the Ward identity in Eq. (4.1). We therefore avoid any ingredients in the action (4.8) that would explicitly introduce additional terms into (4.1). Beyond imposing gauge- and diffeomorphism-invariance of (4.1), we will also restrict our attention to Lagrangians \mathcal{L}_A and \mathcal{L}_V that contain no Levi-Civita tensor. An explicit example with violated (bulk) gauge-invariance that can generate a mixed, gauge-global anomaly on the boundary (altering the Ward identity (4.1)) will be studied in Section 4.4.4.

Furthermore, we assume that the bulk theory admits a homogenous, translationally invariant and asymptotically anti-de Sitter black brane solution of the form

$$ds^{2} = r^{2}f(r)d\bar{t}^{2} + \frac{dr^{2}}{r^{2}g(r)} + r^{2}\left(d\bar{x}^{2} + d\bar{y}^{2} + d\bar{z}^{2}\right),$$

$$A = A_{t}(r)d\bar{t}, \qquad V = V_{t}(r)d\bar{t}, \qquad \phi_{I} = \phi_{I}(r),$$
(4.10)

with f(r) and g(r) two arbitrary functions of the radial coordinate r. At AdS infinity,

$$\lim_{r \to \infty} f(r) = \lim_{r \to \infty} g(r) = 1.$$
(4.11)

The coordinates used in Eq. (4.10), $\{\bar{x}^{\mu}, r\}$, will be referred to as the un-boosted coordinates. Near the (outer) horizon, we assume that the metric can be written in a non-extremal, Rindler form

$$f(r) = f_1(r - r_h) + f_2(r - r_h)^2 + \mathcal{O}(r - r_h)^3,$$
(4.12)

$$g(r) = g_1(r - r_h) + g_2(r - r_h)^2 + \mathcal{O}(r - r_h)^3.$$
 (4.13)

The Hawking temperature of this black brane background (and its dual) is given

theory while the spacetime indices in the dual boundary theory are denoted by the Greek letters $\{\mu, \nu, \rho, \ldots\}$. The indices $\{i, j, k, \ldots\}$ represent the spatial directions of the boundary theory.

by

$$T = \frac{r_h^2}{4\pi} \sqrt{f_1 g_1}.$$
 (4.14)

The classical equations of motion describing this system can be obtained by varying the action (4.8). Firstly, the variations of the two gauge fields give⁷

$$d \star H_5 = 0,$$
 $d \star H = 0,$ (4.15)

where the two-forms H_5 and H are defined as

$$H_{5} = \frac{1}{2} \left(\frac{\delta(\mathcal{L}_{A})}{\delta(\nabla_{a}A_{b})} - \nabla_{c} \frac{\delta(\mathcal{L}_{A})}{\delta(\nabla_{c}\nabla_{a}A_{b})} + \dots \right) dx^{a} dx^{b} + H_{5,CS},$$

$$H = \frac{1}{2} \left(\frac{\delta(\mathcal{L}_{V})}{\delta(\nabla_{a}V_{b})} - \nabla_{c} \frac{\delta(\mathcal{L}_{V})}{\delta(\nabla_{c}\nabla_{a}V_{b})} + \dots \right) dx^{a} dx^{b} + H_{CS},$$
(4.16)

where

$$H_{5,CS} = \kappa \star \omega_A + \gamma \star \omega_V + \lambda \star \omega_\Gamma, \quad H_{CS} = \gamma \star (V \wedge dA), \tag{4.17}$$

The ellipses again denote expressions coming from the higher-derivative terms. The three abelian Chern-Simons three-forms are composed of the two gauge field one-forms $A = A_a dx^a$ and $V = V_a dx^a$, and the Levi-Civita connection one-form $\Gamma^a_{\ b} = \Gamma^a_{\ bc} dx^c$ as

$$\omega_X = \operatorname{Tr}\left(X \wedge dX + \frac{2}{3}X \wedge X \wedge X\right),\tag{4.18}$$

where $X = \{A, V, \Gamma^a_b\}$.⁸

$$\star \Omega = \frac{1}{p!(5-p)!} \sqrt{-G} \Omega_{a_1\dots a_p} \epsilon^{a_1\dots a_p} a_{p+1\dots a_5} dx^{a_{p+1}} \wedge \dots \wedge dx^{a_5}.$$

⁸In terms of the index notation, the Chern-Simons form built out of the Levi-Civita connection is given by

$$\omega_{abc} = \Gamma^{p_1}_{\ p_2 a} \partial_b \Gamma^{p_2}_{\ p_1 c} + (2/3) \Gamma^{p_1}_{\ p_2 a} \Gamma^{p_2}_{\ p_3 b} \Gamma^{p_3}_{\ p_1 c}.$$

⁷In five spacetime dimensions, we define the Hodge dual of a *p*-form $\Omega = (p!)^{-1}\Omega_{a_1...a_p} dx^{a_1} \wedge \ldots \wedge dx^{a_p}$ as

Secondly, varying the metric gives the Einstein's equation

$$R_{ab} - \frac{1}{2}G_{ab}R + \ldots = T_{ab}^{M} + \frac{1}{2}\nabla_c \left(\Sigma_{ab}^{\ c} + \Sigma_{ba}^{\ c}\right), \tag{4.19}$$

where T_{ab}^{M} is the stress-energy tensor for the scalars and the gauge fields, excluding the Chern-Simons terms. The *spin current* $\Sigma_{ab}^{\ c}$ is defined as

$$\Sigma_{ab}^{\ c} = -\lambda \ \epsilon^{ad_1d_2d_3d_4} F_{d_1d_2} R_{d_3d_4b}^{\ c}. \tag{4.20}$$

We refer the reader to [204] for a more general definition of the spin current, its connection to the anomaly polynomial in Eq. (4.75) and expressions for \sum_{ab}^{c} for different anomaly polynomials. We assume that the equations of motion coming from the variations of the scalar fields in (4.7) can also be solved, but we will make no further reference to that set of equations. As stated above, the full system of equations is assumed to result in a non-extremal, asymptotically AdS black brane solution and non-trivial, backreacted profiles for the gauge and the scalar fields.

To find the set of anomalous conductivities $\{\sigma_{J_5B}, \sigma_{JB}, \sigma_{J_5\omega}, \sigma_{J\omega}\}$ in all hypothetical duals of this holographic setup, it is convenient to consider the following perturbed metric in the boosted (fluid-gravity) frame [204]:

$$ds^{2} = -2\sqrt{\frac{f(r)}{g(r)}}u_{\mu}drdx^{\mu} + r^{2}f(r)u_{\mu}u_{\nu}dx^{\mu}dx^{\nu} + r^{2}\Delta_{\mu\nu}dx^{\mu}dx^{\nu} + 2r^{2}h(r)u_{\mu}\omega_{\nu}dx^{\mu}dx^{\nu},$$
(4.21)

where the projector $\Delta_{\mu\nu}$ is defined as $\Delta_{\mu\nu} = \eta_{\mu\nu} + u_{\mu}u_{\nu}$, with $\eta_{\mu\nu}$ the fourdimensional Minkowski metric. Note that once we set the fluid to be stationary, i.e. $u_{eq}^{\mu} = \{-1, 0, 0, 0\}$, the metric (4.21) will return to the un-boosted form (4.10), but in the Eddington-Finkelstein coordinates, as is usual in the fluid-gravity correspondence [110, 183]. The perturbations are organised so that the fluid velocity u_{μ} depends only on the boundary coordinates x^{μ} and all of the *r*-dependence is encoded in h(r). Since the vorticity is defined as $\omega^{\mu} = \epsilon^{\mu\nu\rho\sigma}u_{\nu}\partial_{\rho}u_{\sigma}$, the last term in (4.21) corresponds to the metric perturbations at first order in the derivative expansion (in the x^{μ} coordinates). Similarly, the perturbed axial and vector gauge fields can be written as⁹

$$A = -A_t(r) u_{\mu} dx^{\mu} + \tilde{a}(x^{\mu}) + a(r) \omega_{\mu} dx^{\mu},$$

$$V = -V_t(r) u_{\mu} dx^{\mu} + \tilde{v}(x^{\mu}) + v(r) \omega_{\mu} dx^{\mu}.$$
(4.23)

One may use the one-forms \tilde{a} and \tilde{v} to define the magnetic field source $B^{\mu} = \epsilon^{\mu\nu\rho\sigma}u_{\nu}\partial_{\rho}\tilde{v}_{\sigma}$ and the (fictitious) axial magnetic field source $B^{\mu}_{5} = \epsilon^{\mu\nu\rho\sigma}u_{\nu}\partial_{\rho}\tilde{a}_{\sigma}$.

4.3 **Proof of universality**

In this section, we show that upon expanding the equations of motion (4.15) and (4.19) to first order in the (boundary) derivative expansion, the conserved currents can be expressed as a total radial derivative of some function. This type of a radially conserved quantity is necessary for the applicability of the membrane paradigm, used e.g. in [126] and many other holographic studies. To express all four anomalous conductivities purely in terms of the near-horizon data, our work will generalise the membrane paradigm result for the chiral magnetic conductivity of Gürsoy and Tarrío [217]. This will then enable us to establish the universality of the four transport coefficients in the presence of a general higher-derivative bulk theory specified in Section 4.2. Furthermore, the structure of the equations will single out the properties that holographic theories must violate in order for there to be a possibility that the dual conductivities

$$ds^{2} = -2S(r) u_{\mu}dx^{\mu}dr + F(r) u_{\mu}u_{\nu}dx^{\mu}dx^{\nu} + G(r) \Delta_{\mu\nu}dx^{\mu}dx^{\nu} + 2H^{\perp}_{\mu}(r,x) u_{\nu}dx^{\mu}dx^{\nu} + \Pi(r) \sigma_{\mu\nu}dx^{\mu}dx^{\nu},$$
(4.22)

$$A = C(r)u_{\mu}dx^{\mu} + a_{\mu}^{\perp}(r, x)dx^{\mu}, \qquad V = D(r)u_{\mu}dx^{\mu} + v_{\mu}^{\perp}(r, x)dx^{\mu},$$

where H^{\perp}_{μ} , a^{\perp}_{μ} and v^{\perp}_{μ} are vectors orthogonal to the fluid velocity u^{μ} . Using the equations of motion for $\{H^{\perp}_{\mu}, a^{\perp}_{\mu}, v^{\perp}_{\mu}\}$, one can show that they decouple from all other perturbations at the same order in the gradient expansion (see e.g. [112, 113]). Thus, to compute anomalous conductivities, one can consistently solve for only $\{H^{\perp}_{\mu}, a^{\perp}_{\mu}, v^{\perp}_{\mu}\}$, setting the remaining perturbations to zero. To first order, this gives our Eqs. (4.21) and (4.23).

⁹Our choice of the metric and the gauge fields can be understood in the following way: If one considers the perturbed metric and the gauge fields with all possible terms at first order in gradient expansions, they have the form

may get renormalised.

Our proof can be divided into two steps: First (in Section 4.3.1), we expand the equations of motion for the gauge field (4.15) to first order in the (boundary coordinate) derivative expansions and arrange them into a total-derivative form of a conserved current along the radial direction. This radially conserved current can be written as a sum of the anomalous Chern-Simons terms and terms that come from the rest of the action. We identify the conditions that each of these terms has to satisfy in order for the anomalous conductivities to have a universal form fixed by the Chern-Simons action. Proving the validity of these conditions is then done in Section 4.3.2 by analysing the horizon and the boundary behaviour of the higher-derivative bulk effective action (all possible terms that can appear in the conserved current).

4.3.1 Anomalous conductivities and the membrane paradigm

Let us begin by considering the axial and the vector currents, $\langle \delta J_5^{\mu} \rangle$ and $\langle \delta J^{\mu} \rangle$, sourced by a small magnetic field and a small vortex. As in [217], the membrane paradigm equations follow from the two Maxwell's equations in (4.15). For conciseness, we only show the details of the axial current computation, which involves H_5 from Eq. (4.16). A calculation for the vector current, involving H, proceeds along similar lines. In case of the vector current, we will only state the relevant results.

To first order in the gradient expansion along the boundary directions x^{μ} , both equations in (4.15) can be schematically written as

$$\partial_r \left(\sqrt{-G} H_5^{ra} \left(\partial^1 \right) \right) + \partial_\mu \left(\sqrt{-G} H_5^{\mu a} \left(\partial^0 \right) \right) = 0, \tag{4.24}$$

where $H_5^{ra}(\partial^0)$ and $H_5^{\mu a}(\partial^1)$ are the components of the conserved current two-form in Eq. (4.16) that contain zero- and one-derivative terms (derivatives are taken with respect to x^{μ}).

As our first goal is to rewrite the problem in terms of a radially conserved quantity, we need to consider the structure of second term in (4.24). We will set the index a to the four-dimensional index ν . It is easy to see that only the

Chern-Simons terms from \mathcal{L}_{CS} can enter into this term at zeroth order in the (boundary) derivative expansion, i.e. $\partial_{\mu} \left(\sqrt{-G} H_5^{\mu\nu} \left(\partial^0 \right) \right) |_{\kappa=g=\lambda=0} = 0$ (cf. Eq. (4.9)). This is because $H_5^{\mu\nu}$ can only be constructed out of the (axial) gauge field (4.23) and the metric tensor (4.21), containing no derivatives along x^{μ} . At zeroth-order in the derivative expansion, any two-tensor $X^{\mu\nu}$ can thus be decomposed as

$$X^{\mu\nu} = X_1 u^{\mu} u^{\nu} + X_2 \Delta^{\mu\nu} + X_3 u^{(\mu} A^{\nu)} + X_4 u^{[\mu} A^{\nu]}, \qquad (4.25)$$

where X_i are scalar functions of the radial coordinate. For an anti-symmetric $X^{\mu\nu}$, as are $H_5^{\mu\nu}$ and $H^{\mu\nu}$, X_1 , X_2 and X_3 must vanish and only X_4 can be non-zero. Since such a term can only come from \mathcal{L}_{CS} , \mathcal{L} cannot contribute to the second term in (4.24). For $a = \nu$, the two terms in Eq. (4.24) are therefore given by

$$\partial_{r} \left[\sqrt{-G} H_{5}^{r\nu} \left(\partial^{1} \right) \right] = \frac{\partial}{\partial r} \left[\dots + \kappa \left(A_{t} B_{5}^{\nu} + A_{t}^{2} \omega^{\nu} \right) \right. \\ \left. + \gamma \left(V_{t} B^{\nu} + V_{t}^{2} \omega^{\nu} \right) + \lambda \frac{g(r^{3} f')^{2}}{2r^{2} f} \omega^{\nu} \right], \qquad (4.26)$$
$$\partial_{\mu} \left[\sqrt{-G} H_{5}^{\mu\nu} \left(\partial^{0} \right) \right] = \kappa \left(\partial_{r} A_{t} \right) B_{5}^{\nu} + \gamma \left(\partial_{r} V_{t} \right) B^{\nu} \\ = \partial_{r} \left(\kappa A_{t} B_{5}^{\nu} + g V_{t} B^{\nu} \right).$$

The ellipsis indicates the non-Chern-Simons terms. Hence, one can write the Maxwell's equation for the axial gauge field as a derivative of a conserved current along the r-direction:

$$\partial_r \mathcal{J}_5^\mu(r) = 0. \tag{4.27}$$

The axial bulk current is defined as

$$\mathcal{J}_{5}^{\mu}(r) = \mathcal{J}_{5,mb}^{\mu}(r) + \mathcal{J}_{5,r}^{\mu}(r) + \mathcal{J}_{5,CS}^{\mu}(r), \qquad (4.28)$$

where the membrane current $\mathcal{J}^{\mu}_{5,mb}(r),$ the Chern-Simons current $\mathcal{J}^{\mu}_{5,CS}$ and

 $\mathcal{J}^{\mu}_{5,r}$ are defined as

$$\mathcal{J}_{5,mb}^{\mu} = \sqrt{-G} \left(\frac{\partial \mathcal{L}_A}{\partial A'_{\mu}} - \partial_a \frac{\partial \mathcal{L}_A}{\partial (\partial_a A'_{\mu})} + \dots \right) \Big|_{h(r) \to 0},$$

$$\mathcal{J}_{5,r}^{\mu} = \sqrt{-G} \left(\frac{\partial \mathcal{L}_A}{\partial A'_{\mu}} - \partial_a \frac{\partial \mathcal{L}_A}{\partial (\partial_a A'_{\mu})} + \dots \right) \Big|_{a(r) \to 0},$$

$$\mathcal{J}_{5,CS}^{\mu} = 2\kappa A_t B_5^{\mu} + 2\gamma V_t B^{\mu} + \left(\kappa A_t^2 + \lambda \frac{g(r^2 f')^2}{2f} \right) \omega^{\mu}.$$
(4.29)

Note that the primes indicate derivatives with respect to the radial coordinate.

The expectation value of the external boundary current $\langle \delta J_5^{\mu} \rangle$ that we turned on to excite anomalous transport (cf. Eq. (4.2)) is obtained by varying the perturbed on-shell action (4.8) with respect to the bulk axial gauge field fluctuation at the boundary. We find that it is the membrane current $\mathcal{J}_{5,mb}^{\mu}$ evaluated at the boundary $(r \to \infty)$ that can be interpreted as its expectation value:

$$\langle \delta J_5^{\mu} \rangle = \lim_{r \to \infty} \mathcal{J}_{5,mb}^{\mu}(r). \tag{4.30}$$

This result is of central importance to the existence of the membrane paradigm in our discussion.

Let us now study how $\mathcal{J}^{\mu}_{5,mb}$ can be related to the full conserved current \mathcal{J}^{μ} from Eq. (4.28). What will prove very convenient is the gauge choice for A and V whereby (see e.g. [215])

$$\lim_{r \to \infty} A_t(r) = 0, \qquad \qquad \lim_{r \to \infty} V_t(r) = 0. \tag{4.31}$$

Such a choice results in¹⁰

$$\lim_{r \to \infty} \mathcal{J}_{5,CS}(r) = 0, \tag{4.32}$$

which together with the conservation equation (4.27) and Eq. (4.30) implies that

$$\langle \delta J_5^{\mu} \rangle = \mathcal{J}_{5,mb}^{\mu}(r_h) + \mathcal{J}_{5,r}^{\mu}(r_h) - \mathcal{J}_{5,r}^{\mu}(\infty) + \mathcal{J}_{5,CS}^{\mu}(r_h).$$
 (4.33)

What we will prove in the next section (Sec. 4.3.2) will be the statement that

¹⁰For an alternative gauge choice, see e.g. formalism B from Ref. [229].

for any theory specified by the action in (4.7),

$$\mathcal{J}^{\mu}_{5,mb}(r_h) + \mathcal{J}^{\mu}_{5,r}(r_h) - \mathcal{J}^{\mu}_{5,r}(\infty) = 0, \qquad (4.34)$$

implying that the current $\langle\delta J_5^\mu\rangle$ can be completely determined by only the Chern-Simons current evaluated at the horizon,

$$\langle \delta J_5^{\mu} \rangle = \mathcal{J}_{5,CS}^{\mu}(r_h). \tag{4.35}$$

The same reasoning and equations (4.30)–(4.35) apply also to the case of the vector current, up to the appropriate replacements of A_a by V_a , \mathcal{L}_A by \mathcal{L}_V and the axial Chern-Simons current by

$$\mathcal{J}_{CS}^{\mu} = 2\gamma \left(A_t B^{\mu} + V_t B_5^{\mu} \right) + 2\gamma A_t V_t \, \omega^{\mu}. \tag{4.36}$$

Let us for now assume that the condition (4.34) is satisfied and proceed to compute the anomalous conductivities. In our gauge choice, the gauge fields at the horizon are related to the two chemical potentials via

$$A_t(r_h) = -\mu_5,$$
 $V_t(r_h) = -\mu.$ (4.37)

By using the near-horizon expansions (4.12) and (4.13), the last term in $\mathcal{J}_{5,CS}^{\mu}$ from (4.29) can be related to the temperature

$$\frac{g\left(r^{2}f'\right)^{2}}{f} = r^{4}f_{1}g_{1} = 4\left(2\pi T\right)^{2}.$$
(4.38)

Furthermore, using the horizon values of the gauge fields from Eq. (4.37) along with the definitions of the anomalous conductivities from (4.2), we find

$$\sigma_{J_5B} = -2\gamma\mu, \qquad \sigma_{JB} = -2\gamma\mu_5,$$

$$\sigma_{J_5\omega} = \kappa\mu_5^2 + \gamma\mu^2 + 2\lambda(2\pi T)^2, \qquad \sigma_{J\omega} = 2\gamma\mu_5\mu.$$
(4.39)

Hence, so long as the condition (4.34) is satisfied, the bulk theory (4.7) gives precisely the non-renormalised, universal conductivities stated in Eqs. (4.4), (4.5) and (4.6).

4.3.2 Universality

We will now show that the condition (4.34) always holds in gauge- and diffeomorphism - invariant theories, thus establishing the universality of the anomalyinduced conductivities σ_{J_5B} , σ_{JB} , $\sigma_{J_5\omega}$ and $\sigma_{J\omega}$ from Eq. (4.39) in theories with arbitrary higher-derivative actions, dual to an infinite series of coupling constant corrections expanded around infinite coupling. The condition (4.34) requires us to understand how $\mathcal{J}^{\mu}_{5,mb}$ and $\mathcal{J}^{\mu}_{5,r}$ behave at the two ends of the fivedimensional geometry (boundary and horizon). To make general statements about that, we construct an effective field theory (or the effective current) in terms of the metric, gauge fields and dilatons with first-order perturbations to quadratic order in the amplitude expansion. The two conditions that we impose on the effective theory and the resulting currents are the following:

- (1) The theory must be regular at the non-extremal horizon, by which we mean that any Lorentz scalar present in the action (or a current) must be regular (non-singular) when evaluated at the horizon.
- (2) The bulk spacetime is asymptotically anti-de Sitter.

For conciseness, we again only analyse the axial gauge field, A_a . A completely equivalent procedure can be applied to the case of the vector gauge field, V_a .

From the definitions of $\mathcal{J}_{5,mb}^{\mu}$ and $\mathcal{J}_{5,r}^{\mu}$ in Eq. (4.29), it is clear that the only relevant part of the action (4.8) for this analysis is \mathcal{L}_A . Because the two currents are independent of the Chern-Simons terms, they only depend on the terms encoded in $H_5^{ra}(\partial^1)$ (see discussion below Eq. (4.24)). The possible terms in $H_5^{ra}(\partial^1)$ that correspond to $\mathcal{J}_{5,mb}^{\mu}$ and $\mathcal{J}_{5,r}^{\mu}$ can be written (schematically, up to correct tensor structures of $\mathcal{C}_{A,n}$ and $\mathcal{C}_{G,n}$) as

$$H_5^{r\mu}\left(\partial^1\right) = \sum_{n=1}^{\infty} \left[\mathcal{C}_{A,n}\partial_r^n a(r) + \mathcal{C}_{G,n}\partial_r^n h(r)\right]\omega^{\mu} + H_{5,CS}^{r\mu}\left(\partial^1\right), \quad (4.40)$$

where $H_{5,CS}^{r\mu}$ is the irrelevant Chern-Simons part of $H_5^{r\mu}$, stated explicitly in Eq. (4.26). Since the action \mathcal{L}_A does not contain any Levi-Civita tensors, the terms in $\{\mathcal{C}_{A,n}, \mathcal{C}_{G,n}\}$ can only depend on a(r) and h(r). This implies that

 $C_{A,n} = C_{G,n} = 0$ when a(r) = h(r) = 0, to first order in the boundarycoordinate derivative expansion. Hence, the problem reduces to the question of finding all possible structure of the tensorial coefficients { $C_{A,n}$, $C_{G,n}$ } at the horizon and at the boundary.

It is now convenient to return to the un-boosted coordinates, $\{r, \bar{x}^{\mu}\}$, used in Eq. (4.10). In these coordinates, the perturbed metric and the axial gauge field are (in analogy with (4.21) and (4.23))

$$ds^{2} = -r^{2}f(r)d\bar{t}^{2} + \frac{dr^{2}}{r^{2}g(r)} + r^{2}(d\bar{x}^{2} + d\bar{y}^{2} + d\bar{z}^{2}) + 2h_{\bar{t}i}(r,\bar{x}^{i})d\bar{t}d\bar{x}^{i},$$
(4.41)

$$A = A_t dt + a_i(r, \bar{x}^i) d\bar{x}^i, \tag{4.42}$$

where the perturbations are now denoted by $h_{\bar{t}i}$, a_i and v_i with $i = \{x, y, z\}$. One can relate $\{h_{\bar{t}i}, a_i\}$ to $\{h(r), a(r)\}$ by using the appropriate coordinate transformations, which give

$$h_{\bar{t}i} = \dots + r^2 h(r) \, u_{\mu} \omega_{\nu} \frac{\partial x^{\mu}}{\partial \bar{t}} \frac{\partial x^{\nu}}{\partial \bar{x}^i} + \mathcal{O}\left(\partial^2\right),$$

$$a_i = \dots + a(r) \, \omega_{\mu} \frac{\partial x^{\mu}}{\partial \bar{x}^i} + \mathcal{O}\left(\partial^2\right). \tag{4.43}$$

Here, the ellipses denote the zeroth-order terms in the derivative expansion. It is convenient to consider $u^{\mu} - u^{\mu}_{eq}$ to be small, which gives

$$u_{\mu}dx^{\mu} = dt + \delta u_{i}dx^{i}, \quad dt = d\bar{t} + \frac{1}{r^{2}}\sqrt{\frac{1}{f(r)g(r)}}dr, \quad dx^{i} = d\bar{x}^{i}.$$
 (4.44)

This choice of the fluid velocity further gives $\omega^t = B^t = 0$. Thus, in the remainder in this section, we will only write down the tensors $\{H_5^{r\mu}, \mathcal{J}_5^{\mu}, \mathcal{J}_{5,CS}^{\mu}\}$ with spatial components of $\mu = \{i, j, k, \ldots\}$. It immediately follows that $H_5^{ri}(r, x^{\mu})$ in the boosted coordinates and $H_5^{ri}(r, \bar{x}^{\mu})$ in the un-boosted coordinates have identical expressions. In analogy with (4.40), expanding H_5^{ri} in the un-boosted

coordinates to first order in amplitudes of a_i and $h_{\bar{t}i}$,

$$H_{5}^{ri}[a_{i},h_{ti}] = \left(\mathcal{I}_{A,1}^{rirj}\partial_{r}a_{j} + \mathcal{I}_{A,2}^{rirrj}\partial_{r}^{2}a_{j} + \ldots\right) \\ + \left(\mathcal{I}_{G,0}^{ri\bar{t}j}h_{\bar{t}j} + \mathcal{I}_{G,1}^{ri\bar{t}rj}\partial_{r}h_{\bar{t}j} + \mathcal{I}_{G,2}^{ri\bar{t}rrj}\partial_{r}^{2}h_{\bar{t}j} + \ldots\right) \\ + \left(\text{terms with derivatives along } x^{i}\right).$$

$$(4.45)$$

Note that $\mathcal{I}_{A,0}^{rij} = 0$ because gauge-invariance of \mathcal{L}_A excludes the possibility of any explicit dependence on a_i (only derivatives of a_i may appear). The ellipses represent terms with higher derivatives in r and $\{\mathcal{I}_{A,n}, \mathcal{I}_{G,n}\}$ are tensors contracted with $\partial_r^n a_i$ and $\partial_r^n h_{\bar{t}i}$. To verify (4.45), we can use the coordinate transformations (4.43), which show that all relevant terms from (4.40) are indeed contained in (4.45). Thus, one can determine the coefficients $\{\mathcal{C}_{A,n}, \mathcal{C}_{G,n}\}$ by applying (4.44) to (4.45) and matching the coefficients of $\partial_r^n a(r) \omega^i$ and $\partial_r^n h(r) \omega^i$.

The structure of the $\{\mathcal{I}_{G,n}, \mathcal{I}_{A,n}\}$ tensors near the horizon and the AdSboundary can be understood in the following way: In the un-boosted frame, we define five mutually orthogonal unit-vectors or vielbeins, $e_{\hat{p}a} = \delta_{\hat{p}a}$, where the hatted indices $\{\hat{p}, \hat{q}, ..\} = \{\hat{0}, \hat{1}, \hat{2}, \hat{3}, \hat{4}\}$ are used as (local flat space) bookkeeping indices. The full set of the five-dimensional vectors with upper Lorentz indices can now be written as $e^a_{\ \hat{p}} = \left[\sqrt{G}\right]^{ab} \delta_{\hat{p}b}$:

$$e_{\hat{0}} = \left(\left(r^{2} f \right)^{-1/2}, 0, 0, 0, 0 \right),$$

$$e_{\hat{1}} = \left(0, 1/r, 0, 0, 0 \right),$$

$$e_{\hat{2}} = \left(0, 0, 1/r, 0, 0 \right),$$

$$e_{\hat{3}} = \left(0, 0, 0, 1/r, 0 \right),$$

$$e_{\hat{4}} = \left(0, 0, 0, 0, \left(r^{2} g \right)^{1/2} \right).$$
(4.46)

These normal vectors allow us to write the tensors $\{\mathcal{I}_{G,n}, \mathcal{I}_{A,n}\}$ as

$$\mathcal{I}_{A,n}^{a_{1}a_{2}...a_{m}} = \sum_{\hat{p}_{1},...,\hat{p}_{m}} \mathcal{S}_{A,n}^{\hat{p}_{1}...\hat{p}_{m}} e^{a_{1}}_{\hat{p}_{1}} \dots e^{a_{m}}_{\hat{p}_{m}},
\mathcal{I}_{G,n}^{a_{1}a_{2}...a_{m}} = \sum_{\hat{p}_{1},...,\hat{p}_{m}} \mathcal{S}_{G,n}^{\hat{p}_{1}...\hat{p}_{m}} e^{a_{1}}_{\hat{p}_{1}} \dots e^{a_{m}}_{\hat{p}_{m}},$$
(4.47)

where $\{S_{A,n}, S_{G,n}\}$ are (spacetime) Lorentz-scalars. The regularity condition imposed at the horizon demands that these scalar have to be non-singular at $r = r_h$. The question of whether $\mathcal{I}_{G,n}$ and $\mathcal{I}_{A,n}$ vanish at the horizon is therefore completely determined by the values the projectors $e^{a_1}_{\hat{p}_1} \dots e^{a_m}_{\hat{p}_m}$ take when evaluated at the horizon. To demonstrate this fact more clearly, let us write down the first few relevant components of the tensors $\mathcal{I}_{G,n}$ and $\mathcal{I}_{A,n}$ explicitly:

$$\begin{split} \mathcal{I}_{G,0}^{ri\bar{t}j} &= \left(r^{-2}\sqrt{g/f}\right) \, \mathcal{S}_{G,0}^{4\hat{i}0\hat{j}} \,, \qquad \mathcal{I}_{A,0}^{rij} = 0 \,, \\ \mathcal{I}_{G,1}^{ri\bar{t}rj} &= \left(r^{-1}\sqrt{g^2/f}\right) \, \mathcal{S}_{G,2}^{4\hat{i}04\hat{j}} \,, \qquad \mathcal{I}_{A,1}^{rirj} = g \, \mathcal{S}_{A,1}^{4\hat{i}4\hat{j}} \,, \\ \mathcal{I}_{G,2}^{ri\bar{t}rrj} &= \left(\sqrt{g^3/f}\right) \, \mathcal{S}_{G,2}^{4\hat{i}044\hat{j}} \,, \qquad \mathcal{I}_{A,2}^{rirrj} = \left(rg^{3/2}\right) \, \mathcal{S}_{A,2}^{4\hat{i}44\hat{j}} \,, \\ \mathcal{I}_{G,3}^{ri\bar{t}rrrj} &= \left(r\sqrt{g^4/f}\right) \, \mathcal{S}_{G,3}^{4\hat{i}044\hat{j}} \,, \qquad \mathcal{I}_{A,3}^{rirrrj} = \left(r^2g^2\right) \, \mathcal{S}_{A,3}^{4\hat{i}444\hat{j}} \,, \end{split}$$

with $r = r_h$. As before, the tensor $\mathcal{I}_{A,0}^{rij} = 0$ because of the gauge-invariance of \mathcal{L}_A .

With this decomposition, the problem of determining the non-zero terms in H_5^{ri} has been reduced to simple power-counting. Namely, a tensor $\mathcal{I}^{a_1a_2...}$ can only be non-zero at the horizon if the number of $e_{\hat{0}}^{\bar{t}}$ in its decomposition is greater or equal to the number of $e_{\hat{4}}^r$. The regularity of the scalars $\mathcal{S}_{A,n}$ and $\mathcal{S}_{G,n}$ at the horizon plays a crucial role here. Hence, one can see that the only non-zero tensor from the set of $\{\mathcal{I}_{A,n}, \mathcal{I}_{G,n}\}$ is $\mathcal{I}_{G,0}^{ri\bar{t}j}$. The conserved current evaluated at the horizon thus becomes

$$\mathcal{J}_{5}^{i} = \sqrt{-G} \left(\sqrt{\frac{g}{f}} \, \mathcal{S}_{G,0}^{4\hat{j}0\hat{i}} \right) h(r_{h}) \, u_{\mu} \omega_{\nu} \frac{\partial x^{\mu}}{\partial \bar{t}} \frac{\partial x^{\nu}}{\partial \bar{x}^{j}} + \mathcal{J}_{5,CS}^{i}(r_{h}). \tag{4.48}$$

To see why the first term in (4.48) has to vanish, recall that as other scalars, the

Ricci scalar also has to be regular at the horizon. As pointed out in [217], this condition implies that $h_{ti} \sim (r - r_h)$ at the horizon. Therefore, the conserved current at the horizon is indeed fully determined by the anomalous Chern-Simons term:

$$\mathcal{J}_5^i = \mathcal{J}_{5,CS}^i(r_h). \tag{4.49}$$

With $H_5^{rt} = 0$, Eq. (4.49) implies the first two terms from the condition (4.34) vanish:

$$\mathcal{J}^{\mu}_{5,mb}(r_h) + \mathcal{J}^{\mu}_{5,r}(r_h) = 0.$$
(4.50)

Similarly, we can determine the value of the current $\mathcal{J}_{5,r}^{\mu}$ at the boundary. Since $\mathcal{J}_{5,r}^{\mu}$ includes only terms linear in h(r), it is enough to consider

$$H_5^{ri} = \left(\mathcal{I}_{G,0}^{ri\bar{t}j} h_{\bar{t}j} + \mathcal{I}_{G,1}^{ri\bar{t}rj} \partial_r h_{\bar{t}j} + \mathcal{I}_{G,2}^{ri\bar{t}rrj} \partial_r^2 h_{\bar{t}j} + \dots \right) + \dots$$
(4.51)

Now, because the boundary is asymptotically AdS and higher-derivative terms considered here do not change the scaling behaviour near the boundary, we can use the near-AdS solution for h(r) [204]:

$$h(r) = \frac{\mathcal{H}}{r^4} + \mathcal{O}\left(r^{-5}\right). \tag{4.52}$$

Substituting the expansion for h(r) into (4.51), it immediately follows that the third term in the condition (4.34) vanishes as well when it is evaluated at the boundary (note again that $H_5^{rt} = 0$):

$$\mathcal{J}^{\mu}_{5,r}(\infty) = 0.$$
 (4.53)

Together, Eqs. (4.50) and (4.53) imply the validity of the condition stated in Eq. (4.34), which completes our proof. The analysis of the vector current \mathcal{J}^{μ} and a proof of a condition analogous to (4.34) follow through along exactly the same lines. This implies that all four anomalous conductivities take the universal form of (4.39) for all holographic theories specified in (4.7) so long as the (effective) theory is regular at the non-extremal horizon and the bulk is asymptotically anti-de Sitter.

4.4 Examples and counter-examples

In this section, we turn our attention to explicit examples of theories that obey and violate the conditions used in our proof in Section 4.3 and thus result in universal and renormalised anomalous conductivities, respectively. We will first demonstrate their universality in two- and four-derivative theories with a non-extremal horizon and then move on to describing two holographic models, which violate the assumptions in the proof of Eq. (4.34). More precisely, in Section 4.4.1, we compute the conductivities in the two-derivative Einstein-Maxwell-Dilaton theory. In Section 4.4.2, we then show explicitly how our proof works in the case of the most general four-derivative action with Maxwell fields and dynamical gravity. In both of those case, the conductivities are universal and the current at the horizon only depends on the metric fluctuation, as established by our effective theory method in (4.48).

In Section 4.4.3, we comment on the validity of our proof in gravity duals without a horizon. We use the examples of the confining soft/hard-wall models and charged dilatonic black holes at zero temperature. The membrane paradigm computation goes through as before in the case of confining geometry. However, the conductivities no longer have any temperature dependence, which would require us to augment the replacement rule discussed in Appendix 4.6.1. As for the latter example, the family of theories considered suffers from naked singularities in the bulk. Lastly, in Section 4.4.4, we point out how the bulk terms corresponding to field theories with a gauge-global anomaly violate the assumptions in our proof. This is consistent with the known fact that anomalous conductivities in systems with mixed anomalies receive corrections along the renormalisation group flow. We will not review the details behind the holographic constructions of such systems but rather focus on the reasons for why these models may violate the universality from the point of view of Section 4.3.2.

4.4.1 Einstein-Maxwell-dilaton theory at finite temperature

As for our first example, we consider the two-derivative Einstein-Mawell-dilaton theory with a non-trivial dilaton profile:

$$\mathcal{L}_G = R - 2\Lambda, \qquad \qquad \mathcal{L}_\phi = -(\partial \phi)^2 - V(\phi), \qquad (4.54)$$

$$\mathcal{L}_{A} = -\frac{1}{4} Z_{A}(\phi) F_{A,ab} F_{A}{}^{ab}, \qquad \mathcal{L}_{V} = -\frac{1}{4} Z_{V}(\phi) F_{V,ab} F_{V}{}^{ab}, \qquad (4.55)$$

having used the notation of the action in Eq. (4.8). This is an extension of the case studied in [217], which includes the gravitational anomaly and anomalous conductivities that follow from a response to a small vortex.

The theory has two charges that are conserved along the radial direction at zeroth-order in the boundary-derivative expansion. The expressions follow from the $a = \mu$ component of the Maxwell's equations:

$$Q_5 = r^3 \sqrt{\frac{g}{f}} Z_A \partial_r A_t, \tag{4.56}$$

$$Q = r^3 \sqrt{\frac{g}{f}} Z_V \partial_r V_t. \tag{4.57}$$

At first order in derivatives, the two conserved currents \mathcal{J}^{μ}_5 and \mathcal{J}^{μ} are given by

$$\mathcal{J}_{5}^{\mu} = \left[Q_{5}h + r^{3}\sqrt{fg}Z_{A}\partial_{r}a\right]\omega^{\mu} + \mathcal{J}_{5,CS}^{\mu},$$

$$\mathcal{J}^{\mu} = \left[Qh + r^{3}\sqrt{fg}Z_{V}\partial_{r}v\right]\omega^{\mu} + \mathcal{J}_{CS}^{\mu}.$$
(4.58)

Thus, we can immediately read off the membrane currents:

$$\delta J^{\mu}_{5,mb} = r^3 \sqrt{\frac{g}{f}} Z_A \partial_r a, \qquad (4.59)$$

$$\delta J^{\mu}_{mb} = r^3 \sqrt{\frac{g}{f}} Z_V \partial_r v. \tag{4.60}$$

Moreover, the regularity of the black hole horizon implies that that metric fluctuation has to vanish at the horizon [217], i.e. $h(r_h) = 0$. At the horizon, the two currents $\mathcal{J}_5^{\mu}(r_h)$ and $\mathcal{J}^{\mu}(r_h)$ are therefore completely determined by the anomalous terms $\mathcal{J}_{5,CS}^{\mu}(r_h)$ and $\mathcal{J}_{CS}^{\mu}(r_h)$.

Next, we investigate the behaviour of \mathcal{J}^{μ}_{5} and \mathcal{J}^{μ} at the boundary. Sub-

stituting the near-boundary solutions (4.52) into (4.58), one can see that Q_5h and Qh are sub-leading, which implies that \mathcal{J}_5^{μ} and \mathcal{J}^{μ} at $r \to \infty$ become determined by the membrane currents evaluated at the boundary.

4.4.2 Four-derivative Einstein-Maxwell theory

In this section, we consider the most general four-derivative theory of massless gravitons and gauge fields. The action \mathcal{L}_A can be written as (see [219, 223, 230–232]):

$$\mathcal{L}_{A} = -\frac{1}{4}F_{ab}F^{ab} + \alpha_{4}RF_{ab}F^{ab} + \alpha_{5}R^{ab}F_{ac}F_{b}{}^{c} + \alpha_{6}R^{abcd}F_{ab}F_{cd} + \alpha_{7}(F_{ab}F^{ab})^{2} + \alpha_{8}\nabla_{a}F_{bc}\nabla^{a}F^{bc} + \alpha_{9}\nabla_{a}F_{bc}\nabla^{b}F^{ac} + \alpha_{10}\nabla_{a}F^{ab}\nabla^{c}F_{cb} + \alpha_{11}F^{ab}F_{bc}F^{cd}F_{da},$$

$$(4.61)$$

and similarly \mathcal{L}_V . Note that in Eq. (4.61), all indices A denoting that F_{ab} is the axial field strength have been suppressed. The conserved current two-form, H_5^{ab} , in this theory is

$$H_{5}^{ab} = -F^{ab} + 4\alpha_{4}RF^{ab} + 2\alpha_{5}(R^{ac}F_{c}^{\ b} - R^{bc}F_{c}^{\ a}) + 4\alpha_{6}R^{cdab}F_{cd} + 8\alpha_{7}F_{cd}F^{cd}F^{ab} - 4\alpha_{8}\Box F^{ab} - 2\alpha_{9}\nabla_{c}(\nabla^{a}F^{cb} - \nabla^{b}F^{ca}) + 2\alpha_{10}(\nabla^{b}\nabla_{c}F^{ca} - \nabla^{a}\nabla_{c}F^{cb}) + 8\alpha_{11}F^{bc}F_{cd}F^{da}.$$
(4.62)

The current \mathcal{J}_5^i is then

$$\mathcal{J}_{5}^{\mu} = \mathcal{J}_{5,\text{Maxwell}}^{\mu} + \sum_{n=4}^{11} \alpha_{n} \mathcal{J}_{5,(n)}^{\mu} + \mathcal{J}_{CS}^{\mu}, \qquad (4.63)$$

where $\mathcal{J}_{5,\text{Maxwell}}^{\mu}$ is the axial current that follows from the two-derivative Maxwell action analysed in Section 4.4.1. The remaining terms, $\mathcal{J}_{5,(n)}^{\mu}$, all have the schematic form

$$\mathcal{J}_{5,(n)}^{\mu} = \left[C_{n,1}h + C_{n,2}\partial_r h + C_{n,3}\partial_r^2 h + D_{n,1}\partial_r a + D_{n,2}\partial_r^2 a + D_{n,3}\partial_r^3 a \right] \omega^{\mu},$$
(4.64)

where the coefficients $C_{n,i}$ and $D_{n,i}$ depend on the background and parameters of the action. The full expressions for these coefficients are lengthy and will not be presented here.

Near the non-extremal horizon (assumed to exist), the metric must behave as in Eqs. (4.12) and (4.13). What we find is that when evaluated at the horizon, all coefficients except $C_{n,1}$ vanish. This result therefore precisely agrees with the structure of \mathcal{J}_5^{μ} predicted in (4.48), which followed from our general treatment of $H_5^{r\mu}$ in Section 4.3.2. At the horizon, the full set of $\mathcal{J}_{5,(n)}^{\mu}$ is given by

$$\begin{aligned} \mathcal{J}_{5,(4)}^{\mu}(r_{h}) &= -\frac{2r_{h}^{2}\sqrt{g_{1}}A_{t}'}{f_{1}^{3/2}} \left(20f_{1}g_{1} + 3f_{2}g_{1}r_{h} + f_{1}g_{2}r_{h}\right)h(r_{h})\,\omega^{\mu}, \\ \mathcal{J}_{5,(5)}^{\mu}(r_{h}) &= -\frac{r_{h}\sqrt{g_{1}}A_{t}'}{f_{1}^{3/2}} \left(14r_{h}f_{1}g_{1} + 2r_{h}^{2}g_{1}f_{2} + r_{h}^{2}f_{1}g_{2}\right)h(r_{h})\,\omega^{\mu}, \\ \mathcal{J}_{5,(6)}^{\mu}(r_{h}) &= -\frac{2r_{h}^{2}\sqrt{g_{1}}A_{t}'}{f_{1}^{3/2}} \left(8f_{1}g_{1} + 3r_{h}g_{1}f_{2} + r_{h}f_{1}g_{2}\right)h(r_{h})\,\omega^{\mu}, \\ \mathcal{J}_{5,(7)}^{\mu}(r_{h}) &= -\frac{16r_{h}g_{1}^{3/2}(A_{t}')^{3}}{f_{1}^{3/2}}h(r_{h})\,\omega^{\mu}, \\ \mathcal{J}_{5,(8)}^{\mu}(r_{h}) &= -\frac{28r_{h}^{3}\sqrt{g_{1}}}{f_{1}^{3/2}} \left(-g_{1}f_{2} + f_{1}g_{2} + 2f_{1}g_{1}A_{t}''/A_{t}'\right)h(r_{h})\,\omega^{\mu}, \\ \mathcal{J}_{5,(9)}^{\mu}(r_{h}) &= \frac{1}{2}\mathcal{J}_{5,(8)}^{\mu}, \\ \mathcal{J}_{5,(10)}^{\mu}(r_{h}) &= \frac{r_{h}^{2}\sqrt{g_{1}}}{f_{1}^{3/2}} \left(6f_{1}g_{1} - r_{h}g_{1}f_{2} + r_{h}f_{1}g_{2} + 2r_{h}f_{1}g_{1}A_{t}''/A_{t}'\right)h(r_{h})\,\omega^{\mu}, \\ \mathcal{J}_{5,(11)}^{\mu}(r_{h}) &= -\frac{1}{2}\mathcal{J}_{5,(7)}^{\mu}. \end{aligned}$$

$$(4.65)$$

Finally, imposing the horizon Ricci scalar regularity condition (see the discussion after Eq. (4.48)), $h(r_h) = 0$, we find that all $J_{5,(n)}^{\mu}(r_h) = 0$.

At the AdS boundary $(r \to \infty)$, we further find that all coefficients $C_{n,i} \sim r^{-m}$, where m > 0. With this explicit verification, our results imply that the most general gauge- and diffeomorphism - invariant four-derivative theory (4.61) satisfies the condition (4.34) and that the anomalous conductivities in its dual all have the universal form of Eq. (4.39).

4.4.3 Theories without horizons and theories with scaling geometries at zero temperature

In this section, we consider two classes of backgrounds, each one a possible solution of the Einstein-Maxwell-dilaton theory of Section 4.4.1. The first one belongs to the family of soft/hard wall model that are dual to a field theory with a mass gap [233–236]. The second example is the scaling geometry that can arise as a solution of the Einstein-Maxwell-dilaton theory at zero temperature (see e.g. [121]). What we show is that the criterion for the universality of anomalous conductivities, i.e. Eq. (4.34), is still satisfied in the gapped system. However, the conductivities can no longer computed by using the replacement rule in the form stated in Eq. (4.77). For the scaling geometries, the universality may be violated due to the presence of naked singularities. A way to retain a holographic theory at zero temperature in which the condition (4.34) is satisfied is to put very strong constraints on the geometry that avoid the naked singularity. These constraints restrict the allowed range of value of the hyperscaling violation exponent, θ , and the dynamical critical exponent, z.

Let us start with an example of the soft/hard wall geometry at zero density. In an un-boosted frame, the metric for these models can be written as

$$ds^{2} = e^{-(M/u)^{\nu}} \left(-u^{2}d\bar{t}^{2} + \frac{du^{2}}{u^{2}} + u^{2}(d\bar{x}^{2} + d\bar{y}^{2} + d\bar{z}^{2}) \right), \qquad (4.66)$$

where the parameter M sets the scale of the mass gap. The nature of the spectrum is also controlled by the parameter ν : while the gapped spectrum is continuous above the gap when $\nu = 1$, it is discrete when $\nu > 1$. The hard wall model in which the AdS radius is capped off at $u \ll M$ corresponds to the limiting value of $\nu \to \infty$ [235, 237].

One can change coordinates of the above metric to bring them to the form of (4.10) by redefining the radial coordinate as $r = e^{-\frac{1}{2}(M/u)^{\nu}}$. In the deep IR region, $u \ll M$, the functions f(r) and g(r) can be written as

$$f_{IR}(r) = 1, \qquad g_{IR}(r) = g\left(u \ll M\right) = \nu^2 \left(\frac{M}{u}\right)^{\nu} e^{M/u}.$$
 (4.67)

Despite there being no horizon, the dual of the above geometry can still have non-zero temperature; it can be interpreted as a thermal state before undergoing a phase transition to the black hole phase at high temperature, analogously to the Hawking-Page transition [238].

The two currents, \mathcal{J}_5^{μ} and \mathcal{J}^{μ} , must now be evaluated at r = 0 and at the boundary $(r = \infty)$. Because the geometry is still asymptotically AdS, their near-boundary behaviour is the same as in all the cases studied before. The fact that g(r) exponentially diverges in the IR appears problematic at first. However, the volume form, which is proportional to $\sqrt{-G}$, is exponentially suppressed. Evaluating \mathcal{J}_5^i at u = 0, one finds that $\mathcal{J}_{5,mb}^{\mu}(0) + \mathcal{J}_{5,r}^{\mu}(0) = 0$ as in 4.3.2. Thus, the universality condition (4.34) is still satisfied.

On the other hand, the Chern-Simons current $\mathcal{J}_{5,CS}^{\mu}$ no longer behaves the same way. Although the profiles of the gauge fields A_t , and V_t can be assumed to asymptote to a constant value at r = 0, the derivative of f' can no longer be interpreted as the temperature of the dual theory (substituting (4.67) into (4.29), we see that $\mathcal{J}_{5,CS}^{\mu}$ has no temperature dependence). Therefore, in the confining phase, the replacement rules discussed in the Appendix 4.6.1 are no longer applicable even if the condition (4.34) is satisfied. The above statements also apply to the AdS soliton-like geometries.

Next, we explore the scaling geometries at zero temperature. In the unboosted frame, the metric can now be written as

$$ds^{2} = r^{2} \left(-r^{n_{0}} dt^{2} + d\bar{x}^{2} + d\bar{y}^{2} + d\bar{z}^{2} \right) + \frac{dr^{2}}{r^{n_{1}}},$$
(4.68)

or in terms of θ and z,

$$n_0 = 2 + \frac{6(z-1)}{3-\theta}, \qquad n_1 = 2 + \frac{2\theta}{3-\theta}.$$
 (4.69)

As mentioned in [121], many of these geometries contain a naked singularity. As a result, the scalars $\{S_{A,n}, S_{G,n}\}$ used in Eq. (4.47) no longer have to be finite. Such systems can therefore easily violate the universality condition (4.34). Thus, the universality of the anomalous conductivities is no longer guaranteed in the presence of a naked singularity. In this work, we do not study in de-
tail what happens to anomalous conductivities in such cases and whether they nevertheless remain universal for some geometries.

Are there special values of z and θ for which it is easy to see that the condition (4.34) remains satisfied? In other words, what are the ranges of $\{z, \theta\}$ for which the theory has no naked singularity? This problem was addressed in [239], where it was found that the geometries that satisfy either one of the following two conditions,

$$n_0 = n_1 = 2, \qquad n_0 = n_1 \ge 4,$$
 (4.70)

have no naked singularities. The authors assumed that the matter content has to satisfy the null energy condition, which, for this geometry, is equivalent to imposing the following two inequalities:

$$n_0 \ge n_1, \qquad (n_0 - 2)(n_0 + n_1 + 4) \ge 0.$$
 (4.71)

The first solution in (4.70) is simply the empty AdS solution with z = 1 and $\theta = 0$. The second solution (or a family of solutions) is more involved and requires non-trivial matter to support such geometries.

Of particular interest are charged dilatonic black holes with $z \to \infty$, $\theta \to -\infty$ and a fixed ratio $-\theta/z = \eta$, dual to strongly interacting theories with finite density (see e.g. [240, 241]). While such systems still satisfy the null energy condition, the geometries nevertheless exhibit a naked singularity at zero temperature. This means that unless there is a way to resolve the singularity, the universal structure of anomalous conductivities, although not necessarily, may be violated at zero temperature for all values of η . One way to resolve this issue, as mentioned in [240] for $\eta = 1$, is to lift the black hole solution to a ten- or eleven- dimensional solution of string or M-theory [242]. To study such solutions, one also needs to find the ten- and eleven-dimensional analogues of the Chern-Simons terms (\mathcal{L}_{CS} in (4.9)). In case of a supergravity setup, this was studied in [243] and many subsequent works. An explicit computation of chiral magnetic conductivity, σ_{JB} , in a top-down setup of probe flavour branes can be found in [244]. More generally, it is plausible that the problems of IR

singularities can be avoided when they are of the "good type" [245].¹¹ We defer a more detailed study of these issues and of top-down constructions to future works.

4.4.4 Bulk theories with massive vector fields

In this section, we comment on the universality of anomalous conductivities in field theories with mixed, gauge-global anomalies. Such theories exhibit the following anomalous Ward identity:

$$\partial_{\mu} \langle J_{5}^{\mu} \rangle = \beta \epsilon^{\mu\nu\rho\sigma} \mathcal{F}_{\mu\nu} \mathcal{F}_{\rho\sigma} + (\text{global anomaly terms}), \qquad (4.72)$$

where $\mathcal{F}_{\mu\nu}$ is the field strength of the gluon fields (e.g. in QCD). The global anomaly terms were stated in Eq. (4.1). As shown by perturbative quantum field theory calculations [194, 196, 208, 209], the anomalous conductivities in such theories are known to be renormalised, i.e. they receive quantum corrections.

Holographic models dual to theories with the anomalous Ward identity of the form of Eq. (4.72) were proposed and studied in [88, 243, 246, 247]. In this work, we focus on the bottom-up construction of [247], where the following terms are added to the bulk action (4.8):

$$\Delta S = \int d^5 x \sqrt{-G} \left(-\frac{m^2}{2} (A_a - \partial_a \theta) (A^a - \partial^a \theta) - \frac{\kappa}{3} \epsilon^{abcde} (\partial_a \theta) F_{bc} F_{de} \right)$$
(4.73)

We have set the vector and the gravitational Chern-Simons terms to zero, i.e. $\gamma = \lambda = 0$ (see Eq. (4.1)). The scalar field θ is the Stückelberg axion.

A holographic theory with ΔS in the action can clearly evade the arguments of the proof of universality from Section 4.3. The reason is that the equation of motion for a massive vector field cannot be written in the form of Eq. (4.15). The right-hand-side of (4.15) now contains terms which explicitly depend on A_a and one cannot reduce the equations into a total derivative form, $\partial_r \mathcal{J}^{\mu} = 0$. Hence, in models with massive vector fields, dual to field theories with mixed, gauge-global anomalies, anomalous conductivities can be renormalised. This is

¹¹We thank Umut Gürsoy for discussions on this point.

consistent with field theory calculations mentioned above.

4.5 Discussion

In this work, we studied the coupling constant dependence of the universality of chiral conductivities associated with the anomalous axial and vector currents in holographic models with arbitrary higher-derivative actions of the metric, gauge fields and scalars. We showed that so long as the action was gauge- and diffeomorphism- invariant, the membrane paradigm construction for the chiral conductivities remained valid, resulting in universal chiral conductivities (see Eq. (4.39)). The proof assumed the existence of a regular, non-extremal black brane with an asymptotically AdS geometry. This result is valid for an infiniteorder expansion of coupling constant corrections to holographic results at infinite coupling. Hence, it is complementary to perturbative field theory proofs (expanded around zero coupling) of the non-renormalisation of chiral conductivities in systems with global anomalies and therefore of the anomalous Ward identities with the form of Eq. (4.1). Furthermore, this chapter also explored cases which may violate universality, in particular, in cases with naked singularities and massive vector fields that explicitly violate Eq. (4.1) through mixed, gauge-global anomalies.

This work provides a consistency test of holography in its ability to reproduce the expected non-renormalisation of global Ward identities at the level of (non-zero temperature and density) transport in very general bulk constructions that include arbitrary higher-derivative actions. Furthermore, we believe that the methods presented in this work can be of wider use to other holographic statements of universality that employ the membrane paradigm.

We end this chapter by listing some problems that are left to future works. Most importantly, there exists another anomalous conductivity in the stressenergy tensor, which can be sourced by a small vortex, $\delta T^{\mu\nu} = \sigma^{\epsilon} u^{(\mu} \omega^{\nu)}$. The analysis of this conductivity was not performed in this work. In the fluidgravity framework, σ^{ϵ} was studied in the Einstein-Maxwell theory by [204]. Forming a conserved bulk current for computing components of the stressenergy tensor tends to be significantly more complicated than for those of a boundary current. However, it may be possible to achieve this by using the Hamiltonian methods recently employed for the calculations of the thermoelectric DC conductivities [129, 175, 248] in two-derivative theories, which should be extended to computations of anomalous transport in higher-derivative theories.

One may also wonder what happens to anomalous transport in inhomogeneous and anisotropic systems. In standard non-anomalous transport, it is known that universal relations can be violated, e.g. in η/s [40, 43, 249, 250]. While analysing such systems is in general significantly more difficult, the existence of the membrane paradigm, as e.g. in case of the DC thermo-electric conductivities [129, 175, 176], may still enable one to prove general statements about the behaviour of conductivities in disordered systems [38, 39]. These methods remain to be explored in the context of anomalous transport.

In even-dimensional theories, anomalous conductivities are directly related to the parity-odd hydrodynamic constitutive relation of [85, 195, 206]. These parity-odd terms are related to global anomalies. In odd dimension, one can still construct hydrodynamics with parity-odd terms, as e.g. in [251]. A wellknown parity-odd transport coefficients is the Hall viscosity [252, 253]. This quantity has relations to topological states of matter, such as fractional quantum Hall systems (see e.g. [254] and references therein). A holographic theory with non-zero Hall viscosity can be obtained by adding a topological term similar to the dimensionally-reduced gravitational Chern-Simons term [255]. Recently, in [150], the constitutive relation term associated with the Hall viscosity was generalised to a class of hydrodynamic terms that resemble the Berry curvature. Despite these similarities, there is no known non-renormalisation theorem for parity-odd transport coefficients in odd dimensions.

Lastly, we point out that many recent works have found novel structures in entanglement entropy of theories with anomalies [256–258, 258–260]. As a result of non-renormalisation, one may expect there to exist strong constraints on the structure of extremal bulk surfaces associated with entanglement entropy. It would be interesting to better understand the connection between geomet-

ric constraints on holographic entanglement entropy and non-renormalisation theorems for anomalies.

4.6 Appendices

4.6.1 Anomaly polynomials and the replacement rule

As noted in the Introduction, the full set of chiral conductivities (4.2) can be encoded in the anomaly polynomial defined in terms of the Chern-Simons action [73, 74, 202, 203]:

$$\mathcal{P}(F,R) = dS_{CS}[A,\Gamma]. \tag{4.74}$$

If we restrict ourselves only to global anomalies in four spacetime dimensions, then the anomaly polynomial can be written as

$$\mathcal{P} = \frac{\kappa}{3} \left(F_A \wedge F_A \wedge F_A \right) + \gamma \left(F_A \wedge F_V \wedge F_V \right) + \lambda \left(F_A \wedge R^{\mu}_{\ \nu} \wedge R^{\nu}_{\ \mu} \right).$$
(4.75)

The replacement rule states that, for an anomaly polynomial \mathcal{P} , one can define the generating function $\mathcal{G}[\mu_5, \mu, T]$:

$$\mathcal{G}\left[\mu_{5}, \mu, T\right] = \mathcal{P}\left[F_{A} \to \mu_{5}, F_{V} \to \mu, \operatorname{Tr} R^{2} \to 2(2\pi T)^{2}\right], \qquad (4.76)$$

where T is the temperature and μ_5 , and μ are chemical potentials associated with the axial and the vector currents J_5^{μ} and J^{μ} . The anomalous conductivities can then be computed by using

$$\sigma_{J_5B} = -\frac{\partial^2 \mathcal{G}}{\partial \mu_5 \partial \mu}, \qquad \sigma_{JB} = -\frac{\partial^2 \mathcal{G}}{\partial \mu \partial \mu},$$

$$\sigma_{J_5\omega} = \frac{\partial \mathcal{G}}{\partial \mu_5}, \qquad \sigma_{J\omega} = \frac{\partial \mathcal{G}}{\partial \mu}. \qquad (4.77)$$

For the anomaly polynomial in (4.75), the anomalous conductivities are precisely those stated in Eq. (4.39).

In the work of [204], the replacement rule (4.76) with (4.77) was derived for a field theory dual to the AdS Reissner-Nördstrom background. Our work can

Chapter 4. Universality of anomalous conductivities in theories with higher-derivative holographic duals

be seen a check of the validity of this replacement rule prescription for more general, higher-derivative holographic theories.

5

Magnetohydrodynamic waves in a strongly interacting holographic plasma

5.1 Introduction

Magnetohydrodynamics (MHD) is a hydrodynamic theory of long-range excitations in plasmas (ionised gases) (see e.g. [261, 262]), which has been applied to systems ranging from the physics of fusion reactors to astrophysical objects. In the modern language of hydrodynamics formulated as an effective field theory [63, 263–276], MHD should describe the dynamics of infrared (IR) chargeneutral states in terms of massless effective degrees of freedom. These plasma ground states are characterised by an equation of state with a finite magnetic field. The electric field is suppressed due to the screening of electromagnetic interactions and is only induced on shorter length scales than the (thermodynamic) size of the system. In their standard form, the equations of motion that

describe the evolution of plasmas are formulated as a combination of macroscopic fluid equations (continuity equation and the non-dissipative Euler, or dissipative Navier-Stokes equation), coupled to the microscopic electromagnetic Maxwell's equations. In ideal, non-dissipative form, the set of dynamical equations is

$$\partial_t \rho + \vec{\nabla} \cdot (\rho \, \vec{v}) = 0 \,, \tag{5.1}$$

$$\rho\left(\partial_t + \vec{v} \cdot \nabla\right) \vec{v} = -\vec{\nabla} \, p + \vec{J} \times \vec{B} \,, \tag{5.2}$$

$$\partial_t \vec{B} = \vec{\nabla} \times \left(\vec{v} \times \vec{B} \right), \tag{5.3}$$

$$\left(\partial_t + \vec{v} \cdot \nabla\right) \left(\frac{p}{\rho^{\gamma}}\right) = 0.$$
(5.4)

The magnetic field is constrained by

$$\vec{\nabla} \cdot \vec{B} = 0. \tag{5.5}$$

Eq. (5.1) is the continuity equation and Eq. (5.2) the Euler equation in the presence of the Lorentz force $\vec{J} \times \vec{B}$, with \vec{J} given by the low-frequency limit of the Ampere's law $(\partial_t \vec{E} \to 0)$

$$\vec{J} = \frac{1}{\mu_0} \vec{\nabla} \times \vec{B} \,. \tag{5.6}$$

Eq. (5.3) is the Faraday's induction law with the electric field fixed by the assumption of the ideal Ohm's law

$$\vec{E} + \vec{v} \times \vec{B} = 0, \qquad (5.7)$$

which is derived by taking the conductivity in the (Lorentz transformed) Ohm's law $\vec{J}/\sigma = \vec{E} + \vec{v} \times \vec{B}$ to infinity, i.e. $\sigma \to \infty$. The constraint equation (5.5) is the magnetic Gauss's law. Since the ideal Ohm's law completely fixes \vec{E} , the electric Gauss's law plays no role in the equations of MHD. Eq. (5.4) is the adiabatic equation of state relating density and pressure. Usually, one takes $\gamma = 5/3$. Altogether, Eqs. (5.1)–(5.4) give eight dynamical equations for eight unknown functions ρ , p, \vec{v} and \vec{B} , subject to the magnetic field constraint (5.5).

While the above equations are closed, solvable and have been successfully

applied to a variety of phenomena in plasma physics, they are only applicable within the specific assumptions used to construct them. This means that they are only valid for electromagnetism controlled by Maxwell's equations in the limit of ideal Ohm's law (no possibility of strong-field pair production, etc.) and for the specific equation of state in Eq. (5.4). This equation of state encodes a separation between the fluid and the charge carrying sectors, for which the justification, beyond assuming weakly coupled Maxwell electromagnetism, also assumes very weak interactions between the fluid degrees of freedom and electromagnetism inside the plasma. Concretely, the latter statement is reflected in the equation of state permitting no dependence on the magnetic properties controlled by the charged sector. Furthermore, because of a lack of a symmetry principle behind the construction of ideal MHD, these equations are difficult to extend unambiguously to the most general, higher-order, dissipative theory in the gradient expansion (the Knudsen number expansion) [50, 64, 110, 184].¹ As such, the traditional formulation of MHD lacks generality and cannot be compatible with a variety of IR effective theories of plasmas that could (in principle) be derived from quantum field theory, in particular, in the presence of a strong magnetic field.

These issues were addressed in a recent work [47], where MHD was formulated by following the effective field theory philosophy behind the construction of relativistic hydrodynamics (see e.g. [51, 184]). Namely, MHD was formulated by only considering global conserved operators and writing them in terms of the most general hydrodynamic gradient expansion of the IR hydrodynamic fields [47].² With such an expansion in hand, conservation equations then completely determine the temporal dynamics of a plasma with any equation of state. As in hydrodynamics, all of the details of the equation of state and transport coefficients are left to be determined by the microscopics of the underlying the-

 $^{^{1}}$ We note that in standard MHD, as formulated in Eqs. (5.1)–(5.4), only the fluid sector has a well-defined and finite Knudsen number.

²See also [277] and Ref. [278], which includes a valuable comparison of various related past works, such as [279–281]. For a new treatment of charged fluids in an external electromagnetic field, see [278, 282]. Of further interest is also a recently proposed field theory description of polarised fluids [283].

ory.

The two relevant global symmetries describing the long-range dynamics of a plasma were argued to give the stress-energy tensor $T^{\mu\nu}$ and a conserved anti-symmetric two-form current $J^{\mu\nu}$ [47]:

$$\nabla_{\mu}T^{\mu\nu} = H^{\nu}_{\ \mu\sigma}J^{\mu\sigma}\,,\tag{5.8}$$

$$\nabla_{\mu}J^{\mu\nu} = 0. \tag{5.9}$$

While $T^{\mu\nu}$ corresponds to conserved energy-momentum, $J^{\mu\nu}$ is the manifestation of a generalised global U(1) symmetry, which can be sourced (and gauged) by a two-form gauge field $b_{\mu\nu}$ [46]. $H^{\nu}_{\mu\sigma}$ is a three-form field strength that can be turned on by an external two-form gauge field, $H = db_{ext}$. This generalised global symmetry is a consequence of the absence of magnetic monopoles and directly corresponds to the conserved number of magnetic flux lines crossing a co-dimension two surface (in a four dimensional plasma). Normally, it is expressed in terms of the (topological) Bianchi identity

$$dF = 0, (5.10)$$

where F = dA and A is the abelian electromagnetic field. In the language of a two-form current used in Eq. (5.9),

$$J^{\mu\nu} = \frac{1}{2} \epsilon^{\mu\nu\rho\sigma} F_{\rho\sigma} \,. \tag{5.11}$$

Eqs. (5.8) and (5.9) give seven dynamical equations of motion (and one constraint). To solve the equations, we introduce the following hydrodynamical fields: a velocity field u^{μ} , a temperature field T, a chemical potential μ that corresponds to the density of magnetic flux lines and a vector h^{μ} , which can be though as a hydrodynamical realisation of a fluctuating magnetic field. The vectors are normalised as $u_{\mu}u^{\mu} = -1$, $h_{\mu}h^{\mu} = 1$, $u_{\mu}h^{\mu} = 0$, together resulting in 10 - 3 = 7 degrees of freedom. The velocity flow of the plasma breaks the Lorentz symmetry from SO(3, 1) to SO(3), which is further broken by the additional vector (magnetic field) to SO(2).³ The projector transverse to both

³Note that at zero temperature, in a plasma with a non-fluctuating temperature field, the

 u^{μ} and h^{μ} is defined as $\Delta^{\mu\nu} = g^{\mu\nu} - u^{\mu}u^{\nu} - h^{\mu}h^{\nu}$ and has a trace $\Delta^{\mu}_{\ \mu} = 2$.

The constitutive relations for the conserved tensors with a positive local entropy production [63] and charge conjugation symmetry can now be expanded to first order in derivatives as [47]

$$T^{\mu\nu} = (\varepsilon + p) u^{\mu} u^{\nu} + p g^{\mu\nu} - \mu \rho h^{\mu} h^{\nu} + \delta f \Delta^{\mu\nu} + \delta \tau h^{\mu} h^{\nu} + 2 \ell^{(\mu} h^{\nu)} + t^{\mu\nu}$$
(5.12)

$$J^{\mu\nu} = 2\rho \, u^{[\mu} h^{\nu]} + s^{\mu\nu} \,, \tag{5.13}$$

where

$$\delta f = -\zeta_{\perp} \Delta^{\mu\nu} \nabla_{\mu} u_{\nu} - \zeta_{\times}^{(1)} h^{\mu} h^{\nu} \nabla_{\mu} u_{\nu} , \qquad (5.14)$$

$$\delta\tau = -\zeta_{\times}^{(2)}\Delta^{\mu\nu}\nabla_{\mu}u_{\nu} - \zeta_{\parallel}h^{\mu}h^{\nu}\nabla_{\mu}u_{\nu}, \qquad (5.15)$$

$$\ell^{\mu} = -2\eta_{\parallel} \Delta^{\mu\sigma} h^{\nu} \nabla_{(\sigma} u_{\nu)} , \qquad (5.16)$$

$$t^{\mu\nu} = -2\eta_{\perp} \left(\Delta^{\mu\rho} \Delta^{\nu\sigma} - \frac{1}{2} \Delta^{\mu\nu} \Delta^{\rho\sigma} \right) \nabla_{(\rho} u_{\sigma)} , \qquad (5.17)$$

$$m^{\mu} = -2r_{\perp}T\Delta^{\mu\beta}h^{\nu}\nabla_{[\beta}\left(\frac{h_{\nu]}\mu}{T}\right), \qquad (5.18)$$

$$s^{\mu\nu} = -2r_{\parallel}\mu\Delta^{\mu\rho}\Delta^{\nu\sigma}\nabla_{[\rho}h_{\sigma]}\,.$$
(5.19)

The thermodynamic relations between ε , p and ρ , which need to be obeyed by the equation of state $p(T, \mu)$ are

$$\varepsilon + p = sT + \mu\rho, \qquad (5.20)$$

$$dp = s \, dT + \rho \, d\mu \,. \tag{5.21}$$

Furthermore, for the theory to be invariant under time-reversal, the Onsager relation implies that $\zeta_{\times}^{(1)} = \zeta_{\times}^{(2)} \equiv \zeta_{\times}$. Thus, first-order dissipative corrections to ideal MHD are controlled by seven transport coefficients: η_{\perp} , η_{\parallel} , ζ_{\perp} , ζ_{\parallel} , ζ_{\times} , r_{\perp} and r_{\parallel} . Each one can be computed from a set of Kubo formulae presented in [47, 278] and reviewed in Appendix 5.6.1. The transport coefficients should obey the following positive entropy production constraints: $\eta_{\perp} \geq 0$, $\eta_{\parallel} \geq 0$,

symmetry is enhanced to $SO(1,1) \times SO(2)$ [47].

 $r_{\perp} \geq 0, r_{\parallel} \geq 0, \zeta_{\perp} \geq 0$ and $\zeta_{\perp}\zeta_{\parallel} \geq \zeta_{\times}^2$. In absence of charge conjugation symmetry, the theory has four additional transport coefficients, resulting in total in eleven transport coefficients [278]. The precise connection between the above formalism of MHD using the concept of generalised global symmetries and MHD expressed in terms of electromagnetic fields, which match in the limit of a small magnetic field, was established in Ref. [278].

Since the effective theory [47] makes no assumption regarding the microscopic details of the plasma, then, should such details somehow be computable from quantum field theory, or otherwise, the effective MHD can be used in solar plasma physics, fusion reactor physics, astrophysical plasma physics and even QCD quark-gluon plasma resulting from nuclear collisions. Of course, computing the microscopic properties of such systems is extremely difficult. In this work, we will resort to using holographic duality. By using standard holographic methods applicable to hydrodynamics [103–105], our analysis will provide us with the required microscopic data of a strongly interacting toy model plasma needed to describe the phenomenology of MHD waves.

The paper is structure as follows: first, in Section 5.2 we review important aspects of gauge theories with a sector coupled to dynamical U(1), which can describe a plasma in the IR limit. In particular, we focus on the discussion of how to couple a strongly interacting field theory with a holographic dual to dynamical electromagnetism, all within a holographic setup. Then, in Section 5.3, we explore this holographic setup in detail, develop the holographic dictionary and use it to compute the microscopic properties of the dual plasma, i.e. the equation of state and first-order transport coefficients. In Section 5.4, we then use this data to analyse the phenomenology of propagating MHD modes—Alfvén and magnetosonic waves. Finally, we conclude with a discussion and a summary of the most important findings in Section 5.5. Three appendices are devoted to a derivation of the relevant Kubo formulae (Appendix 5.6.1), details regarding the derivation of the magnetosonic dispersion relations (Appendix 5.7) and a derivation of the magnetosonic dispersion relations (Appendix 5.8).

5.2 Matter coupled to electromagnetic interactions

A microscopic theory from which an effective description of a plasma can arise comprises of a matter sector that interacts through an electromagnetic U(1)gauge field. The simplest example of such a theory is quantum electrodynamics. In other theories, the matter sector may itself exhibit complicated physics with additional gauge interactions, such as in QCD. In this work, the theory that we will study contains an infinitely strongly coupled holographic matter sector (closely related to $\mathcal{N} = 4$ supersymmetric $SU(N_c)$ Yang-Mills) with infinite N_c . Because of the coupling between matter and dynamical electromagnetism, the holographic setup and the interpretation of results is somewhat subtle. For this reason, we begin our discussion by reviewing some relevant aspects of quantum field theory in a line of arguments similar to [284].

5.2.1 Quantum electrodynamics

The simplest example of a theory coupling matter to electromagnetism is quantum electrodynamics (QED). QED is a U(1) gauge theory that contains a (massive) Dirac fermion ψ (describing electrons and positrons) and a massless photon field A_{μ} :⁴

$$S_{QED} = -\int d^4x \left[i\bar{\psi}\gamma^{\mu}D_{\mu}\psi + m\bar{\psi}\psi + \frac{1}{4e^2}F_{\mu\nu}F^{\mu\nu} \right].$$
 (5.22)

 D_{μ} is the gauge covariant derivative that couples A_{μ} to the fermion current (with the coupling *e* scaled out from the interaction). For a detailed discussion of various properties of QED, see e.g. [189, 285, 286].

The stress-energy tensor of the theory is

$$T^{\mu\nu} = \frac{1}{2} \bar{\psi} i \left(\gamma^{\mu} D^{\nu} + \gamma^{\nu} D^{\mu} \right) \psi - \eta^{\mu\nu} \bar{\psi} \left(i \gamma^{\mu} D_{\mu} + m \right) \psi$$
 (5.23)

$$+\frac{1}{e^2}\left[F^{\mu\lambda}F^{\nu}_{\ \lambda}-\frac{1}{4}\eta^{\mu\nu}F^{\rho\sigma}F_{\rho\sigma}\right].$$
(5.24)

In the massless limit (m = 0), the theory is classically scale invariant, which is

⁴We use the mostly positive convention for the metric tensor, so that $\eta_{\mu\nu} = \{-1, +1, +1, +1\}$.

reflected in the vanishing trace of the stress-energy tensor, $T^{\mu}_{\ \mu} = 0$. Quantum mechanically, the theory does not remain scale invariant. The trace receives a correction proportional to the beta function of the electromagnetic coupling

$$T^{\mu}_{\ \mu} = -\frac{\beta(e)}{2e^3} F_{\mu\nu} F^{\mu\nu} \,. \tag{5.25}$$

This is the anomalous breaking of scale invariance—the so-called trace anomaly. The running electromagnetic coupling $e(\mu)$ depends on the renormalisation group scale μ . To first order in perturbation theory, the beta function is

$$\beta(e) = \mu \frac{de}{d\mu} = \frac{e^3}{12\pi^2},$$
(5.26)

which, integrated on the interval $\mu \in [M, \Lambda]$, gives the running coupling

$$\frac{1}{e(\Lambda)^2} = \frac{1}{e(M)^2} - \frac{\ln(\Lambda/M)}{6\pi^2}.$$
 (5.27)

Here, M is some IR RG scale at which the electric charge takes the renormalised physical value, $e_r = e(M)$, and Λ is the UV cut-off. Note that at the Landau pole, $\Lambda = \Lambda_{EM}$, the left-hand-side of (5.27) vanishes. On the other hand, the expectation value of the stress-energy tensor is a physical quantity and therefore cannot depend on μ . This statement is encoded in the following identity, which leads to the Callan-Symanzik equation:

$$\mu \frac{d}{d\mu} \left\langle T^{\mu\nu} \right\rangle = 0 \,. \tag{5.28}$$

Since we are interested in neutral IR plasma states in QED that can be described by an effective theory of MHD, we can consider the expectation value of the photon field to produce a non-zero magnetic field and zero electric field,

$$\langle A_{\mu} \rangle = \frac{1}{2} \mathcal{B} \left(x^1 \delta^2_{\ \mu} - x^2 \delta^1_{\ \mu} \right).$$
(5.29)

 \mathcal{B} is the magnitude of the "background" magnetic field in the $x^3 = z$ direction. The IR spectrum of the theory has a gapped-out photon, i.e. long-range charge neutrality, which allows us to neglect quantum fluctuations of A_{μ} . For such a

plasma state, Eq. (5.25) yields

$$\langle T^{\mu}_{\nu} \rangle = -\frac{\beta(e)}{e^3} \mathcal{B}^2 = -\frac{1}{12\pi^2} \mathcal{B}^2 + \mathcal{O}\left(e^2\right).$$
 (5.30)

Furthermore, the expectation value of the stress-energy tensor can be conveniently split into the matter (containing matter-light interactions) and the purely electromagnetic parts

$$\langle T^{\mu\nu} \rangle = \langle T^{\mu\nu}_{matter}(\mu) \rangle + \frac{1}{e(\mu)^2} \left[F^{\mu\lambda} F^{\nu}_{\lambda} - \frac{1}{4} \eta^{\mu\nu} F^{\rho\sigma} F_{\rho\sigma} \right]$$

$$= \langle T^{\mu\nu}_{matter}(\Lambda/M) \rangle + \left(\frac{1}{e_r^2} - \frac{\ln\left(\Lambda/M\right)}{6\pi^2} \right) \frac{\mathcal{B}^2}{2} \times \begin{bmatrix} 1 & 0 & 0 & 0 \\ 0 & 1 & 0 & 0 \\ 0 & 0 & 1 & 0 \\ 0 & 0 & 0 & -1 \end{bmatrix},$$
(5.31)

where in the second line, we chose to evaluate the expectation value at the UV cut-off $\mu = \Lambda$. Note that because $\langle T^{\mu\nu} \rangle$ is μ -independent (cf. (5.28)), this choice does not influence the final value of $\langle T^{\mu\nu} \rangle$.

5.2.2 Strongly interacting holographic matter coupled to dynamical electromagnetism

We now turn our attention to the holographic strongly interacting theory that will be investigated in the remainder of this paper. Throughout our discussion, it will prove useful to think of the matter sector as that of the best understood holographic example—the conformal $\mathcal{N} = 4$ supersymmetric Yang-Mills theory (SYM) with an infinite number of colours N_c and an infinite 't Hooft coupling λ . However, as will become clear below, the theory dual to our holographic setup will not be precisely the $\mathcal{N} = 4$ SYM coupled to a U(1) gauge field, but rather its deformation, of which the microscopic definition will not be investigated in detail. Instead, the model studied here should be considered as a bottom-up construction—the simplest dual of a strongly coupled plasma, which can be described with magnetohydrodynamics in the infrared limit.

The field content of $\mathcal{N} = 4$ SYM are four Weyl fermions, three complex scalars and a vector field, all transforming under the adjoint representation

of $SU(N_c)$. The theory also has an $SU(4)_R$ R-symmetry owing to its extended supersymmetry. The adjoint fields together represent the matter content of a hypothetical plasma, which further requires the fields to be (minimally) coupled to an electromagnetic U(1) gauge group (with e the electromagnetic coupling). In $\mathcal{N} = 4$ SYM, this can be achieved by gauging the $U(1)_R$ subgroup of $SU(4)_R$. Under $U(1)_R$, the Weyl fermions transform with the charges $\{+3, -1, -1, -1\}/\sqrt{3}$ and the complex scalars all have charge $+2/\sqrt{3}$ (for details regarding the choice of the normalisation, see [284]). Such a system can be considered as a strongly coupled toy model for a QCD plasma in which the quarks interact with photons as well as with the SU(3) vector gluons.

A crucial fact about $\mathcal{N} = 4$ SYM is that the *R*-current of $\mathcal{N} = 4$ becomes anomalous in the presence of electromagnetism. For this reason, the $U(1)_R$, which is gauged, is also anomalous and thus the theory has to be deformed in some way to reestablish its self-consistency. As pointed out in [284], one way to do this is by adding a set of spectator fermions that only interact electromagnetically and "absorb" the anomaly. We will assume that the gauge anomaly can be cancelled by some way of deforming the theory, so that the quantum expectation value of the $U(1)_R$ R-current J^{μ}_R remains conserved, $\nabla_{\mu} \langle J^{\mu}_R \rangle = 0$. We can then write the total bare action of the $SU(N_c) \times U(1)$ gauge theory as

$$S_{plasma} = S_{matter} + \int d^4x \, A_\mu J_R^\mu - \frac{1}{4e^2} \int d^4x \, F_{\mu\nu} F^{\mu\nu} \,, \tag{5.32}$$

where A_{μ} is the dynamical electromagnetic gauge field and F = dA. The expectation value of the conserved operator J_R^{μ} contains a trace over the colour index of the adjoint matter field and therefore scales as N_c^2 . Since it is coupled to a single photon, the Maxwell part of the total plasma action S_{plasma} contains no powers of N_c .

As in the QED plasma, we will consider the photons to be gapped out from the IR spectrum so that A_{μ} will only produce a (classical) magnetic field

$$\langle A_{\mu} \rangle = \frac{1}{2} \mathcal{B} \left(x^1 \delta^2_{\ \mu} - x^2 \delta^1_{\ \mu} \right).$$
(5.33)

In order to maintain the neutrality of the plasma, we will set the electric $U(1)_R$

chemical potential to zero, $\mu_R = \langle A_0 \rangle = 0.5$ For this reason, the electric one-form (or vector) conserved $U(1)_R$ R-current will play no role in the hydrodynamic IR limit of the theory, so $\langle J_R^{\mu} \rangle = 0$.

The plasma has a conserved stress-energy tensor to which both the matter (along with its interaction with the electromagnetic field) and the purely electromagnetic sectors contribute,

$$\langle T^{\mu\nu} \rangle = \langle T^{\mu\nu}_{matter}(\Lambda/M) \rangle + \frac{1}{e(\Lambda/M)^2} \left[\langle F^{\mu\lambda} F^{\nu}_{\lambda} \rangle - \frac{1}{4} \eta^{\mu\nu} \langle F^{\rho\sigma} F_{\rho\sigma} \rangle \right].$$
(5.34)

The trace of the superconformal theory again experiences an anomaly proportional to the beta function of the electromagnetic coupling (cf. Eq. (5.30)), which in $\mathcal{N} = 4$ theory turns out to be one-loop exact in the presence of a background electromagnetic field and follows from a special case of the NSVZ beta function (see Refs. [284, 289, 290]),⁶

$$\langle T^{\mu}_{\ \mu} \rangle = -\frac{\beta(e)}{e^3} \mathcal{B}^2 = -\frac{N_c^2}{4\pi^2} \mathcal{B}^2 \,.$$
 (5.35)

The beta function for the inverse electromagnetic coupling is then

$$\beta \left(1/e^2 \right) = \mu \frac{de^{-2}}{d\mu} = -\frac{N_c^2}{2\pi^2} \left[\frac{1}{6} \sum_{\alpha=1}^4 \left(q_{\rm f}^{\alpha} \right)^2 + \frac{1}{12} \sum_{a=1}^3 \left(q_{\rm s}^{a} \right)^2 \right] = -\frac{N_c^2}{2\pi^2}, \tag{5.36}$$

with the fermionic and the scalar R-charges being $q_{\rm f}^{\alpha} = \{+3, -1, -1, -1\}/\sqrt{3}$ and $q_{\rm s}^{\alpha} = \{2, 2, 2\}/\sqrt{3}$, respectively. In analogy with Eq. (5.27) in QED, by integrating the beta function equation, we find

$$\frac{1}{e^2(\Lambda)} = \frac{1}{e^2(M)} - \frac{N_c^2}{2\pi^2} \ln\left(\Lambda/M\right).$$
(5.37)

It is essential to stress that even though our holographic theory will not be exactly dual to the $\mathcal{N} = 4$ SYM theory, it will give us the same trace anomaly

⁵For a discussion of supersymmetric gauge theories with non-zero R-charge densities, see e.g. [287, 288]

⁶Note that as $N_c \to \infty$, $N_c^2 - 1 \approx N_c^2$.

and thus the same electromagnetic beta function. Since the NSVZ beta function (5.36) is only sensitive to the matter content, this match can be interpreted as our working with a theory with the U(1)-gauged matter content and R-charges of $\mathcal{N} = 4$ but with a deformed Lagrangian and possibly additional matter that is ungauged under the U(1).

Beyond the stress-energy tensor of the theory discussed thus far, the only other (generalised) global symmetry of interest to describing a plasma state is the higher-form U(1) that corresponds to the conserved number of magnetic flux lines crossing a two-surface. The symmetry results in a conserved two-form current $\langle J^{\mu\nu} \rangle \neq 0$ and was discussed in Section 5.1. The generating function of the field theory that can be used to study MHD of a magnetised plasma in which the two globally conserved operators are $T^{\mu\nu}$ and $J^{\mu\nu}$ is

$$W\left[g_{\mu\nu}, b_{\mu\nu}\right] = \left\langle \exp\left[i\int d^4x \sqrt{-g}\left(\frac{1}{2}T^{\mu\nu}g_{\mu\nu} + J^{\mu\nu}b_{\mu\nu}\right)\right]\right\rangle.$$
(5.38)

The simplest holographic dual of such a state is one that contains a five-dimensional bulk with a dynamical graviton (metric tensor G_{ab}) described by the Einstein-Hilbert action, a negative cosmological constant and a two-form bulk gauge field B_{ab} :⁷

$$S = \frac{1}{2\kappa_5^2} \int d^5x \sqrt{-G} \left(R + \frac{12}{L^2} - \frac{1}{3e_H^2} H_{abc} H^{abc} \right).$$
(5.39)

In standard (Dirichlet) quantisation, the two fields asymptote to $g_{\mu\nu}$ and $b_{\mu\nu}$ at the boundary and source $T^{\mu\nu}$ and $J^{\mu\nu}$. Furthermore, H is the three-form defined as H = dB. In component notation, $B = \frac{1}{2}B_{ab} dx^a \wedge dx^b$ and $H = \frac{1}{6}H_{abc} dx^a \wedge dx^b \wedge dx^c$. The two-form gauge field action is the bulk Maxwell Lagrangian $F \wedge \star F$ written in terms of the five-dimensional Hodge dual threeform $H = \star F$, giving the Lagrangian term $H \wedge \star H$. In most of our work, we will set $e_H = L = 1$. Because the two bulk theories are related by dualisation, the background solution to the equations of motion derived from (5.39) give rise to the same magnetised black brane solution known from the Einstein-Maxwell

⁷Throughout this paper, we use Greek and Latin letters to denote the boundary and bulk theory indices, respectively.

Chapter 5. Magnetohydrodynamic waves in a strongly interacting holographic plasma

theory [291].

In the absence of the two-form term, the action (5.39) arises from a consistent truncation of type IIB string theory on S^5 and is upon identification of the Newton's constant $\kappa_5 = 2\pi/N_c$ dual to pure $\mathcal{N} = 4$ SYM at infinite N_c and infinite 't Hooft coupling λ . For reasons discussed above, the full dual of the action (5.39) is unknown and we are not aware of the mechanism for deriving this action from a consistent truncation of ten-dimensional type IIB supergravity. Nevertheless, for purposes of comparing the sizes of matter and electromagnetic contributions to total operator expectation values, it will prove useful to keep the definition of κ_5 in terms of the number of colours N_c of the hypothetical dual deformed $\mathcal{N} = 4$ SYM coupled to dynamical electromagnetism.

To show further evidence that the action (5.39) is a sensible dual of a strongly coupled MHD plasma, it is useful to elucidate the connection between Eq. (5.39) and the Einstein-Maxwell theory. To put an uncharged holographic theory in an external magnetic field, one normally adds the Maxwell action ${\cal F}^2$ with F = dA to the Einstein-Hilbert bulk action. If one imposes the Dirichlet boundary conditions on the bulk one-form A_a , then A_a sources the R-current J_R^{μ} at the boundary, $\int d^4x J^{\mu}_B \delta A_{\mu}$, and thus the electromagnetic field A_{μ} is external and non-dynamical. The investigation of the physics of such a setup was initiated in [291] and studied in numerous subsequent works, including recent [280, 281, 284, 292, 293]. Instead, one can work in alternative quantisation and impose Neumann boundary conditions on A_a . Such a choice exchanges the interpretation of the normalisable and the non-normalisable mode in A_a . From the dual field theory point of view, this can be interpreted as the Legendre transform of the boundary coupling, leading to the variation $\int d^4x A_{\mu} \delta J^{\mu}_{R}$. Physically, this means that in alternative quantisation, an external current sources a dynamical (boundary) vector field (see e.g. [294, 295]). The two boundary theories, one with Dirichlet and one with Neumann boundary conditions, are related by a double-trace deformed RG flow.⁸ Since J_B^{μ} is conserved, one can express it through an anti-symmetric $b_{\mu\nu}$ as $\epsilon^{\mu\nu\rho\sigma}\partial_{\nu}b_{\rho\sigma}$, which, upon integration by parts, yields a dualised $\int d^4x J^{\mu\nu} \delta b_{\mu\nu}$, where $J^{\mu\nu}$ is the anti-symmetric

⁸See Refs. [116–118, 296] and references therein.

current from Eq. (5.11).

From the point of view of the bulk, as in a lower-dimensional theory [297], the Einstein-Maxwell bulk (quantum) path integral runs over the metric and the Maxwell field A_a . Alternatively, one can write the path integral over the fields strength F_{ab} , but at the expense of ensuring the Bianchi identity dF = 0 by introducing a Lagrange multiplier B_{ab} :

$$Z \supset \int \mathcal{D}F_{ab} \mathcal{D}B_{ab} \exp\left\{i \frac{N_c^2}{8\pi^2} \int d^5x \sqrt{-G} \left(F_{ab}F^{ab} + e_H^{-1}B_{ab}\epsilon^{abcde} \nabla_c F_{de}\right)\right\}.$$
(5.40)

Since the second (Bianchi identity) term vanishes for any classical field solution, it has no influence on the saddle point of the path integral. However, it does generate a non-zero contribution to the boundary action, which is precisely the source term $\int d^4x J^{\mu\nu} b_{\mu\nu}$ once we identify $B_{\mu\nu} \sim b_{\mu\nu}$ and $J^{\mu\nu} \sim \epsilon^{\mu\nu\rho\sigma} F_{\rho\sigma}$. The precise dictionary between the bulk and boundary quantities will be discussed in Section 5.3.2. By varying the action with respect to F_{ab} , one obtains the equation of motion

$$F^{ab} = e_H^{-1} \epsilon^{abcde} \nabla_c B_{de} \,. \tag{5.41}$$

Then, the field strength F_{ab} can be integrated out in the saddle point approximation which gives the two-form gauge field Lagrangian term from Eq. (5.39). Furthermore, in the language of the Einstein-Maxwell theory, by using Eq. (5.41), one finds the relation between the one-form R-current J_R^{μ} and B_{ab} field:

$$\langle J_R^{\mu} \rangle = -\frac{N_c^2}{2\pi^2} \lim_{u \to 0} F^{u\mu} = -\frac{N_c^2}{2\pi^2 e_H} \lim_{u \to 0} \epsilon^{\mu\nu\rho\sigma} \partial_{\nu} B_{\rho\sigma} , \qquad (5.42)$$

where u is the radial coordinate and $u \rightarrow 0$ the boundary of the bulk spacetime. Thus, imposing the Dirichlet boundary condition on B_{ab} corresponds to treating J_R^{μ} as a source, which is the same as performing alternative quantisation discussed above. This consistent with our interpretation that the dual field theory of (5.39) contains a dynamical photons. Furthermore, as we will see from a detailed holographic renormalisation in Section 5.3.2, the boundary counter-terms, which are required to keep the on-shell action finite will give us

precisely the Maxwell theory for A_{μ} (dual of $b_{\mu\nu}$) on the boundary, including a renormalised electromagnetic coupling e_r , as in QED.⁹ All further details of this holographic setup, in particular, the renormalisation of our strongly coupled theory, will be presented in Section 5.3.

5.3 Holographic analysis: equation of state and transport coefficients

In this section, we study the relevant details of the simplest holographic theory with Einstein gravity coupled to a two-form bulk field, cf. (5.39), which can source a two-form current associated with the U(1) generalised global symmetry in the boundary theory. As our goal is to study the phenomenology of MHD waves in a strongly coupled plasma using the dispersion relations of [47], we will use holography only to provide us with the necessary microscopic data: the equation of state and the transport coefficients.

In Section 5.3.1, we will begin by discussing details of the magnetic brane solution [291, 299] supported by the bulk action introduced in Section 5.2.2. In Section 5.3.2, we will consider holographic renormalisation of the theory in question and show how the bulk gives rise to a dual theory coupled to dynamical electromagnetism (as in Section 5.2). In particular, we will derive the expectation values of the stress-energy tensor $\langle T^{\mu\nu} \rangle$ and the two-form $\langle J^{\mu\nu} \rangle$ and show that they satisfy the Ward identities (5.8) and (5.9). We will also match and reproduce all of the expected renormalisation group properties, such as the beta function of the electromagnetic coupling, from the point of view of the bulk calculation. In Section 5.3.3, we will then compute and analyse thermal and magnetic properties of the equation of state of the dual plasma. Finally, in Section 5.3.4, we will derive the membrane paradigm formulae for the seven

⁹We note that the way the Maxwell Lagrangian arises on the boundary is equivalent to the way holographic matter can be coupled to dynamical gravity on a cut-off brane [298]. There too, a holographic counter-term gives rise to the Einstein-Hilbert action at the cut-off brane (the boundary) of a more intricately foliated bulk. As shown by Gubser in [298], such a theory can result in a radiation (CFT)-dominated FRW universe at the boundary with the stress-energy tensor of the $\mathcal{N} = 4$ SYM driving the expansion.

transport coefficients required to describe first-order dissipative MHD [47] and compute them.¹⁰ Further details regarding the horizon formulae for the transport coefficients can be found in Appendix 5.7.

5.3.1 Holographic action and the magnetic brane

A holographic action dual to a plasma state with a low-energy limit that can be described by MHD was stated in Eq. (5.39). Including the boundary Gibbons-Hawking term and the holographic counter-term, the full action is

$$S = \frac{N_c^2}{8\pi^2} \left[\int d^5 X \sqrt{-G} \left(R + 12 - \frac{1}{3e_H^2} H_{abc} H^{abc} \right) + \int_{\partial M} d^4 x \sqrt{-\gamma} \left(2 \operatorname{Tr} K - 6 + \frac{1}{e_H^2} \mathcal{H}_{\mu\nu} \mathcal{H}^{\mu\nu} \ln \mathcal{C} \right) \right],$$
(5.43)

where TrK is the trace of the extrinsic curvature of the boundary (∂M) defined by an outward normal vector n^a . For convenience, we now set $e_H = 1$. The two-form $\mathcal{H}_{\mu\nu}$ is defined as a projection of the three-form field strength in the direction normal the boundary, $\mathcal{H}_{\mu\nu} = n^a H_{a\mu\nu}$. C is a dimensionless number that needs to be adjusted to fix the renormalisation condition, of which the details will be discussed below. The equations of motion that follow are

$$R_{ab} + 4G_{ab} - \left(H_{acd}H_b^{\ cd} - \frac{2}{9}G_{ab}H_{cde}H^{cde}\right) = 0, \qquad (5.44)$$

$$\frac{1}{\sqrt{-G}}\partial_a\left(\sqrt{-G}H^{abc}\right) = 0.$$
 (5.45)

Since the theory (5.43) is S-dual to the Einstein-Maxwell theory, we can express the magnetised black brane solution of [291] by dualising the Maxwell

¹⁰These horizon formulae are analogous to the expression for shear viscosity in $\mathcal{N} = 4$ theory [126]. For more recent discussions of other transport coefficients that can be computed directly from horizon data, see e.g. [44, 129, 217, 248, 255, 300, 301].

terms and writting

$$ds^{2} = G_{ab}dx^{a}dx^{b}$$

$$= r_{h}^{2} \left(-F(u)dt^{2} + \frac{e^{2\mathcal{V}(u)}}{v}(dx^{2} + dy^{2}) + \frac{e^{2\mathcal{W}(u)}}{w}dz^{2}\right) + \frac{du^{2}}{4u^{3}F(u)},$$

$$H = \frac{Br_{h}^{2}e^{-2\mathcal{V}+\mathcal{W}}}{2u^{3/2}\sqrt{w}}dt \wedge dz \wedge du.$$
(5.46)

The equations of motion (5.44) for this ansatz reduce to three second-order ordinary differential equations (ODE's) for $\{F, \mathcal{V}, \mathcal{W}\}$ and one additional firstorder constraint. The equation of motion derived from the variation of the twoform (5.45) is automatically satisfied. The equations are equivalent to those derived from the Einstein-Maxwell theory [291] upon identification of the Maxwell bulk two-form F with $F = \mathcal{B} \, dx \wedge dy$, where $\mathcal{B} = Br_h^2/v$.¹¹ The undetermined functions F, \mathcal{V} and \mathcal{W} are can be obtained numerically by using the shooting method. We first expand the background fields near the horizon as

$$F = f_1^h (1-u) + f_2^h (1-u)^2 + \mathcal{O}(1-u)^3,$$

$$\mathcal{V} = v_0^h + v_1^h (1-u) + \mathcal{O}(1-u)^2,$$

$$\mathcal{W} = w_0^h + w_1^h (1-u) + \mathcal{O}(1-u)^2,$$

(5.47)

where the constants $\{f_i^h, v_i^h, w_i^h\}$ can be written in terms of $\{f_1^h, v_0^h, w_0^h\}$ after solving the equations of motion order-by-order near the horizon. The scaling symmetry of our background ansatz then allows us to rescale dx and dy so that $v_0^h = w_0^h = 0$. Next, we match the numerical solutions generated by shooting from the horizon towards the boundary, where the analytical near-boundary

¹¹The metric ansatz is chosen to have the form used in [292]. It can be obtained from the ansatz $ds^2 = -U(r)dt^2 + e^{2V(r)}(dx^2 + dy^2) + e^{2W(r)}dz^2 + dr^2/U(r)$ used in [291] by performing a coordinate transformation $r = r_h/\sqrt{u}$ and shifting V and W by constant $-\ln v$ and $-\ln w$, respectively, which are chosen so that the near-boundary expansion has the form $ds^2 = (1/u) \eta_{\mu\nu} dx^{\mu} dx^{\nu} + du^2/(4u^2)$.

expansions of the metric functions are

$$u F = 1 + f_1^b \sqrt{u} + \frac{f_1^b}{4} u + f_4^b u^2 + \left(\frac{\mathcal{B}^2}{3} + \right) u^2 \ln u + \mathcal{O}(u^{3/2}),$$

$$u e^{2\mathcal{V}} = v + v f_1^b \sqrt{u} + \frac{v(f_1^b)^2}{4} u + v_4^b u^2 - \left(\frac{\mathcal{B}^2}{6}\right) u^2 \ln u + \mathcal{O}(u^{3/2}),$$

$$u e^{2\mathcal{W}} = w + w f_1^b \sqrt{u} + \frac{w(f_1^b)^2}{4} u - \frac{2w v_4^b}{v} u^2 - \left(\frac{w \mathcal{B}^2}{3}\right) u^2 \ln u + \mathcal{O}(u^{3/2}).$$

(5.48)

As before, one can solve for the coefficients $\{f_i^b, v_i^b, w_i^b\}$ in terms of $\{f_1^b, f_4^b, v_4^b\}$. Furthermore, f_1^b can be removed by residual diffeomorphism freedom of the metric ansatz [292]. For a given value of $B = v\mathcal{B}/r_h^2$, we can therefore generate a numerical background by shooting from the initial conditions of the functions set by the near-horizon expansion with $\{f_1^h, v_0^h, w_0^h\} = \{\hat{f}, 0, 0\}$. The numerical value of \hat{f} is chosen so that the near-boundary expansion has $f_1^b = 0$. The near-boundary behaviour of this function then determines the properties of the dual field theory. Note that the theory is governed by a one-parameter family of such numerical solutions characterised by the dimensionless ratio $T/\sqrt{\mathcal{B}}$. In practice, this ratio can be tuned by changing the parameter B of the background ansatz. The numerical solver encounters stiffness problems when $B \approx \sqrt{3}$, i.e. where the temperature is close to zero. All of our numerical results will therefore stop near $T/\sqrt{\mathcal{B}} = 0$. In this work, we do not attempt an independent analysis of the theory at T = 0.

5.3.2 Holographic renormalisation and the bulk/boundary dictionary

The next step in analysing the dual of (5.43) is a systematic holographic renormalisation. In this section, we derive the one-point functions of the stressenergy tensor $\langle T_{\mu\nu} \rangle$ and the two-form current $\langle J_{\mu\nu} \rangle$, and show that they satisfy the Ward identities of magnetohydrodynamics (5.8) and (5.9) [47], which

in terms of operator expectation values take the form

$$\nabla_{\nu} \langle T^{\mu\nu} \rangle = \tilde{H}^{\mu}_{\ \lambda\sigma} \langle J^{\lambda\sigma} \rangle , \qquad \qquad \nabla_{\mu} \langle J^{\mu\nu} \rangle = 0 , \qquad (5.49)$$

where $\tilde{H} = db$ is the field strength of the background gauge field b in field theory. The precise definition of these quantities will become clear below. Since we are only interested in the expansion of MHD to first order in the gradient expansion around a flat (boundary) background, it will be sufficient to only work with terms that contain no more than two derivatives along the boundary directions. The procedure for obtaining holographic renormalisation will closely follow Refs. [102, 302].¹²

We begin by writing the bulk metric in the Fefferman-Graham coordinates [102]

$$ds_{\rm FG}^2 = G_{ab} \, dx^a dx^b = \frac{d\rho^2}{4\rho^2} + \gamma_{\mu\nu}(\rho, x) dx^{\mu} dx^{\nu} = \frac{d\rho^2}{4\rho^2} + \frac{1}{\rho} g_{\mu\nu}(\rho, x) dx^{\mu} dx^{\nu} \,,$$
(5.50)

so that near the boundary, $\rho \approx 0,$ the metric $g_{\mu\nu}$ can be expanded as

$$g_{\mu\nu}(\rho, x) = g^{(0)}_{\mu\nu}(x) + \rho g^{(1)}_{\mu\nu}(x) + \rho^2 \left(g^{(2)}_{\mu\nu}(x) + \tilde{h}_{\mu\nu}(x) \ln \rho \right) + \mathcal{O}(\rho^3) \,.$$
(5.51)

Note that Greek (boundary) indices in a tensor $A^{\mu\nu}$ are raised with the metric $g_{(0)}^{\mu\nu}$, which satisfies $g_{\mu\nu}^{(0)}g_{(0)}^{\mu\nu} = 4$. There are two types of covariant derivative that we will use: $\nabla_{\mu}^{(g)}$ and ∇_{μ} . Firstly, $\nabla_{\mu}^{(g)}$ and $\nabla_{\mu}^{\mu} \equiv g^{\mu\nu}\nabla_{\mu}^{(g)}$ are defined with respect to the metric $g_{\mu\nu}(\rho, x)$, while ∇_{μ} and $\nabla^{\mu} \equiv g_{(0)}^{\mu\nu}\nabla_{\mu}$ are defined through the metric $g_{\mu\nu}^{(0)}(x)$. The Ricci tensors of $g_{\mu\nu}$ and $g_{\mu\nu}^{(0)}$ are denoted by $R_{\mu\nu}^{(g)}$ and $R_{\mu\nu}^{(0)}$, respectively.

The components of bulk two-form gauge field B_{ab} in the boundary field theory directions can similarly be expanded near the boundary as

$$B_{\mu\nu}(\rho, x) = B^{(0)}_{\mu\nu}(x) + B^{(1)}_{\mu\nu}(x) \ln \rho + \mathcal{O}(\rho) .$$
(5.52)

In the boundary directions, the three-form field strength is defined as $H_{\mu\nu\sigma} = \partial_{\mu}B_{\nu\sigma} + \partial_{\nu}B_{\sigma\mu} + \partial_{\sigma}B_{\mu\nu}$, with the near-boundary expansion $H_{\mu\nu\sigma}(\rho, x) =$

¹²This part of the calculation was performed by using the Mathematica package xAct [303].

 $H^{(0)}_{\mu\nu\sigma}(x) + H^{(1)}_{\mu\nu\sigma}(x) \ln \rho + \mathcal{O}(\rho)$. Each $H^{(n)}$ is defined in terms of $B^{(n)}$, i.e. in the same way at each order. $B^{(0)}$ can now be related to the two-form gauge field source of the boundary theory, $\int d^4x \sqrt{-g} \, J^{\mu\nu} \delta b_{\mu\nu}$. In the bulk, the variation of the on-shell contribution from the H^2 term is

$$\delta S_{on-shell} = \frac{N_c^2}{2\pi^2} \int d^4x \sqrt{-g} \,\mathcal{H}^{\mu\nu} \delta B^{(0)}_{\mu\nu} + \dots$$
 (5.53)

The expectation value of the operator sourced by $B^{(0)}_{\mu\nu}$ thus depends on a factor of N_c^2 . However, since, by definition, $J^{\mu\nu} = \frac{1}{2} \epsilon^{\mu\nu\lambda\sigma} F_{\lambda\sigma}$, of which the expectation value contains no colour trace, we need to identify the bulk $B^{(0)}_{\mu\nu}$ and the field theory source $b_{\mu\nu}$ as

$$B^{(0)}_{\mu\nu} = \frac{2\pi^2}{N_c^2} b_{\mu\nu} \,. \tag{5.54}$$

The expectation value of $J^{\mu\nu}$ can then be obtained by taking a variational derivative of the on-shell action with respect to the source $b_{\mu\nu}$.

The Ward identities (5.49) can be obtained by solving the equations of motion (5.44) and (5.45) [302]. In Fefferman-Graham coordinates (5.50), these equations (together with the trace of (5.44)) become

$$\begin{split} 0 &= \frac{1}{2} \mathrm{Tr} \left[g^{-1} g'' \right] - \frac{1}{4} \mathrm{Tr} \left[g^{-1} g' g^{-1} g' \right] + \frac{1}{3} \rho^{2} \mathrm{Tr} \left[g^{-1} B' g^{-1} B' \right] - \frac{1}{18} \rho \operatorname{Tr} [g^{-1} H^{2}], \\ 0 &= \frac{1}{2} \left(\nabla_{\mu}^{(g)} \mathrm{Tr} g' - \nabla_{(g)}^{\nu} g'_{\mu\nu} \right) - 2 \rho^{2} H_{\mu\alpha\beta} \left(g^{-1} B' g^{-1} \right)^{\alpha\beta}, \\ 0 &= \rho \left(2 g''_{\mu\nu} - 2 (g' g^{-1} g')_{\mu\nu} + g'_{\mu\nu} \mathrm{Tr} [g^{-1} g'] \right) + R^{(g)}_{\mu\nu} - 2 g'_{\mu\nu} - g_{\mu\nu} \mathrm{Tr} [g^{-1} g'] \\ &+ 8 \rho^{3} \left((B' g^{-1} B')_{\mu\nu} - \frac{1}{3} g_{\mu\nu} \mathrm{Tr} \left[g^{-1} B' g^{-1} B' \right] \right) \\ &+ \rho^{2} \left(H^{2}_{\mu\nu} - \frac{2}{9} g_{\mu\nu} \mathrm{Tr} [g^{-1} H^{2}] \right), \\ 0 &= \frac{d}{d\rho} \left(2 \rho \left(g^{-1} B' g^{-1} \right)^{\mu\nu} \right) + \frac{1}{2} \nabla_{(g)}^{\lambda} \left(g^{\mu\alpha} g^{\nu\beta} H_{\lambda\alpha\beta} \right), \\ 0 &= \nabla_{\nu} \left(g^{-1} B' g^{-1} \right)^{\mu\nu}, \end{split}$$

where g^{-1} denotes the matrix inverse of g (in components, this is $g^{\mu\nu})$ and

where

$$\operatorname{Tr}[g^{-1}B'g^{-1}B'] = -B'_{\mu_1\mu_2}B'_{\nu_1\nu_2}g^{\mu_1\nu_1}g^{\mu_2\nu_2}, H^2_{\mu\nu} = H_{\mu\lambda_1\lambda_2}H_{\nu\sigma_1\sigma_2}g^{\lambda_1\sigma_1}g^{\lambda_2\sigma_2}.$$
(5.55)

Expanding equations (5.55) around small ρ , we find that

$$g_{\mu\nu}^{(1)} = \frac{1}{2} \left(R_{\mu\nu}^{(0)} - \frac{1}{6} g_{\mu\nu}^{(0)} R^{(0)} \right), \qquad (g^{(1)})^{\mu}_{\ \mu} = \frac{1}{6} R.$$
 (5.56)

Since $g^{(1)}_{\mu\nu}$ is proportional to second derivatives of the boundary metric, and we are only keeping track of terms up to second orders in boundary derivaties, we can ignore terms with $g^{(1)}_{\mu\nu}$. The remaining equations of motion can thus be written as

$$(g^{(2)})^{\mu}_{\ \mu} + \frac{1}{3} B^{(1)}_{\mu\nu} B^{(1)\mu\nu} = 0, \qquad \tilde{h}^{\mu}_{\ \mu} = 0, \qquad \nabla_{\nu} B^{(1)\mu\nu} = 0, \qquad (5.57)$$

$$-2H^{(0)}_{\mu\nu\lambda}B^{(1)\nu\lambda} + \nabla^{\nu}_{(0)} \left(g^{(0)}_{\mu\nu}(g^{(2)})^{\lambda}_{\lambda} - g^{(2)}_{\mu\nu} - \frac{1}{2}\tilde{h}_{\mu\nu}\right) = 0, \quad (5.58)$$

$$\tilde{h}_{\mu\nu} + \frac{1}{2} \left(4B^{(1)}_{\mu\lambda} (B^{(1)})^{\lambda}_{\nu} - g^{(0)}_{\mu\nu} B^{(1)}_{\lambda\sigma} B^{(1)\lambda\sigma} \right) = 0.$$
 (5.59)

The expectation values of the stress-energy tensor and the two-form current follow from the generating functional (5.38):

$$\langle T^{\mu\nu}\rangle = -\frac{2i}{\sqrt{-g^{(0)}}}\frac{\delta \ln W}{\delta g^{(0)}_{\mu\nu}}, \qquad \langle J^{\mu\nu}\rangle = -\frac{i}{\sqrt{-g^{(0)}}}\frac{\delta \ln W}{\delta b_{\mu\nu}}.$$
 (5.60)

In holography, W is computed from the (on-shell) action (5.43), giving us¹³

$$\langle T_{\mu\nu} \rangle = -\frac{N_c^2}{4\pi^2} \lim_{\epsilon \to 0} \frac{r_h^2}{\epsilon} \left(K_{\mu\nu} - \gamma_{\mu\nu} \text{Tr}K - 3\gamma_{\mu\nu} + \frac{1}{2}R[\gamma]_{\mu\nu} -\frac{1}{4}\gamma_{\mu\nu}R[\gamma] - \left(\mathcal{H}_{\mu\lambda}\mathcal{H}_{\nu}^{\ \lambda} - \frac{1}{4}\gamma_{\mu\nu}\mathcal{H}_{\alpha\beta}\mathcal{H}^{\alpha\beta} \right) \ln(\mathcal{C}\rho) \right) \Big|_{\rho=\epsilon}, \quad (5.61)$$

$$\langle J_{\mu\nu} \rangle = -\lim_{\epsilon \to 0} \left. \mathcal{H}_{\mu\nu} \right|_{\rho=\epsilon}.$$
 (5.62)

Note that by taking $\kappa_5 \sim N_c$, while the expectation value of $T^{\mu\nu}$ scales as N_c^2 , the expectation value of $J^{\mu\nu}$ is of order $\mathcal{O}(1)$.

¹³Note that in order to raise indices of the boundary theory expectation values, one needs to use the induced metric $\gamma_{\mu\nu}$.

By using Eq. (5.57) and the fact that $\mathcal{H}_{\mu\nu} = n^{\rho}H_{\rho\mu\nu} = -2B^{(1)}_{\mu\nu} + \mathcal{O}(\rho)$, we find that the boundary two-form current is conserved:

$$\nabla^{\mu}_{(0)} \langle J_{\mu\nu} \rangle = -2 \nabla^{\mu} B^{(1)}_{\mu\nu} = 0.$$
 (5.63)

Using the definition (5.11), which gives $\langle J_{\mu\nu} \rangle = \frac{1}{2} \epsilon_{\mu\nu\rho\sigma} \langle F^{\rho\sigma} \rangle$ and connects Eq. (5.63) with the Bianchi identity, we find that $\star B^{(1)}$ sets the expectation value of the Maxwell field strength $\langle F \rangle$. Furthermore, the (regularised) stress-energy tensor (5.61) becomes

$$\langle T_{\mu\nu} \rangle = \lim_{\epsilon \to 1/\Lambda^2} \frac{N_c^2}{2\pi^2} \left(-g_{\mu\nu}^{(2)} + g_{\mu\nu}^{(0)} (g^{(2)})^{\lambda}_{\ \lambda} - \frac{1}{2} \tilde{h}_{\mu\nu} - \frac{1}{2} \tilde{h}_{\mu\nu} \ln \left(\mathcal{C}_M \epsilon \right) \right)$$
(5.64)

where Λ is the UV cutoff of the theory. As discussed in Section 5.2, the choice of the dimensionful constant C_M (including C and e_H from Eq. (5.61)) must now be made in order to fix the renormalisation condition, which will render the renormalised expectation value $\langle T_{\mu\nu} \rangle$ physical.

To see how the constant C_M in Eq. (5.64) is related to our discussion in Section 5.2, we write the last term by introducing a mass scale M:

$$\frac{N_c^2}{2\pi^2} \tilde{h}_{\mu\nu} \ln(\Lambda/\mathcal{C}) = \frac{N_c^2}{2\pi^2} \tilde{h}_{\mu\nu} \ln(\Lambda/M) + \tilde{h}_{\mu\nu} \left(\frac{1}{e_r^2} - \frac{N_c^2}{2\pi^2} \ln(\Lambda/M)\right).$$
(5.65)

What can be seen from Eq. (5.65) is that this splitting precisely reproduces the way the logarithmic divergence enter into the stress-energy tensor from two different pieces of the Lagrangian: the matter content (with its coupling to the photons) and the electromagnetic (Maxwell) part:

$$\langle T_{\mu\nu} \rangle = \langle T^{matter}_{\mu\nu} \rangle + \langle T^{EM}_{\mu\nu} \rangle, \qquad (5.66)$$

with the two terms being

$$\langle T^{matter}_{\mu\nu} \rangle = \frac{N_c^2}{2\pi^2} \left(-g^{(2)}_{\mu\nu} + g^{(0)}_{\mu\nu} (g^{(2)})^{\lambda}_{\lambda} - \frac{1}{2} \tilde{h}_{\mu\nu} \right) + \frac{N_c^2}{2\pi^2} \tilde{h}_{\mu\nu} \ln(\Lambda/M) ,$$
(5.67)

(5.68)

and

$$\langle T_{\mu\nu}^{EM} \rangle = \frac{1}{e_r^2} \tilde{h}_{\mu\nu} - \frac{N_c^2}{2\pi^2} \tilde{h}_{\mu\nu} \ln\left(\Lambda/M\right) \,.$$
 (5.69)

Finally, we note that the electromagnetic $\langle T^{EM}_{\mu\nu}\rangle$ would follow precisely from the Maxwell boundary action

$$S_{EM} = -\frac{1}{4e(\Lambda/M)^2} \int d^4x \sqrt{-g} F_{\mu\nu} F^{\mu\nu} , \qquad (5.70)$$

upon using Eq. (5.59) and the fact that the bulk $\star B^{(1)}$ determines $\langle F_{\mu\nu} \rangle$:

$$\langle T_{\mu\nu}^{EM} \rangle = \frac{1}{e(\Lambda/M)^2} \left(\langle F_{\mu\alpha} F_{\nu}^{\ \alpha} \rangle - \frac{1}{4} \eta_{\mu\nu} \langle F_{\alpha\beta} F^{\alpha\beta} \rangle \right)$$

= $\frac{1}{e(\Lambda/M)^2} \left(\langle F_{\mu\alpha} \rangle \langle F_{\nu}^{\ \alpha} \rangle - \frac{1}{4} \eta_{\mu\nu} \langle F_{\alpha\beta} \rangle \langle F^{\alpha\beta} \rangle \right),$ (5.71)

where the last equality follows from the fact that quantum fluctuations of the photon field are suppressed in the boundary QFT. Our holographic calculation thus fully reproduces Eq. (5.34), which followed from the field theory discussion in Section 5.2.2. Furthermore, the running electromagnetic coupling constant matches the one found from field theory (cf. Eq. (5.37)) [284]. Hence, our holographic setup contains the a U(1)-gauged matter content of the $\mathcal{N} = 4$ SYM theory. In terms of bulk quantities, the renormalised stress-energy tensor and the two-form current are

$$\langle T_{\mu\nu} \rangle = \frac{N_c^2}{2\pi^2} \left(-g_{\mu\nu}^{(2)} + g_{\mu\nu}^{(0)} (g^{(2)})^{\lambda}_{\lambda} - \frac{1}{2} \tilde{h}_{\mu\nu} \right) + \frac{1}{e_r^2} \tilde{h}_{\mu\nu} , \qquad (5.72)$$

$$\langle J_{\mu\nu} \rangle = 2B^{(1)}_{\mu\nu}, \qquad (5.73)$$

where, as in Section 5.2, e_r is the renormalised coupling which needs to be set by experimental input—the renormalisation condition. In practice, this constant is fixed by choosing the value of C in (5.61). For the same reasons as in QFT, there is therefore an inherent ambiguity in holographic results, which has to be fixed by external physically-motivated input.

We conclude this section by noting that since $\nabla^{\mu} \langle T^{EM}_{\mu\nu} \rangle = 0$ follows from Maxwell's equations, using the relation (5.58) implies that the Ward identity

for the stress-energy tensor satisfies Eq. (5.8), or in terms of our holographic notation, $\nabla_{\nu} \langle T^{\mu\nu} \rangle = \tilde{H}^{\mu}_{\lambda\sigma} \langle J^{\lambda\sigma} \rangle$ in Eq. (5.49).

5.3.3 Equation of state

To find the equation of state of our theory, we can use the renormalised stressenergy tensor (5.72) and the two-form current (5.73) computed in the previous section. Expressed in terms of the near-boundary expansions (5.48), which can be read off from the numerical background, we find

$$\left\langle T^{tt} \right\rangle = \frac{N_c^2}{2\pi^2} \left[-\frac{3}{4} f_4^b r_h^4 + \frac{\mathcal{B}^2}{8\pi\bar{\alpha}} \right],\tag{5.74}$$

$$\langle T^{xx} \rangle = \langle T^{yy} \rangle = \frac{N_c^2}{2\pi^2} \left[\left(-\frac{1}{4} f_4^b + \frac{v_4^b}{v} \right) r_h^4 - \frac{\mathcal{B}^2}{4} + \frac{\mathcal{B}^2}{8\pi\bar{\alpha}} \right], \tag{5.75}$$

$$\langle T^{zz} \rangle = \frac{N_c^2}{2\pi^2} \left[\left(-\frac{1}{4} f_4^b - 2\frac{v_4^b}{v} \right) r_h^4 - \frac{\mathcal{B}^2}{8\pi\bar{\alpha}} \right], \tag{5.76}$$

where we have used the (renormalised) fine-structure constant $\alpha = e_r^2/4\pi$ of the electromagnetic coupling in the plasma, which, as e_r , is fixed by the choice of C in (5.61), and rescaled it by $N_c^2/2\pi^2$ (or $|\beta(1/e^2)|$) as

$$\bar{\alpha} = \frac{N_c^2}{2\pi^2} \alpha \,. \tag{5.77}$$

The coupling $\bar{\alpha}$ has to be fixed by experimental observations as in any other quantum field theory, which is not easy in an unrealistic toy model.

To studying strongly coupled MHD, it is phenomenologically relevant to not only consider the matter and light-matter interactions, but also include large electromagnetic self-interactions encoded in the Maxwell action. However, since we are working with a holographic large- N_c matter sector and a single photon, it is rather unnatural to expect a Maxwell term of the same order, even though N_c controls the running of the electromagnetic coupling. The choice that we make here is to set the rescaled constant $\bar{\alpha}$ to the physically motivated $\bar{\alpha} = 1/137$. There are several ways to think about this choice: one is imagining that our plasma contains magnetic properties, which have non-

trivial scalings with N_c , while another interpretation may assume that the bulk studied here could remain a valid dual of a theory with a reasonably small N_c . Of course, by considering only a classical bulk theory, we are restricting the strict validity of any computed observable to the limit of $N_c \rightarrow \infty$. As soon as one moves towards finite N_c , it becomes crucial to estimate the size of subleading $1/N_c^2$ corrections (topological expansion in the string coupling g_s)—an endeavour in holography (and string theory) which to date has been largely neglected and will continue being neglected in this work.¹⁴ A less problematic limit is that of the infinite 't Hooft coupling, which is also implied by the choice of our action.¹⁵ Perhaps the best interpretation is one of an "agnostic choice" led by our having to fix a free parameter to some value. We will return to a more careful investigation of the dependence of our results on this choice in Section 5.4.3.

The expectation values of the stress-energy tensor expressed in (5.74)-(5.76) are related to the MHD stress-energy tensor in Eq. (5.12) by

$$\langle T^{tt} \rangle = \varepsilon, \qquad \langle T^{xx} \rangle = p, \qquad \langle T^{zz} \rangle = p - \mu \rho.$$
 (5.78)

We note that, as required in a conformal field theory with a trace anomaly induced by electromagnetic interactions, the trace of stress-energy tensor is non-zero. The holographic two-form current,

$$\langle J^{tz} \rangle = \mathcal{B} = \frac{Br_h^2}{v} \,, \tag{5.79}$$

is related to the equilibrium magnetic flux line density appearing in the MHD equation (5.13) as $\langle J^{tz} \rangle = \rho$. Temperature and entropy can be expressed in terms of the background geometry as

$$T = \frac{1}{2\pi} f_1^h r_h , \qquad s = \frac{N_c^2}{2\pi^2} \left(\frac{\pi r_h^3}{v\sqrt{w}} \right), \qquad (5.80)$$

¹⁴For some discussions of $1/N_c^2$ corrections to the thermodynamic free energy (the equilibrium partition function) and hydrodynamic long-time tails, see [131–133].

¹⁵For recent discussions of coupling-dependent holography, see [232, 304–307] and references therein.

and are therefore independent of the renormalised electromagnetic charge. The chemical potential, which corresponds to the density of magnetic flux lines, can be computed by using the thermodynamic identity $\varepsilon + p = sT + \mu\rho$ (cf. (5.20)):

$$\mu = \frac{\langle T^{xx} \rangle - \langle T^{zz} \rangle}{\langle J^{tz} \rangle} = \frac{N_c^2}{2\pi^2} \left(\frac{3v_4^b}{B} - \frac{B}{v} + \frac{B}{4\pi v\bar{\alpha}} \right) r_h^2.$$
(5.81)

Note that with our choice of the bulk theory scalings, $\rho \sim \mathcal{O}(1)$ and $\mu \sim \mathcal{O}(N_c^2)$. Furthermore, while $T \sim \mathcal{O}(1)$, p, ε and s all scale as $\mathcal{O}(N_c^2)$.



Figure 5.1. Dimensionless energy density ε/B^2 (top-left), pressure p/B^2 (top-right), entropy density $s/B^{3/2}$ (bottom-left) and chemical potential μ/B (bottom-right), in units of $N_c^2/(2\pi^2)$, plotted as a function of the dimensionless parameter T/\sqrt{B} . The first three plots use logarithmic scales on both axes.

Using the above relations, we can perform two consistency checks on our holographic setup and numerical calculations of the background. First, the value of the pressure computed from the stress-energy tensor component $\langle T^{xx} \rangle =$

p can be compared with the value of the Euclidean on-shell action,

$$p = -i(\beta V_3)^{-1} S_{on-shell},$$
 (5.82)

where $\beta = 1/T$ and V_3 is the spatial volume of the theory. Secondly, we can compute $\varepsilon + p - \mu \rho$ from the stress-energy tensor evaluated near the boundary and by using the thermodynamic relation (5.20), check whether its values agrees with sT computed purely from horizon quantities. Both calculations show consistency of our setup in that we find $\langle T^{xx} \rangle = -i(\beta V_3)^{-1}S_{on-shell}$ and $\langle T^{tt} \rangle + \langle T^{zz} \rangle = sT$, within numerical precision.

We can now plot various thermodynamic quantities in a dimensionless manner by dividing them by appropriate powers of \mathcal{B} . The natural dimensionless parameter with respect to which we present our numerical results is $T/\sqrt{\mathcal{B}}$. The results for the energy density, pressure, entropy density and chemical potential are shown in Figure 5.1. The theory has two distinct regimes: the low-and the high-temperature regimes, or alternatively, the strong and weak magnetic field regimes, respectively. The high-temperature regime $T/\sqrt{\mathcal{B}} \gg 1$ is one to which MHD has been historically applied and to which the formulation of MHD, which assumes a weak-field separation between fluid and charge degrees of freedom can be applied. The claim presented in the Ref. [47] is that within the dual formulation, however, MHD applies for all values of $T/\sqrt{\mathcal{B}}$ provided that the theory remains in the hydrodynamic regime. The profiles of the thermodynamic functions in Figure 5.1 show a smooth crossover between the two regimes, which occurs around

$$T/\sqrt{\mathcal{B}} \approx 0.5 - 0.7. \tag{5.83}$$

By using numerical fits, the equation of state in the two limits behaves as expected on dimensional grounds [47]. We present the numerical results in Table 5.1.

In the limit of $\mathcal{B} \to 0$, the weak-field result approximately limits to the equation of state of a strongly coupled, thermal $\mathcal{N} = 4$ plasma, dual to a five dimensional AdS-Schwarzschild black brane with $p_{\mathcal{N}=4} = \frac{1}{8}N_c^2\pi^2T^4$; i.e.

	weak field $(T/\sqrt{\mathcal{B}} \gg 1)$	strong field $(T/\sqrt{\mathcal{B}} \ll 1)$
ε	$\frac{N_c^2}{2\pi^2} \left(74.1 \times T^4 \right)$	$rac{N_c^2}{2\pi^2} \left(5.62 imes \mathcal{B}^2 ight)$
p	$\frac{N_c^2}{2\pi^2} \left(25.3 \times T^4 \right)$	$\frac{N_c^2}{2\pi^2} \left(5.32 \times \mathcal{B}^2 \right)$
s	$\frac{N_c^2}{2\pi^2} \left(99.4 \times T^3\right)$	$\frac{N_c^2}{2\pi^2} \left(7.41 \times \mathcal{B}T \right)$
μ	$\frac{N_c^2}{2\pi^2} \left(10.9 \times \mathcal{B}\right)$	$\frac{N_c^2}{2\pi^2} \left(2.88 \times \mathcal{B} ight)$

Chapter 5. Magnetohydrodynamic waves in a strongly interacting holographic plasma

Table 5.1. Asymptotic behaviour of the equation of state in weak and strong field limits for $\bar{\alpha} = 1/137$.

 $\lim_{\mathcal{B}\to 0} p_{weak} \approx 1.28 \times N_c^2 T^4$ and $p_{\mathcal{N}=4} \approx 1.23 \times N_c^2 T^4$. We also note that the value of the pressure at low temperature strongly depends on the renormalised (re-scaled) fine structure constant $\bar{\alpha}$, which we set to $\bar{\alpha} = 1/137$.

5.3.4 Transport coefficients

Next, we compute the seven transport coefficients, η_{\perp} , η_{\parallel} , r_{\perp} , r_{\parallel} , ζ_{\perp} , ζ_{\parallel} and ζ_{\times} , by using the Kubo formulae derived in [47, 278] and reviewed in Appendix 5.6.1. The procedure only requires us to turn on time-dependent fluctuations of the background fields without any spatial dependence, $G_{ab} \rightarrow G_{ab} + \delta G_{ab}(t)$ and $B_{ab} \rightarrow B_{ab} + \delta B_{ab}(t)$. The perturbations asymptote to the boundary sources $\delta g^{(0)}_{\mu\nu}$ and $\delta b^{(0)}_{\mu\nu}$ of the dual stress-energy tensor and the two-form current. In the absence of spatial dependence, the fluctuations decouple into five separate channels, from which the seven transport coefficients are computed, with each channel containing one independent dynamical second-order equation. The sets of decoupled fluctuations responsible for their respective transport coefficients are

$$\eta_{\perp} : \quad \delta G_{xy} ,$$

$$\eta_{\parallel} : \quad \delta G_{xz}, \, \delta B_{tx}, \, \delta B_{xu} ,$$

$$\zeta_{\perp}, \, \zeta_{\parallel}, \, \zeta_{\times} : \quad \delta G_{tt}, \, \delta G_{xx}, \, \delta G_{yy}, \, \delta G_{zz}, \, \delta B_{tz}, \, \delta G_{tu}, \, \delta B_{zu}, \, \delta G_{uu} , \quad (5.84)$$

$$r_{\perp} : \quad \delta B_{xz}, \, \delta G_{tx}, \, \delta G_{xu} ,$$

$$r_{\parallel} : \quad \delta B_{xy} ,$$

with only one out of the three bulk viscosities being independent. Each one of the transport coefficients can then be related to a membrane paradigm formula and computed entirely in terms of the horizon quantities. We summarise these relations here and discuss their derivation below:

$$\eta_{\perp} = \frac{N_c^2}{2\pi^2} \left(\frac{r_h^3}{4v\sqrt{w}} \right) = \frac{1}{4\pi} s,$$

$$\eta_{\parallel} = \frac{N_c^2}{2\pi^2} \left(\frac{r_h^3}{4w^{3/2}} \right) = \frac{1}{4\pi} \frac{v}{w} s,$$

$$r_{\perp} = \frac{2\pi^2}{N_c^2} \left(\frac{\sqrt{w}}{2r_h} \right) \left(\frac{\mathfrak{b}_{xz}^{(-)}(1)}{\mathfrak{b}_{xz}^{(-)}(0)} \right)^2,$$

$$r_{\parallel} = \frac{2\pi^2}{N_c^2} \left(\frac{v}{2r_h\sqrt{w}} \right),$$

$$\zeta_{\perp} = \frac{1}{4} \zeta_{\parallel} = -\frac{1}{2} \zeta_{\times} = \frac{N_c^2}{2\pi^2} \left(\frac{r_h^3}{12v\sqrt{w}} \left(\frac{6+B^2}{6-B^2} \right)^2 \left[\frac{\mathfrak{Z}^{(-)}(1)}{\mathfrak{Z}^{(-)}(0)} \right]^2 \right),$$

(5.85)

where $\mathfrak{b}^{(-)}$ and $\mathfrak{Z}^{(-)}$ are the time-independent solutions of the fluctuations δB_{xz} and $Z_s = \delta G_x^x + \delta G_y^y - (2\mathcal{V}'/\mathcal{W}')\delta G_z^z$, respectively. The arguments denote that the functions are evaluated either at the horizon, u = 1, or the boundary, u = 0. Note that the value at the boundary is set by the Dirichlet boundary conditions.

What we see is that the ratio of the transverse shear viscosity (w.r.t to the background magnetic field) to entropy density is universal, resulting in $\eta_{\perp}/s = 1/4\pi$. Furthermore, the expressions for η_{\parallel} and r_{\parallel} only depend on the background quantities v and w, while ζ_{\perp} , ζ_{\parallel} and r_{\perp} also depend on the fluctuations of the fields.¹⁶

In order to derive the horizon formulae, we use a method similar to [301, 308]. Here, we will only explicitly show the derivation of the transverse resistivity r_{\perp} . The other formulae from Eq. (5.85) are derived in Appendix 5.7. First, we combine the equations of motion for the relevant fluctuations δB_{xz} , δG_{tx} ,

¹⁶For a holographic derivation of bulk viscosity in neutral relativistic hydrodynamics, see [300].

and δG_{xu} by eliminating the metric fluctuations into a single second-order differential equation

$$\delta B_{xz}'' + \left(\frac{3}{2u} + \frac{F'}{F} - \mathcal{W}'\right) \delta B_{xz}' + \left(\frac{\omega^2}{4r_h^2 u^3 F^2} - \frac{B^2 e^{-4\mathcal{V}}}{u^3 F}\right) \delta B_{xz} = 0.$$
(5.86)

Since we are only computing first-order transport coefficients, it is sufficient to solve Eq. (5.86) to linear order in ω . To find the solution, we assume that there exists a time-independent solution $\mathfrak{b}_{xz}^{(-)}(u)$, which asymptotes to a constant both at the boundary and the horizon. At the boundary, this asymptotic value is related to the source of the two-form background gauge field, i.e. $\mathfrak{b}_{xz}^{(-)}(u \to 0) = \delta B_{xz}^{(0)}$. The time-dependent information is contained in the second solution, linearly-independent from $\mathfrak{b}_{xz}^{(-)}$. We refer to this solution as $\mathfrak{b}_{xz}^{(+)}$. It can be expressed in terms on integral over the Wronskian W_R of (5.86) as

$$\mathfrak{b}_{xz}^{(+)}(u) = \mathfrak{b}_{xz}^{(-)}(u) \int_{u}^{1} du' \frac{W_{R}(u')}{\left(\mathfrak{b}_{xz}^{(-)}(u')\right)^{2}},$$
(5.87)

where

$$W_R(u) = \exp\left[-\int_u^1 du' \left(\frac{3}{2u'} + \frac{F'(u')}{F(u')} - \mathcal{W}'(u')\right)\right] = \frac{1}{u^{3/2} F e^{-\mathcal{W}}}.$$
(5.88)

The near-boundary and the near-horizon expansions of $\mathfrak{b}_{xz}^{(+)}$ are

$$\mathfrak{b}_{xz}^{(+)} = \begin{cases} \sqrt{w} \left[\mathfrak{b}_{xz}^{(-)}(0) \right]^{-1} \ln u + \mathcal{O}(\sqrt{u}), & \text{for } u \approx 0, \\ -r_h \left[2\pi T \mathfrak{b}_{xz}^{(-)}(1) \right]^{-1} \ln(1-u) + \mathcal{O}(1-u), & \text{for } u \approx 1. \end{cases}$$
(5.89)

Finally, $\delta B_{xz}(\omega, u)$ is then a the following linear combination of the two solutions:

$$\delta B_{xz}(\omega, u) = \mathfrak{b}_{xz}^{(-)}(u) + \alpha(\omega)\mathfrak{b}_{xz}^{(+)}(u) + \mathcal{O}(\omega^2).$$
(5.90)

The coefficient $\alpha(\omega)$ can be determined by imposing regular ingoing boundary conditions at the horizon, which corresponds to computing a retarded dual
correlator [309, 310]:

$$\delta B_{xz}(u) = (1-u)^{-\frac{i\omega}{4\pi T}} \tilde{B}_{xz}, \qquad (5.91)$$

The function $\tilde{B}_{xz}(u)$ is regular at the horizon. This choice of the boundary condition implies that near the horizon, δB_{xz} behaves as

$$\delta B_{xz}(u) = \mathfrak{b}_{xz}^{(-)}(u) + \alpha(\omega)\mathfrak{b}_{xz}^{(+)}(u) + \ldots = \mathfrak{b}_{xz}^{(-)}(1)\left(1 - \frac{i\omega}{4\pi T}\ln(1-u)\right) + \ldots$$
(5.92)

Comparing Eq. (5.92) with the asymptotic behaviour of $\mathfrak{b}_{xz}^{(+)}$ in (5.89), we find $\alpha = (i\omega/2r_h) \left[\mathfrak{b}_{xz}^{(-)}(1)\right]^2$. Thus, the near-boundary expression for δB_{xz} becomes

$$\delta B_{xz}(u) = \mathfrak{b}_{xz}^{(-)}(0) \left(1 + \frac{i\omega}{2r_h} \sqrt{w} \left[\frac{\mathfrak{b}_{xz}^{(-)}(1)}{\mathfrak{b}_{xz}^{(-)}(0)} \right]^2 \ln u \right) + \mathcal{O}(u) \,. \tag{5.93}$$

By substituting this expression into the expectation value of the two-form current $\langle J^{\mu\nu} \rangle$, as found in Eq. (5.62), we obtain

$$\langle \delta J^{xz} \rangle = \lim_{u \to 0} \left(2u^{3/2} \sqrt{F} \delta B'_{xz}(u) \right) = \frac{2\pi^2}{N_c^2} \left(i\omega r_h^{-1} \sqrt{w} \left[\frac{\mathfrak{b}_{xz}^{(-)}(1)}{\mathfrak{b}_{xz}^{(-)}(0)} \right]^2 \right) \delta b_{xz}^{(0)} \,.$$
(5.94)

Finally, using the Kubo formula for r_{\perp} , which is derived and presented in Eq. (5.119) of Appendix 5.6.1, we recover the expression presented in Eq. (5.85). All of the six remaining transport coefficients can be obtained by following the same procedure. We refer the reader to Appendix 5.7 for their detailed derivations.

The plots of the (dimensionless) transport coefficients η_{\parallel} , ζ_{\parallel} , r_{\perp} and r_{\parallel} as a function of $T/\sqrt{\mathcal{B}}$ are presented in Figure 5.2. The remaining three viscosities can easily be inferred from Eq. (5.85). In particular, $\eta_{\perp}/s = 1/(4\pi)$, $\zeta_{\perp} = \zeta_{\parallel}/4$ and $\zeta_{\times} = -\zeta_{\parallel}/2$. We note that all transport coefficients satisfy the positive entropy production bounds discussed in Section 5.1. It is interesting that the bulk viscosity inequality $\zeta_{\perp}\zeta_{\parallel} \ge \zeta_{\times}^2$ is saturated, i.e. $\zeta_{\perp}\zeta_{\parallel} = \zeta_{\times}^2$ in



Chapter 5. Magnetohydrodynamic waves in a strongly interacting holographic plasma

Figure 5.2. The plots of (dimensionless) first-order transport coefficients as a function of T/\sqrt{B} .

the plasma studied here for all parameters of the theory.

We can now investigate the behaviour of the transport coefficients in the two extreme limits of $T/\sqrt{\mathcal{B}} \to 0$ and $T\sqrt{\mathcal{B}} \to \infty$, i.e. the strong- and the weak-field regimes, respectively. The leading-order power-law scaling (which we assume) and the coefficient follow from numerical fits. The results are presented in Table 5.2.

Since the entropy density s vanishes in the limit of zero temperature, all first-order transport coefficients vanish in the strong-field limit of $T \rightarrow 0$. This observation is consistent with predictions of [47], based on symmetry arguments. As a consequence, all (first-order) dissipative effects also vanish in the $T \rightarrow 0$ limit.

In the regime of a weak magnetic field, $T \gg \sqrt{\mathcal{B}}$, we find that both shear viscosities η_{\perp} and η_{\parallel} converge to $\eta_{\perp} = \eta_{\parallel} = s/(4\pi)$ as $\mathcal{B}/T^2 \to 0$. On the other hand, the longitudinal bulk viscosity limits to $\zeta_{\parallel} \to 4\eta/3$, which is

	weak field $(T/\sqrt{\mathcal{B}} \gg 1)$	strong field $(T/\sqrt{\mathcal{B}} \ll 1)$
η_{\perp}	$\frac{s}{4\pi}$	$\frac{s}{4\pi}$
η_{\parallel}	$1.00 \times \frac{s}{4\pi}$	$\frac{s}{4\pi}\left(21.32 \times \frac{T^2}{\mathcal{B}}\right)$
ζ_{\perp}	$0.33 imes rac{s}{4\pi}$	$\frac{s}{4\pi} \left(16.34 \times \frac{T^3}{\mathcal{B}^{3/2}} \right)$
ζ_{\parallel}	$1.33 imes rac{s}{4\pi}$	$\frac{s}{4\pi} \left(65.37 \times \frac{T^3}{\mathcal{B}^{3/2}} \right)$
ζ_{\times}	$-0.66 imes rac{s}{4\pi}$	$-\frac{s}{4\pi}\left(32.69\times\frac{T^3}{\mathcal{B}^{3/2}}\right)$
r_{\perp}	$\frac{\mathcal{B}}{\mu}\left(1.84 \times \frac{1}{T}\right)$	$\frac{\sqrt{\mathcal{B}}}{\mu} \left(2.34 \times \frac{T^3}{\mathcal{B}^{3/2}} \right)$
r_{\parallel}	$\frac{\mathcal{B}}{\mu}\left(1.71 \times \frac{1}{T}\right)$	$\frac{\sqrt{\mathcal{B}}}{\mu} \left(31.14 \times \frac{T}{\sqrt{\mathcal{B}}} \right)$

Chapter 5. Magnetohydrodynamic waves in a strongly interacting holographic plasma

Table 5.2. Asymptotic behaviour of all first-order transport coefficients in weak- and strong-field limits. The temperature-dependent scaling of the shear viscosities at low temperature agrees with what was reported in Ref. [280].

consistent with expectations that as $\mathcal{B}/T^2 \to 0$, the plasma should becomes an uncharged relativistic conformal hydrodynamics (see e.g. [51] or Appendix 5.7). Furthermore, the resistivities r_{\perp} and r_{\parallel} also tend to zero in the limit.

We also note that the weak-field behaviour of r_{\perp} and r_{\parallel} is consistent with assumptions of standard (ideal) MHD, where conductivity is taken to infinity, $\sigma \approx 1/r \rightarrow \infty$, and where one adds corrections proportional to $1/\sigma$.¹⁷ In other words, small weak-field resistivities are compatible with the assumption of ideal Ohm's law, which gave rise to Eq. (5.7) (see also our discussion around this equation in Section 5.1.). Furthermore, note that in standard MHD, only one resistivity (conductivity) is typically added to include dissipative corrections. What we see is that in our theory, the two resistivities take similar values in the weak-field limit in which standard MHD applies. However, in the strongfield limit, they assume drastically different values, including a different scaling with T/\sqrt{B} . This observation therefore further points to the important role of anisotropic effects in MHD [47] and the necessity for using the formulation of [47, 278] as one moves from the weak to the strong-field regime.

The fact that r_{\perp} and r_{\parallel} tend to zero both in the limits of $T/\sqrt{\mathcal{B}} \rightarrow 0$

¹⁷See Ref. [47] for a discussion regarding the subtleties in relating resistivities to conductivities.

and $T\sqrt{B} \to \infty$, along with the positivity of the entropy production bounds $r_{\perp} \ge 0$ and $r_{\parallel} \ge 0$ [47], implies that there always exists a maximum value of the resistivities at some intermediate T/\sqrt{B} . It would be interesting to find the sizes of these maxima in experimentally realisable systems and probe the regimes of the "least conductive" plasmas. Finally, it would be interesting to further investigate the connection between maximal r and various discussions of lower bounds on conductivities, e.g. [38, 42, 311].

5.4 Magnetohydrodynamic waves in a strongly coupled plasma

We are now ready to use the information obtained from the holographic analysis of Section 5.3 to study dissipative dispersion relations of magnetohydrodynamic waves in a toy model of a strongly coupled plasma. We will use the theory of MHD [47], which is a phenomenological effective theory, and supplement it with microscopic details—the equation of state and transport coefficients—of the holographic setup investigated above. We will be particularly interested in the dependence of the MHD modes on the angle between momentum and magnetic field, as well as the ratio between temperature and the strength of the magnetic field. The 't Hooft coupling of interactions in the matter sector is not tuneable in our model, however, the electromagnetic coupling is. In all sections, except in Section 5.4.3, it will be set to $\alpha = 2\pi^2/137N_c^2$.

Before presenting the numerical result, we review the relevant facts about MHD modes. For a detailed derivation of these results, see Ref. [47] and for a discussion of the general procedure, see Refs. [51, 312]. First, we write the hydrodynamic variables u^{μ} , h^{μ} , T and μ in terms of oscillating modes perturbed around their near-equilibrium values, e.g.

$$u^{\mu} \to (1,0,0,0) + \delta u^{\mu} e^{-i\omega t + ikx\sin\theta + ikz\cos\theta}, \tag{5.95}$$

so that $\theta \in [0, \pi/2]$ measures the angle between the equilibrium magnetic field pointing in the z-direction and the wave momentum k in the x-z plane.

The dispersion relations $\omega(k)$ are then derived from the equations of MHD, i.e. Eqs. (5.8) and (5.9), with the external $H_{\mu\nu\rho} = 0$. The solutions depend on the angle θ , temperature T and the strength of the magnetic field (or the chemical potential of the magnetic flux number density), parametrised in our solutions by \mathcal{B} . Any dimensionless quantity will only depend on the single dimensionless ratio $T/\sqrt{\mathcal{B}}$. The resulting modes can be decomposed into two channels—odd and even under the reflection of $y \to -y$. The first channel is the transverse Alfvén channel. The second is the magnetosonic channel with two branches of solutions: slow and fast magnetosonic waves.

The linearised MHD equations of motion (5.8) and (5.9) need to be expanded in the hydrodynamic regime in powers of small $\omega/\Lambda_h \ll 1$ and $k/\Lambda_h \ll 1$, where Λ_h is the UV cut-off of the effective theory. In standard MHD, where $T \gg \sqrt{\mathcal{B}}$, then $\Lambda_h \approx T$, whereas in the strong-field regime of $T \ll \sqrt{\mathcal{B}}$, the cut-off can be set by the magnetic field, then $\Lambda_h \approx \sqrt{\mathcal{B}}$. As shown in [47], hydrodynamics can exist all the way to $T \to 0$, even when $\delta T = 0$. Such an expansion, performed to some order, gives rise to a polynomial equation in ω and k. For example, in the Alfvén channel, within first-order dissipative MHD,

$$\frac{\omega^2}{k^2} = \left(\frac{\mu\rho\cos^2\theta}{\varepsilon+p}\right) - i\left[\left(\frac{\mu r_{\perp}}{\rho} + \frac{\eta_{\parallel}}{\varepsilon+p}\right)\cos^2\theta + \left(\frac{\mu r_{\parallel}}{\rho} + \frac{\eta_{\perp}}{\varepsilon+p}\right)\sin^2\theta\right]\omega + \frac{\mu}{2\rho(\varepsilon+p)}\left(r_{\perp}\cos^2\theta + 2r_{\parallel}\sin^2\theta\right)\left(\eta_{\perp}\sin^2\theta + \eta_{\parallel}\cos^2\theta\right)k^2.$$
(5.96)

The two solutions of the quadratic equation for ω are given by

$$\omega = -\frac{i}{2} (\mathcal{D}_{A,+}) k^2 \pm \frac{k}{2} \sqrt{\mathcal{V}_A^2 \cos^2 \theta - (\mathcal{D}_{A,-})^2 k^2}, \qquad (5.97)$$

where $\mathcal{D}_{A,+}$ and $\mathcal{D}_{A,-}$ are

$$\mathcal{D}_{A,\pm} = \left(\frac{\mu r_{\perp}}{\rho} \pm \frac{\eta_{\parallel}}{\varepsilon + p}\right) \cos^2 \theta + \left(\frac{\mu r_{\parallel}}{\rho} \pm \frac{\eta_{\perp}}{\varepsilon + p}\right) \sin^2 \theta \,. \tag{5.98}$$

One can now series expand $\omega(k) = D_0 k + D_1 k^2$, or alternatively, plug this ansatz in Eq. (5.96) and solve order-by-oder in k. What we find is the Alfvén

wave dispersion relation [47]:

$$\omega = \pm \mathcal{V}_A k \cos \theta - \frac{i}{2} \Biggl\{ \frac{1}{\varepsilon + p} \left(\eta_\perp \sin^2 \theta + \eta_\parallel \cos^2 \theta \right) + \frac{\mu}{\rho} \left(r_\perp \cos^2 \theta + r_\parallel \sin^2 \theta \right) \Biggr\} k^2 ,$$
(5.99)

where the speed is given by $\mathcal{V}_A^2 = \mu \rho / (\varepsilon + p)$. The dispersion relation appears to be well-defined for any angle $\theta \in [0, \pi/2]$ between momentum and equilibrium magnetic field. In particular, if we were to take the $\theta \to \pi/2$ limit, (5.99) would yield two diffusive modes, both with dispersion relation

$$\omega = -\frac{i}{2} \left(\frac{\eta_{\perp}}{\varepsilon + p} + \frac{\mu r_{\parallel}}{\rho} \right) k^2, \qquad (5.100)$$

which are, however, unphysical and only result from an incorrect order of limits of k and θ .

As can be seen from the structure of the square-root in Eq. (5.97), the expansion in small k is only sensible so long as $k^2 \ll \mathcal{V}_A^2 \cos^2 \theta / (\mathcal{D}_{A,-})^2$. Hence, even for a small finite k, this expansion is inapplicable for angles θ near $\theta = \pi/2$ where $\cos \theta$ becomes very small. In fact, for

$$\mathcal{V}_A^2 \cos^2 \theta \le (\mathcal{D}_{A,-})^2 k^2 \,,$$
 (5.101)

the propagating modes cease to exist altogether and the two modes become purely imaginary (diffusive to $\mathcal{O}(k^2)$). The transmutation of two propagating Alfvén modes into two non-propagating modes occurs when the inequality in (5.101) is saturated, i.e. at the critical angle θ_c when $\text{Re}[\omega] = 0$:

$$\frac{\cos(\theta_c)}{\mathcal{D}_{A,-}(\theta_c)} = \frac{k}{\mathcal{V}_A}.$$
(5.102)

In other words, the plasma exhibits propagating (sound) modes for $0 \le \theta < \theta_c$ and non-propagating (diffusive) modes for $\theta_c < \theta \le \pi/2$. We plot the dependence of the critical angle θ_c on k/\sqrt{B} and T/\sqrt{B} for the Alfvén waves in our model in Figure 5.3. What we see is that for small k/\sqrt{B} and small

 $T/\sqrt{\mathcal{B}}$, the transition to diffusive modes occurs closer to $\theta_c \approx \pi/2$. For any fixed and finite $T/\sqrt{\mathcal{B}}$, Eq. (5.102) indeed implies that $\theta_c \to \pi/2$ as $k \to 0$.

We note that as already pointed out in [278], the limits of $k \to 0$ and $\theta \to \pi/2$ do not commute and we obtain different results depending on which expansion ($k \approx 0$ or $\theta \approx \pi/2$) is performed first. If one first takes the limit $\theta \to \pi/2$, then Eq. (5.96) becomes

$$-\omega^2 - i\left(\frac{\mu r_{\parallel}}{\rho} + \frac{\eta_{\perp}}{\varepsilon + p}\right)\omega k^2 + \frac{\mu r_{\parallel}\eta_{\perp}}{\rho(\varepsilon + p)}k^4 = 0, \qquad (5.103)$$

which instead of Eq. (5.100) results in two non-degenerate diffusive modes

$$\omega = -i\frac{\eta_{\perp}}{\varepsilon + p}k^2, \qquad \qquad \omega = -i\frac{\mu r_{\parallel}}{\rho}k^2.$$
(5.104)

The dispersion relation (5.99) is therefore only sensible at a finite $T/\sqrt{\mathcal{B}}$ and infinitesimally small k/Λ_h .

In the magnetosonic channel, the story is entirely analogous to the one described for the Alfvén waves. By expanding around $k \approx 0$ first, we obtain the dispersion relation of [47]:

$$\omega = \pm v_M k - i\tau k^2 \,, \tag{5.105}$$

where the speed of magnetosonic wave is given by

$$v_{M}^{2} = \frac{1}{2} \Biggl\{ (\mathcal{V}_{A}^{2} + \mathcal{V}_{0}^{2}) \cos^{2} \theta + \mathcal{V}_{S}^{2} \sin^{2} \theta \\ \pm \sqrt{[(\mathcal{V}_{A}^{2} + \mathcal{V}_{0}^{2}) \cos^{2} \theta + \mathcal{V}_{S}^{2} \sin^{2} \theta]^{2} + 4\mathcal{V}^{4} \cos^{4} \sin^{2} \theta} \Biggr\}.$$
(5.106)

The functions V_A , V_0 , V_S and V appearing in (5.106) are

$$\begin{aligned}
\mathcal{V}_{A}^{2} &= \frac{\mu\rho}{\varepsilon + p}, \\
\mathcal{V}_{0}^{2} &= \frac{s}{T\chi_{11}}, \\
\mathcal{V}_{S}^{2} &= \frac{(s - \rho\chi_{12})(s + \rho\chi_{21}) + \rho^{2}\chi_{11}\chi_{22}}{(\varepsilon + p)\chi_{11}}, \\
\mathcal{V}^{4} &= \frac{s(s - \rho\chi_{12})(s + \rho\chi_{21})}{T(\varepsilon + p)\chi_{11}^{2}}.
\end{aligned}$$
(5.107)

The susceptibilities are¹⁸

$$\chi_{11} = \left(\frac{\partial s}{\partial T}\right)_{\rho}, \quad \chi_{12} = \left(\frac{\partial s}{\partial \rho}\right)_{T}, \quad \chi_{21} = \left(\frac{\partial \mu}{\partial T}\right)_{\rho}, \quad \chi_{22} = \left(\frac{\partial \mu}{\partial \rho}\right)_{T}.$$
(5.108)

The two types of magnetosonic waves, corresponding to \pm solutions in (5.106), are known as the fast (with +) and the slow (with -) magnetosonic waves. We refer the reader to Appendix 5.8 for further details regarding the derivation of the magnetosonic modes. Each pair of the propagating slow magnetosonic modes also splits, in analogy with the Alfvén waves, into two non-propagating diffusive modes for $\theta \geq \theta_c$. The critical angle θ_c for magnetosonic modes is also defined as in the Alfvén channel: the angle at which $\operatorname{Re}[\omega] = 0$. We plot the numerically-computed dependence of the magnetosonic θ_c on k/\sqrt{B} and T/\sqrt{B} in Fig. 5.3. As can be seen from the plot, the critical angles for the two types of waves are independent. However, they show similar qualitative dependence on the parameters that characterise the waves.

We summarise the θ -dependent characteristics of MHD modes in Fig. 5.4. We observe the pattern of a transmutation of sound modes into diffusion to be different in the weak- and strong-field regimes. Namely, the two magnetosonic waves interchange their dispersion relations at small θ . Since the complicated expressions for dispersion relations greatly simplify at $\theta = 0$ and $\theta = \pi/2$, we

¹⁸Note that these susceptibilities are different to the ones used in [47], where independent thermodynamic quantities were T and μ , not T and ρ . For this reason we also use different notation.

Chapter 5. Magnetohydrodynamic waves in a strongly interacting holographic plasma



Figure 5.3. The critical angle θ_c for Alfvén waves (left) and slow magnetosonic waves (right), plotted as a function of T/\sqrt{B} for $k/\sqrt{B} = \{0.1, 0.2, 0.4, 0.6\}$. The dashed line at the top of both sub-figures indicates the value of $\theta_c = \pi/2$.

state them below. The sound mode dispersion relations, denoted by S, are

S1:
$$\omega = \pm \mathcal{V}_{S}k - \frac{i}{2} \Biggl\{ \frac{\zeta_{\perp} + \eta_{\perp}}{\varepsilon + p} + r_{\perp} \Biggl\{ \frac{\left[(s - \rho\chi_{12})(\mu - T\chi_{21}) - \rho T\chi_{11}\chi_{22} \right] \times \left[\chi_{12} \leftrightarrow -\chi_{21} \right]}{T^{2}\chi_{11} \left[(s - \rho\chi_{12})(s + \rho\chi_{21}) + \rho^{2}\chi_{11}\chi_{22} \right]} \Biggr\} \Biggr\} k^{2}$$
S2:
$$\omega = \pm \mathcal{V}_{A}k - \frac{i}{2} \left(\frac{\eta_{\parallel}}{\varepsilon + p} + \frac{\mu r_{\perp}}{\rho} \right) k^{2},$$
S3:
$$\omega = \pm \mathcal{V}_{0}k - \frac{i}{2} \frac{\zeta_{\parallel}}{sT} k^{2},$$
(5.109)

and the diffusive modes, denoted by D, are

ρ

D1:
$$\omega = -i\frac{\eta_{\parallel}}{sT}k^{2}$$
,
D2: $\omega = -\frac{ir_{\perp}(\varepsilon + p)^{2}\chi_{22}}{T^{2}[(s - \rho\chi_{12})(s + \rho\chi_{21}) + \rho^{2}\chi_{11}\chi_{22}]}k^{2}$,
D3: $\omega = -i\frac{\eta_{\perp}}{\varepsilon + p}k^{2}$,
D4: $\omega = -i\frac{r_{\parallel}\mu}{k}k^{2}$.
(5.110)

Chapter 5. Magnetohydrodynamic waves in a strongly interacting holographic plasma



Figure 5.4. Diagrams depicting the θ -dependent pattern of transmutation from sound to diffusive modes for Alfvén waves and slow and fast magnetosonic waves. The left and right diagrams correspond to weak- and strong-field regimes. The relevant dispersion relation are stated in Eqs. (5.109) and (5.110).

In the regime of a large T/\sqrt{B} , the results agree with those of [278]. Furthermore, using the asymptotic form of the thermodynamics quantities and transport coefficients in the $T/\sqrt{B} \to \infty$ limit, one can show that these modes reduce to sound and diffusive modes of uncharged relativistic hydrodynamics.

In the strong-field regime, which cannot be described within standard MHD, the speeds of S1 and S3 become large and approach the speed of light in the limit of $T \rightarrow 0$. As discussed before, all diffusion constants vanish and the system becomes controlled by second-order MHD [47], which we do not investigate in this work. Further details of angle-dependent wave propagation are presented in Section 5.4.2.

5.4.1 Speeds and attenuations of MHD waves

Here, we plot the speeds (phase velocities) and first-order attenuation coefficients of the three types of MHD sound waves: the Alfvén and the fast and slow magnetosonic waves for the holographic strongly coupled plasma discussed above. These results assume an infinitesimally small value of momentum k, and follow from first expanding the polynomial equation of the type of (5.96) around $k \approx 0$ and writing the dispersion relation as $\omega = \pm vk - i\mathcal{D}k^2$. The

speeds v (presented in Fig. 5.5) and attenuation coefficients \mathcal{D} (presented in Fig. 5.6) are then plotted for all $0 \le \theta \le \pi/2$, which, as discussed above, is only physically sensible when $\theta_c \to \pi/2$, i.e. as $k \to 0$.



Figure 5.5. Angular dependence of the speeds of Alfvén (black, solid), fast (blue, dotted) and slow (red, dashed) magnetosonic waves in the strong-field, the crossover and the weak-field regimes.

The angular profiles of the speeds and the dissipative attenuation coefficients show distinct behaviour in the strong-, the crossover (cf. Eq. (5.83)) and the weak-field regimes. In particular, the speeds of sound enter the weak-field regime, where they reduce to well-known standard MHD results, rapidly after the temperature exceeds $T/\sqrt{\mathcal{B}} \approx 0.7$. There, Alfvén and slow magnetosonic waves travel with very similar speeds for all θ and their speeds coincide at $\theta = 0$ and $\theta = \pi/2$. The situation is qualitatively different in the strong-field regime where the profiles of speeds qualitatively match the strong-field predictions of [47], but of whicthe sh the behaviour was to our knowledge previously unknown. There, slow magnetosonic and Alfvén waves can travel faster at small θ , with speeds comparable to those of fast magnetosonic waves. At $\theta = 0$, the



Chapter 5. Magnetohydrodynamic waves in a strongly interacting holographic plasma

Figure 5.6. Angular dependence of the (dimensionless) attenuation coefficients of Alfvén (black, solid), fast (blue, dotted) and slow (red, dashed) magnetosonic waves, $\mathcal{D}\sqrt{\mathcal{B}}$, in the strong-field, the crossover and the weak-field regimes.

Alfvén speed equals that of fast, instead of slow, magnetosonic waves (cf. Fig. 5.4). It should also be noted that there exists a value of T/\sqrt{B} in the crossover regimes where all three speeds are equal at $\theta = 0$.

The attenuation coefficients, computed with all seven transport coefficients [47, 278], are computed for the first time for a concrete microscopically (holographically) realisable plasma and therefore difficult to compare with other past results. What we observe is that the Alfvén waves experience the strongest damping for all values of $T/\sqrt{\mathcal{B}}$. Beyond that, the qualitative behaviour again displays distinct angle-dependent features in the three regimes, which are apparent from Fig. 5.6. A noteworthy, but not surprising feature is that the strength of attenuation appears to be much more strongly dependent on the angle between momentum and magnetic field in the regime of small $T/\sqrt{\mathcal{B}}$. Furthermore, in the crossover regime, we find that the strengths of fast and slow magnetosonic mode attenuations interchange roles at $T/\sqrt{\mathcal{B}}$ increases. In plots at $T/\sqrt{\mathcal{B}} = 0.5$ and $T/\sqrt{\mathcal{B}} = 0.66$, there exists an angle θ at which the two attenuation strengths coincide.

5.4.2 MHD modes on a complex frequency plane

By assuming a finite value of momentum k, a full analysis of the spectrum requires us to take into account the transmutation of sound modes into nonpropagating diffusive modes. The pattern of this behaviour, as a function of the angle between momentum and the direction of the equilibrium magnetic field θ , was summarised in Fig. 5.4. Motivated by holographic quasinormal mode (poles of two-point correlators) analyses, we plot the motion of the MHD modes on the complex frequency plane—here, as a function of θ and $T/\sqrt{\mathcal{B}}$. One should consider these plots as a prediction of how the first-order approximation to the hydrodynamic sector of the full quasinormal spectrum computed from the theory (5.43) is expected to behave.

In Fig. 5.7, we plot the typical θ -dependent trajectories of $\omega(\theta)$ on the complex ω -plane for Alfvén and magnetosonic modes in distinctly strong- and weak-field regimes. At all temperatures (except at T = 0 where $\mathcal{D} = 0$), the behaviour is consistent with our previous discussions, including the fact that the transmutation of Alfvén and slow magnetosonic waves into diffusive modes occurs at lower θ_c as $k/\sqrt{\mathcal{B}}$ increases.

In the crossover temperature regime (around $T/\sqrt{B} \approx 0.6$), we find another manifestation of the interplay between fast and slow magnetosonic modes, which was noted in Section 5.4.1. While the speed of fast magnetosonic waves always exceeds that of slow waves, their attenuation strength exchange roles around $T/\sqrt{B} \simeq 0.675$, which manifests in characteristically distinct behaviour for $\theta < \theta_c$. The behaviour is presented in Fig. 5.8 (see also Fig. 5.6). The θ -dependence of Alfvén waves remains qualitatively similar to those depicted in Fig. 5.7.

For a fixed $\theta < \theta_c$, where θ_c depends on k and $T/\sqrt{\mathcal{B}}$, we plot the typical behaviour of $\omega(k)$ as a function of $T/\sqrt{\mathcal{B}}$ in Fig. 5.9. At T = 0, all poles start from the non-dissipative regime (the real \mathfrak{w} axis), with the speed of fast magnetosonic waves given by v = 1. As they move towards larger $T/\sqrt{\mathcal{B}}$, the Alfvén and slow magnetosonic modes again asymptote to each other, eventually



Chapter 5. Magnetohydrodynamic waves in a strongly interacting holographic plasma

Figure 5.7. Dependence of the complex (dimensionless) frequency $\mathfrak{w} = \omega/\sqrt{\mathcal{B}}$ on θ , plotted for Alfén (black) and fast (blue) and slow (red) magnetosonic waves in the strong- and weak-field regimes with $T/\sqrt{\mathcal{B}} = 0.4$ and $T/\sqrt{\mathcal{B}} = 1.15$, respectively. The arrows represent the motion of poles as θ is tuned from 0 to $\pi/2$. Momentum is set to $k/\sqrt{\mathcal{B}} = 0.05$.

transforming into diffusive modes, while the speed of the fast magnetosonic modes gradually converges towards that of neutral conformal sound with $v = 1/\sqrt{3}$.

In the high temperature limit, the collision of the Alfvén and, independently, the slow magnetosonic modes on the imaginary axis occurs close to the real axis, which follows from the fact that for both types of waves,

$$\operatorname{Im}\left[\mathfrak{w}\right] \approx \frac{(\eta_{\perp} \sin^2 \theta + \eta_{\parallel} \cos^2 \theta) \sqrt{\mathcal{B}}}{2(\varepsilon + p)} \sim \frac{\sqrt{\mathcal{B}}}{T} \to 0, \qquad (5.111)$$

as $T/\sqrt{B} \to \infty$. The Alfvén waves then become the diffusive modes of uncharged conformal hydrodynamics with $\omega = -i\eta k^2/(2sT)$. As for our final



Chapter 5. Magnetohydrodynamic waves in a strongly interacting holographic plasma

Figure 5.8. Dependence of the complex (dimensionless) frequency $\mathfrak{w} = \omega/\sqrt{\mathcal{B}}$ of fast (blue) and slow (red) magnetosonic modes on θ in the crossover regime. The arrows represent the motion of poles as θ is tuned from 0 to $\pi/2$. Momentum is set to $k/\sqrt{\mathcal{B}} = 0.05$.

plot, in Fig. 5.10, we present the dependence of the four diffusion constants and one sound attenuation coefficient on the temperature at $\theta = \pi/2$ (cf. Fig. 5.4 and Eqs. (5.109)–(5.110)). The modes D1, D3 and S1 reduce to dispersion relations of uncharged relativistic hydrodynamics. D2 and D4 are new.

5.4.3 Electric charge dependence

We end our discussion of MHD dispersion relations by investigating their dependence on the choice of the U(1) coupling constant, which has so far been set to the (N_c -rescaled) $\bar{\alpha} = 1/137$. All dependence on $\bar{\alpha}$ enters into the expectation value of the stress-energy tensor through the term proportional to $\mathcal{H}_{\mu\nu}\mathcal{H}^{\mu\nu} \ln C$ (cf. Eq. (5.61)), which contributes no terms linear in ω . For this



Chapter 5. Magnetohydrodynamic waves in a strongly interacting holographic plasma

Figure 5.9. Dependence of the complex (dimensionless) frequency $\mathfrak{w} = \omega/\sqrt{\mathcal{B}}$ on $T/\sqrt{\mathcal{B}}$, plotted for Alfén (black) and fast (blue) and slow (red) magnetosonic waves for $\theta < \theta_c$. The arrows represent the motion of poles as $T/\sqrt{\mathcal{B}}$ is tuned from 0 towards the weak-field regime. Momentum is set to $k/\sqrt{\mathcal{B}} = 0.01$.

reason, while the equation of state strongly depends on $\bar{\alpha}$, the first-order transport coefficients do not. Hence, all speeds of sound and attenuation (and diffusive) coefficients depend on the choice of $\bar{\alpha}$ through the equation of state and susceptibilities.

What we observe is that the speeds of waves and attenuation coefficients strongly depend on the renormalised electromagnetic coupling, so unsurprisingly, the strength of electromagnetic interactions plays an important role in the phenomenology of MHD. For concreteness, we only present the detailed behaviour of the Alfvén waves (with speed $\mathcal{V}_A \cos \theta$), which reduce the neutral hydrodynamic diffusion mode D3 (and D4) at $\theta = \pi/2$. Both \mathcal{V}_A and the diffusion constant of D3, \mathcal{D}_{D3} , strongly depend on $\bar{\alpha}$. On the other hand, it is interesting that the speed of S3 mode, \mathcal{V}_0 , does not depend on $\bar{\alpha}$. We note that \mathcal{V}_0 is the $\theta = 0$ limit of fast and slow magnetosonic mode speeds in the weak-

Chapter 5. Magnetohydrodynamic waves in a strongly interacting holographic plasma



Figure 5.10. Plots of the four diffusion constants (D1, D2, D3, D4) and the sound attenuation (S1) as a function of T/\sqrt{B} at $\theta = \pi/2$. Black, red and blue curves depict dissipative coefficients that originate from the Alfvén, slow magnetosonic and fast magnetosonic waves, respectively.

and strong-field limits, For reasonable values of $\bar{\alpha}$, we plot the results in Fig. 5.11. To show the importance of a sensible choice of the renormalisation condition, we also vary the coupling over a larger range (to $\bar{\alpha} = 80/137$), where we see that the system develops unphysical behaviour with instabilities. As is apparent from Fig. 5.12, Alfvén waves become unstable at low $T/\sqrt{\mathcal{B}}$.



Figure 5.11. The plot of \mathcal{V}_A^2 and the diffusion constant \mathcal{D}_{D3} at $\bar{\alpha} = \{\bar{\alpha}_0/2, \bar{\alpha}_0, 2\bar{\alpha}_0\}$, where $\bar{\alpha}_0 = 1/137$. The dashed line is the $\bar{\alpha}$ -independent \mathcal{V}_0^2 , which is plotted for comparison.

In all to us known literature, the unavoidable choice of the constant C, which sets $\bar{\alpha}$ is made in a different way. Either C is chosen so that the loga-

Chapter 5. Magnetohydrodynamic waves in a strongly interacting holographic plasma



Figure 5.12. The plot of the Alfvén \mathcal{V}_A^2 at a varying $\bar{\alpha}$ ranging from $\bar{\alpha} = \bar{\alpha}_0$ to $\bar{\alpha} = 80\bar{\alpha}_0$, where $\bar{\alpha}_0 = 1/137$. We see that as $\bar{\alpha}$ increases, the waves develop an instability in the strong-field regime.

rithmic terms vanish altogether, or so that it sets the UV scale to be set by the magnetic field. The latter choice is convenient when studying strong magnetic fields, as e.g. in [284, 292]. Here, we would like to point out some of the consequences of setting C to either of the two standard options. Following the first choice, which eliminates the logarithmic terms, the resulting thermodynamics quantities are

$$\varepsilon = \frac{N_c^2}{2\pi^2} \left(-\frac{3}{4} f_4^b r_h^4 \right) ,$$

$$p = \frac{N_c^2}{2\pi^2} \left[\left(-\frac{1}{4} f_4^b + \frac{v_4^b}{v} \right) r_h^4 - \frac{\mathcal{B}^2}{4} \right] ,$$

$$\mu \rho = \frac{N_c^2}{2\pi^2} \left(\frac{3v_4^b}{v} r_h^4 - \frac{\mathcal{B}^2}{4} \right) .$$

(5.112)

The second choice, results in

1

$$\varepsilon = \frac{N_c^2}{2\pi^2} \left(-\frac{3}{4} f_4^b r_h^4 + \frac{\mathcal{B}^2}{4} \ln \mathcal{B} \right),$$

$$p = \frac{N_c^2}{2\pi^2} \left[\left(-\frac{1}{4} f_4^b + \frac{v_4^b}{v} \right) r_h^4 - \frac{\mathcal{B}^2}{4} + \frac{\mathcal{B}^2}{4} \ln \mathcal{B} \right],$$
 (5.113)

$$\mu \rho = \frac{N_c^2}{2\pi^2} \left(\frac{3v_4^b}{v} r_h^4 - \frac{\mathcal{B}^2}{4} - \frac{\mathcal{B}^2}{4} \ln \mathcal{B} \right).$$

What we would like to claim is that while these two convenient renormalisa-

Chapter 5. Magnetohydrodynamic waves in a strongly interacting holographic plasma

tion conditions are suitable for studying certain physical setups involving static electromagnetic fields, they lead to unphysical results when the boundary U(1)gauge field is dynamical. By comparing the renormalised stress-energy tensor (5.74)–(5.76) to expressions in (5.112) and (5.113), we find that the two choices correspond to the renormalised coupling being $e_r^2 \to \infty$ and $e_r^2 \sim \ln \mathcal{B}$, respectively. The first choice is clearly unusual and unphysical. The problem with the second choice is that in certain regimes, $\ln \mathcal{B}$ can become negative and e_r imaginary, which is again unphysical. Both lead to instabilities and superluminal propagation, which were absent in from our results for $\bar{\alpha}$ near 1/137. We plot the Alfvén speed parameter \mathcal{V}_A for the two couplings from (5.112) and (5.113) in Fig. 5.13.



Figure 5.13. The θ -independent factor \mathcal{V}_A of the Alfvén wave speed plotted for the renormalised $e_r^2 \to \infty$ (left) from Eq. (5.112) and for $e^2 \sim \ln \mathcal{B}$ (right) from Eq. (5.113).

5.5 Discussion

This work should be considered as the first holographic step in a long road towards a better understanding of magnetohydrodynamics in plasmas outside of the regime of validity of standard MHD, be it in the presence of strong magnetic fields or in a strongly interacting (or dense) plasma with a complicated equation of state and transport coefficients—all claimed to be describable within the recent (generalised global) symmetry-based formulation of MHD of Ref. [47]. In order to supply a hydrodynamical theory of MHD with the necessary micro-

scopic information of a strongly coupled plasma, we resorted to the simplest, albeit experimentally inaccessible option: holography. Nevertheless, our hope is that in analogy with the myriad of works on holographic conformal hydrodynamics, which have led to important new insights into strongly interacting realistic fluids, holography can also help us understand MHD in the presence of strong fields, high density and of strongly interacting gauge theories, such as QCD.

With this view, we constructed the simplest theory dual to the operator structure and Ward identities used in MHD of [47], investigated the relevant aspects of the holographic dictionary and used it to compute the equation of state and transport coefficients of the dual plasma state. This information was then used to analyse the dependence of MHD waves—Alfvén and magnetosonic waves—on tuneable parameter of the state: the strength of the magnetic field, temperature, the angle between momentum of propagation and the equilibrium magnetic field direction, as well as the strength of the U(1) electromagnetic gauge coupling. We believe that the latter feature of our model—dynamical electromagnetism on the boundary—which in the S-dual language of two-form gauge fields in the bulk allows for standard (Dirichet) quantisation, could in its own right be used for holographic studies of U(1)-gauged systems, unrelated to MHD.

Our results have revealed several new qualitative features of MHD waves, particularly in the regime of a strong magnetic field, which was previously inaccessible to standard MHD methods. Various properties of the equation of state, transport coefficients and dispersion relations may now be compared to those in experimentally realisable plasmas, or at the least, used as a toy model for future studies of MHD. The scalings are collected in Tables 5.1 and 5.2. Here, we summarise some of the most interesting observations:

• The equation of state and transport coefficients strongly depend on the strength of the magnetic field, i.e. on whether the plasma is in weak-field, crossover, or strong-field regime.

• In the weak-field limit of $T/\sqrt{\mathcal{B}} \gg 1$, the system is well-described by

standard MHD (see [278] for a full description) with small resistivities (large conductivity regime, which is assumed by ideal Ohm's law) and small effects of anisotropy. As $T/\sqrt{\mathcal{B}} \to \infty$, the plasma becomes an uncharged, conformal fluid with a single independent transport coefficient, $\eta = s/4\pi$. In the strong-field limit of $T/\sqrt{\mathcal{B}} \ll 1$, the plasma limits to a non-dissipative regime with all first-order transport coefficients (along with sound attenuations and diffusion constants) tending to zero. Effects of anisotropy are large.

• Resistivities have a global maximum in the intermediate T/\sqrt{B} regime, which indicates a regime of least conductive plasma. If the assumptions of standard MHD are correct at $T/\sqrt{B} \gg 1$ and the symmetry-based predictions of [47] are correct at $T/\sqrt{B} \ll 1$, such a regime should be generically exhibited by any plasma.

• Out of the three bulk viscosities, ζ_{\perp} , ζ_{\parallel} and ζ_{\times} , only one is independent and they saturate the positivity of the entropy production inequality, i.e. they are related by $\zeta_{\perp}\zeta_{\parallel} = \zeta_{\times}^2$. One may speculate on how general this result is and whether it is related to the suppression of entropy production at strong coupling [149, 150] or perhaps some form holographic universality at infinite (or strong) coupling.

• Various qualitative features of slow and fast magnetosonic modes are exchanged in the weak- and strong-field regimes at small angle θ between momentum and equilibrium magnetic field direction, such as their asymptotic tendency to the speed of Alfvén waves and the strength of sound attenuation.

• For a finite momentum, propagating (sound to $\mathcal{O}(k^2)$) Alfvén and slow magnetosonic modes transmute into pairs of non-propagating diffusive (to $\mathcal{O}(k^2)$) modes at large angles between the direction of momentum propagation and the equilibrium magnetic field, $\theta_c < \theta \leq \pi/2$, where θ_c is some momentum- and $T/\sqrt{\mathcal{B}}$ -dependent critical angle (cf. Eq. (5.102) for Alfvén waves).

• The phenomenology of MHD modes strongly depends on the strength of electromagnetic coupling and can, for large ranges of the coupling, lead to unstable or superluminal propagation.

Beyond the types of waves studied in this work, it would be particularly

interesting to better understand the role finite charge density, as studied in [278], within the formalism of [47]. The important question then is how the phenomenology of such MHD waves, which typically experience gapped propagation (Langmuir waves) and instabilities (e.g. the infamous Weibel instability), becomes altered by strong interactions, strong fields and more 'exotic' field content.

Finally, the holographic setup studied here will need to undergo extensive further tests and analyses in order to unambiguously establish its connection to plasma physics and MHD. In particular, it is essential to study the quasinormal spectrum of the theory to verify that the hydrodynamic modes indeed describe the small- ω and small-k expansion of the leading infrared poles. Furthermore, it will be interesting to understand the role of higher-frequency spectrum and its interplay with MHD modes. We leave all these and many other interesting questions to the future.

5.6 Appendices

5.6.1 Kubo formulae for first-order transport coefficients

In this appendix, we outline the derivation of the Kubo formulae that have been used to compute the seven first-order transport coefficients in (5.12) and (5.14)–(5.19) in Section 5.3.4 [47, 278]. We derive the Kubo formulae by using the variational background field method (see e.g. [51] for a review), which amounts to varying the background metric $g_{\mu\nu}$ and background two-form gauge field $b_{\mu\nu}$, sourcing $T^{\mu\nu}$ and $J^{\mu\nu}$, by writing

$$g_{\mu\nu} \to \eta_{\mu\nu} + \int \frac{d\omega}{2\pi} e^{-i\omega t} \delta h_{\mu\nu}(\omega), \quad b_{\mu\nu} \to b_{\mu\nu}^{\rm eq} + \int \frac{d\omega}{2\pi} e^{-i\omega t} \delta b_{\mu\nu}(\omega),$$
(5.114)

where $\delta h_{\mu\nu}$ and $\delta b_{\mu\nu}$ are small variation, η is the flat Minkowski metric and $b_{\mu\nu}^{\rm eq} = 2\mu u^{[\mu} h^{\nu]}$. These variations of background fields can be viewed as a sources that generate a variation in the hydrodynamic variables T, ρ (which

we use here instead of μ in [47]), u^{μ} and h^{μ} :

$$T(t) \to T + \delta T(t), \qquad \rho \to \rho + \delta \rho(t), \qquad (5.115)$$

$$u^{\mu} \rightarrow u^{\mu}_{\rm eq} + \delta u(t) , \qquad \qquad h^{\mu} \rightarrow h^{\mu}_{\rm eq} + \delta h^{\mu}(t) , \qquad (5.116)$$

where we choose the equilibrium configuration to be $u_{eq}^{\mu} = \delta_t^{\mu}$ and $h_{eq}^{\mu} = \delta_z^{\mu}$. The normalisation and orthogonality conditions for the two vectors ($u_{\mu}u^{\mu} = -1$, $h_{\mu}h^{\mu} = 1$, $u_{\mu}h^{\mu} = 0$) imply

$$\delta u^t = \frac{1}{2} \delta h_{tt}, \qquad \delta h^t = \delta u^z + \delta h_{tz}, \qquad \delta h^z = -\frac{1}{2} \delta h_{zz}. \tag{5.117}$$

After writing δT , $\delta \rho$, δu^{μ} and δh^{μ} in terms of $\delta h_{\mu\nu}$ and $\delta b_{\mu\nu}$, we can insert these solution into

$$\mathcal{T}^{\mu\nu} \equiv \sqrt{-g} \langle T^{\mu\nu} \rangle|_{g,b}, \qquad \qquad \mathcal{J}^{\mu\nu} \equiv \sqrt{-g} \langle J^{\mu\nu} \rangle|_{g,b}, \qquad (5.118)$$

which give

$$\operatorname{Im} \mathcal{T}^{xx} + \operatorname{Im} \mathcal{T}^{yy} = \omega \zeta_{\perp} (\delta h_{xx} + \delta h_{yy}) + \omega \zeta_{\times}^{(1)} \delta h_{zz} + \mathcal{O}(\omega^{2}, \delta h^{2}, \delta b^{2}),$$

$$\operatorname{Im} \mathcal{T}^{zz} = \frac{1}{2} \omega \zeta_{\parallel} \delta h_{zz} + \frac{1}{2} \omega \zeta_{\times}^{(2)} (\delta h_{xx} + \delta h_{yy}) + \mathcal{O}(\omega^{2}, \delta h^{2}, \delta b^{2}),$$

$$\operatorname{Im} \mathcal{T}^{xy} = \omega \eta_{\perp} \delta h_{xy} + \mathcal{O}(\omega^{2}, \delta h^{2}, \delta b^{2}),$$

$$\operatorname{Im} \mathcal{T}^{xz} = \omega \eta_{\parallel} \delta h_{xz} + \mathcal{O}(\omega^{2}, \delta h^{2}, \delta b^{2}),$$

$$\operatorname{Im} \mathcal{J}^{xy} = 2 \omega r_{\parallel} \delta b_{xy} + \mathcal{O}(\omega^{2}, \delta h^{2}, \delta b^{2}),$$

$$\operatorname{Im} \mathcal{J}^{xz} = 2 \omega r_{\perp} \delta b_{xz} + \mathcal{O}(\omega^{2}, \delta h^{2}, \delta b^{2}),$$

$$(5.119)$$

(5.119) where we have not imposed the Onsager relation equating $\zeta_{\times}^{(1)}$ with $\zeta_{\times}^{(2)}$ [47, 278]. By using the linear response formulae relating the variations of one-point functions to retarded two-point Green's functions,

$$\delta \mathcal{T}^{\mu\nu}(\omega, \mathbf{k}) = -\frac{1}{2} G_{TT}^{\mu\nu,\lambda\sigma}(\omega, \mathbf{k}) \delta h_{\lambda\sigma}(\omega, \mathbf{k}) - \frac{1}{2} G_{TJ}^{\mu\nu,\lambda\sigma}(\omega, \mathbf{k}) \delta b_{\lambda\sigma}(\omega, \mathbf{k}),$$

$$\delta \mathcal{J}^{\mu\nu}(\omega, \mathbf{k}) = -\frac{1}{2} G_{JT}^{\mu\nu,\lambda\sigma}(\omega, \mathbf{k}) \delta h_{\lambda\sigma}(\omega, \mathbf{k}) - \frac{1}{2} G_{JJ}^{\mu\nu,\lambda\sigma}(\omega, \mathbf{k}) \delta b_{\lambda\sigma}(\omega, \mathbf{k}),$$

(5.120)

it is then easy to extract the relevant Kubo formulae for the seven transport coefficients [47, 278], which we used in this work:

$$\eta_{\parallel} = \lim_{\omega \to 0} \frac{G_{TT}^{xz,xz}(\omega,0)}{-i\omega}, \qquad \eta_{\perp} = \lim_{\omega \to 0} \frac{G_{TT}^{xy,xy}(\omega,0)}{-i\omega}, \qquad (5.121)$$

$$\zeta_{\parallel} = \lim_{\omega \to 0} \frac{G_{TT}^{zz,zz}(\omega,0)}{-i\omega} , \qquad \zeta_{\perp} + \eta_{\perp} = \lim_{\omega \to 0} \frac{G_{TT}^{xx,xx}(\omega,0)}{-i\omega} , \qquad (5.122)$$

as well as

$$\zeta_{\times} = \lim_{\omega \to 0} \frac{G_{TT}^{zz,xx}(\omega,0)}{-i\omega} = \lim_{\omega \to 0} \frac{G_{TT}^{xx,zz}(\omega,0)}{-i\omega} \,. \tag{5.123}$$

and

$$r_{\parallel} = \lim_{\omega \to 0} \frac{G_{JJ}^{xy,xy}(\omega,0)}{-i\omega}, \qquad r_{\perp} = \lim_{\omega \to 0} \frac{G_{JJ}^{xz,xz}(\omega,0)}{-i\omega}.$$
(5.124)

5.7 Further details regarding the derivation of the transport coefficients

Here, we show the details of the derivation of horizon formulae of all remaining transport coefficient: η_{\perp} , η_{\parallel} , ζ_{\perp} , ζ_{\parallel} , ζ_{\times} and r_{\parallel} . The computation are analogous to the calculation of r_{\perp} in Section 5.3.4.

(i) Shear viscosity η_{\perp}

The only relevant bulk fluctuation for η_{\perp} is δG_{xy} with the equation of motion

$$\delta G_x^{y\,''} + \left(\frac{3}{2u} + \frac{F'}{F} + 2\mathcal{V}' + \mathcal{W}'\right) \delta G_x^{y\,'} + \frac{\omega^2}{4r_h^2 u^3 F^2} \delta G_x^y = 0.$$
(5.125)

The solution to leading order in the frequency ω can be found analytically and its near-boundary expansion gives

$$\delta G^y_{\ x} = \delta h_{xy} \left(1 + \frac{i\omega u^2}{2r_h v\sqrt{w}} + \mathcal{O}(u^3) \right), \tag{5.126}$$

where δh_{xy} is the Dirichlet background condition and the boundary theory

source. If we plug in this solution into to the stress-energy tensor, we find that

$$\langle \delta T^{xy} \rangle = \frac{N_c^2}{2\pi^2} \left(\frac{r_h^4 e^{2\nu} \sqrt{uF}}{2\nu} \delta G_x^{y'} \right) + \dots$$

$$= \frac{N_c^2}{2\pi^2} \left(\frac{i\omega r_h^3}{4\nu\sqrt{w}} \right) \delta h_{xy} + \dots ,$$
(5.127)

Using Eq. (5.119), we find that

$$\eta_{\perp} = \frac{N_c^2}{2\pi^2} \left(\frac{r_h^3}{4v\sqrt{w}} \right) = \frac{1}{4\pi} s \,, \tag{5.128}$$

as stated in Eq. (5.85).

(ii) Shear viscosity η_{\parallel}

Similarly to the computation of r_{\perp} , the *xu*-component of the two-form gauge field fluctuation equation can be used to reduced the two coupled second order differential equations coupling δG_{xz} and δB_{tx} to a single equation:

$$\delta G_x^{z}'' + \left(\frac{3}{2u} + \frac{F'}{F} + 3\mathcal{W}'\right) \delta G_x^{z}' + \frac{\omega^2}{4r_h^2 u^3 F^2} \delta G_x^z = 0.$$
(5.129)

The solution to linear order in ω can again be found analytically and in the near-boundary region yields

$$\delta G_x^z = \delta h_x^z \left(1 + \frac{i\omega}{r_h w^{3/2}} u^2 + \mathcal{O}(u^3) \right).$$
 (5.130)

The relevant component of the stress-energy tensor is then

$$\langle T^{xz} \rangle = \frac{N_c^2}{2\pi^2} \left(\frac{i\omega r_h^3}{4w^{3/2}} \right) \delta h_{xz} + \dots , \qquad (5.131)$$

which gives

$$\eta_{\parallel} = \frac{N_c^2}{2\pi^2} \left(\frac{r_h^3}{4w^{3/2}} \right) = \frac{1}{4\pi} \frac{v}{w} s \,, \tag{5.132}$$

as stated in Eq. (5.85).

(iii) Resistivity r_{\parallel}

The only equation of motion in this channel is

$$\delta B_{xy}'' + \left(\frac{3}{u} + \frac{F'}{F} - 2\mathcal{V}' + \mathcal{W}'\right) \delta B_{xy}' + \frac{\omega^2}{4r_h^2 u^3 F^2} \delta B_{xy} = 0, \qquad (5.133)$$

which leads to the near-boundary solution

$$\delta B_{xy} = \delta B_{xy}^{(0)} \left(1 + \frac{i\omega v}{2r_h \sqrt{w}} \ln u + \mathcal{O}(u) \right).$$
(5.134)

The two-form current can then be written as

$$\langle \delta J^{xy} \rangle = \frac{2\pi^2}{N_c^2} \left(\frac{i\omega v}{r_h \sqrt{w}} \right) \delta b_{xy} + \dots ,$$
 (5.135)

which yields

$$r_{\parallel} = \frac{2\pi^2}{N_c^2} \left(\frac{v}{2r_h\sqrt{w}}\right),\tag{5.136}$$

as stated in Eq. (5.85).

(iii) Bulk viscosities ζ_{\perp} , ζ_{\parallel} and ζ_{\times}

By counting the number of the relevant degrees of freedom, it turns out that there is only one dynamical mode in this decoupled systems coming from $4 \times (2^{\text{nd}}\text{-order ODE for } \delta g_{tt}, \delta g_{aa}, \delta g_{zz}, \delta b_{tz}) - 3 \times (1^{\text{st}}\text{-order ODE for } \delta g_{tu}, \delta g_{uu}, \delta b_{zu})$. To find this dynamical mode, we start by solving the algebraic equation for $\delta g_{tu}, \delta g_{uu}$ and δb_{zu} from the tu and uu components of Einstein's equation combined with the zu component of Maxwell's equations. Plugging these solutions into the four second-order equations involving $\delta g_{tt}, \delta g_{aa}, \delta g_{zz}$ and δb_{tz} , we find that the remaining two non-trivial equations involve only δg_{aa} and δg_{zz} . The single resulting equation of motion can then be expressed in terms of the gauge-invariant variable $Z_s(u)$ defined as

$$Z_s(u) = \delta G^a_{\ a} - \frac{2\mathcal{V}'}{\mathcal{W}'} \delta G^z_{\ z} \,, \tag{5.137}$$

where $\delta g_{aa} = \delta g_{xx} + \delta g_{yy}$. The equation of motion for Z_s can be written

$$Z_s''(u) + C_1(\omega, u) Z_s'(u) + C_2(\omega, u) Z_s(u) = 0, \qquad (5.138)$$

where

$$C_{1} = \frac{3}{2u} + \frac{F'}{F} + \frac{2\mathcal{W}''}{W'} + 2\mathcal{V}' + \mathcal{W}' - 2\left(\frac{2\mathcal{V}'' + \mathcal{W}'}{2\mathcal{V}' + \mathcal{W}'}\right),$$

$$C_{2} = -\frac{b^{2}e^{-4\mathcal{V}}}{3u^{3}F\mathcal{W}'}\left(\frac{F'}{F} + 4\mathcal{W}'\right) + \frac{\omega^{2}}{4r_{h}^{2}u^{3}F^{2}} - \frac{2F'^{2}}{3F^{2}\mathcal{W}'}(\mathcal{V}' - \mathcal{W}') \quad (5.139)$$

$$+ \frac{4\mathcal{V}'F'(\mathcal{V}' - \mathcal{W}')^{2}}{3F\mathcal{W}'(2\mathcal{V}' + \mathcal{W}')} + \frac{8\mathcal{V}'^{2}(\mathcal{V}' + 2\mathcal{W}')(\mathcal{V}' - \mathcal{W}')}{2\mathcal{W}'(2\mathcal{V}' + \mathcal{W}')}.$$

Now, suppose that the time-independent solution for Z_s is $\mathfrak{Z}^{(-)}$, so that $\mathfrak{Z}^{(-)}(u \to 0) = Z^{(0)} \equiv \delta h_{aa} - 2\delta h_{zz}$ (note that $\mathcal{V}'/\mathcal{W}' \to 1$ and $u \to 0$). The second solution, denoted as $\mathfrak{Z}^{(+)}$, contains the time-dependent information and can be found from the Wronskian

$$\mathfrak{Z}^{(+)}(u) = \mathfrak{Z}^{(-)}(u) \int_{u}^{1} du' \, \frac{W_R(u')}{\left(\mathfrak{Z}^{(-)}(u')\right)^2}, \quad W_R = \left(\frac{2\mathcal{V}' + \mathcal{W}'}{\mathcal{W}'}\right)^2 \frac{e^{2\mathcal{V} + \mathcal{W}}}{u^{3/2}F}.$$
(5.140)

We then find that the asymptotic solution for $\mathfrak{Z}^{(+)}$ are

$$\mathfrak{Z}^{(+)} = \frac{9}{2v\sqrt{w}} \left[\mathfrak{Z}^{(-)}(0)\right]^{-1} u^2 + \mathcal{O}(u^3), \tag{5.141}$$

near the boundary u = 0 and

$$\mathfrak{Z}^{(+)} = -9r_h \left(\frac{6+B^2}{6-B^2}\right)^2 \left[2\pi T\mathfrak{Z}^{(-)}(1)\right]^{-1} \ln(1-u) + \mathcal{O}(1-u), \quad (5.142)$$

near the horizon u = 1. As the full solution is a linear combination, $Z_s(u) = \mathfrak{Z}^{(-)} + \alpha \mathfrak{Z}^{(+)}$, the ingoing boundary condition set the frequency-dependent function $\alpha(\omega)$ to be

$$\alpha(\omega) = \frac{i\omega}{2r_h} \left(\frac{(6+B^2)}{3(6-B^2)}\right)^2 \left[\mathfrak{Z}^{(-)}(1)\right]^2, \tag{5.143}$$

which allows us to write the solution for Z_s near the boundary as

$$Z_s = Z^{(0)} \left(1 + \frac{i\omega}{4r_h v\sqrt{w}} \left(\frac{6+B^2}{6-B^2} \right)^2 \left[\frac{\mathfrak{Z}^{(-)}(1)}{\mathfrak{Z}^{(-)}(0)} \right]^2 u^2 \right) + \dots$$
(5.144)

This expression can then be used to compute the bulk viscosities, for which we follow the approach by [313] and their analysis of the Green's function in sound channel. In summary, we first find the expression for $\langle \delta T^{xx} + \delta T^{yy} \rangle$ and $\langle \delta T^{zz} \rangle$ in terms of δh_{tt} , δh_{aa} , δh_{zz} and δb_{tz} , and then relate the near-boundary data of the bulk modes δG_{tt} , δG_{aa} , δG_{zz} and δB_{tz} to those of Z_s . Then, we impose the radial gauge, $\delta G_{u\mu} = 0$ and $\delta B_{u\mu} = 0$ and solve the equations of motion near the boundary. The first-order equations of motion gives the following relations

$$h_{aa}^{(2)} + h_{tt}^{(2)} + h_{zz}^{(2)} + \frac{B^2 h_{aa}^{(0)}}{36v^2} = 0,$$

$$2\left(h_{aa}^{(2)} + h_{zz}^{(2)}\right) + \frac{v_4^b}{v}\left(h_{aa}^{(0)} - 2h_{zz}^{(0)}\right) - f_4^b\left(h_{aa}^{(0)} + h_{zz}^{(0)}\right) = 0.$$
(5.145)

The coefficients $h_{\mu\nu}^{(n)}$ are defined through the near-boundary expansion of the metric fluctuation. By using the second-order dynamical equation and the radial gauge, the solutions are

$$\delta G^{a}_{\ a} = h^{(0)}_{aa} + h^{(2)}_{aa} u^{2} + \frac{h^{(0)}_{aa} B^{2}}{10v^{2}} u^{2} \ln u + \mathcal{O}(\omega^{2}, u^{3}),$$

$$\delta G^{t}_{\ t} = h^{(0)}_{tt} + h^{(2)}_{tt} u^{2} - \frac{h^{(0)}_{aa} B^{2}}{20v^{2}} u^{2} \ln u + \mathcal{O}(\omega^{2}, u^{3}),$$

$$\delta G^{z}_{\ z} = h^{(0)}_{zz} + h^{(2)}_{zz} u^{2} - \frac{h^{(0)}_{aa} B^{2}}{20v^{2}} u^{2} \ln u + \mathcal{O}(\omega^{2}, u^{3}),$$

(5.146)

where $h_{\mu\nu}^{(0)} \equiv \delta h_{\mu\nu}$ is the metric perturbation used throughout the paper. By combining Eqs. (5.145) and (5.146), and using definition of gauge-invariant mode Z_s , we find that

$$Z_s = Z^{(0)} + \frac{Z^{(0)}\omega^2}{6r_h^2} + Z^{(2)}u^2 + \frac{h_{aa}^{(0)}B^2}{5v^2}u^2\ln u + \mathcal{O}(\omega^4), \qquad (5.147)$$

where

$$Z^{(0)} = h^{(0)}_{aa} - 2h^{(0)}_{zz}, \quad Z^{(2)} = -3h^{(2)}_{zz} - \frac{v^b_4}{v}Z^{(0)} + f^b_4\left(h^{(0)}_{aa} + h^{(0)}_{zz}\right).$$
(5.148)

It is most convenient to extract the transport coefficients from $\langle \delta T^{zz} \rangle$:

$$\begin{split} \langle \delta T^{zz} \rangle &= -\frac{N_c^2}{2\pi^2} \frac{r_h^4 e^{2\mathcal{W}}}{w} \left(\frac{1}{2} \sqrt{uF} \left(\delta G_a^{a\prime} + \delta G_t^{t\prime} \right) \right. \\ &+ \left(\frac{3}{2u} + \frac{\sqrt{uF'}}{2F} + 2\sqrt{uF} \mathcal{V}' \right) \delta G_z^z \right) + \dots \\ &= -\frac{N_c^2}{2\pi^2} \left(\frac{i\omega r_h^3}{12v\sqrt{w}} \left(\frac{6+B^2}{6-B^2} \right)^2 \left[\mathfrak{Z}^{(-)}(1)/\mathfrak{Z}^{(-)}(0) \right]^2 \right) \left(\delta h_{aa} - 2h_{zz} \right) + \dots . \end{split}$$

The ellipses denote various time-independent terms, which are irrelevant for computing the first-order transport coefficients of interest. Using the Kubo formula (5.119), we find that

$$\zeta_{\parallel} = \frac{N_c^2}{2\pi^2} \left(\frac{r_h^3}{3v\sqrt{w}} \left(\frac{6+B^2}{6-B^2} \right)^2 \left[\mathfrak{Z}^{(-)}(1)/\mathfrak{Z}^{(-)}(0) \right]^2 \right) = \frac{s}{4\pi} \left(\frac{4}{3} \left(\frac{6+B^2}{6-B^2} \right)^2 \left[\mathfrak{Z}^{(-)}(1)/\mathfrak{Z}^{(-)}(0) \right]^2 \right),$$
(5.149)

and $\zeta_{\times}^{(2)} = -\zeta_{\parallel}/2.$

Similarly, we can extract ζ_{\perp} and $\zeta_{\times}^{(1)}$ from

$$\begin{split} \langle \delta T^{xx} \rangle + \langle \delta T^{yy} \rangle &= -\frac{N_c^2}{2\pi^2} \frac{r_h^4 e^{2\mathcal{V}}}{v} \left[\frac{1}{2} \sqrt{uF} \left(\delta G_a^{a\prime} + \delta G_t^{t\prime} + \delta G_z^{z\prime} \right) \right. \\ &\left. + \left(\frac{3}{2u} + \frac{\sqrt{uF'}}{2F} + \sqrt{uF} (\mathcal{V}' + \mathcal{W}') \right) \delta G_z^z \right] + \dots \end{split}$$

$$= \frac{N_c^2}{2\pi^2} \left(\frac{i\omega r_h^3}{12v\sqrt{w}} \left(\frac{6+B^2}{6-B^2} \right)^2 \left[\mathfrak{Z}^{(-)}(1)/\mathfrak{Z}^{(-)}(0) \right]^2 \right) \left(\delta h_{aa} - 2h_{zz} \right) + \dots$$

This gives $\zeta_{\perp} = \zeta_{\parallel}/4 = -\zeta_{\times}^{(1)}/2$. Hence, we find that $\zeta_{\times}^{(1)} = \zeta_{\times}^{(2)}$, which is the manifestation of the Onsager relation imposed in [47, 278]. This completes the derivation of expressions stated in Eq. (5.85).

As a simple check of our results, in the zero magnetic field limit, one can

show that

$$\zeta_{\parallel} = \lim_{\omega \to 0} \partial_{\omega} G_{TT}^{zz, zz}(\omega, 0) = \frac{4}{3} \lim_{\omega \to 0} \partial_{\omega} G_{TT}^{xy, xy}(\omega, 0) , \qquad (5.150)$$

which implies, as expected, that the bulk viscosity of conformal relativistic hydrodynamics vanishes (see e.g. [51]). For another check, one can write the relation $\zeta_{\parallel} = 2 \zeta_{\perp}$ in the language of two-point functions and obtain the relation $\lim_{\omega \to 0} \left[\partial_{\omega} G_{TT}^{aa,aa}(\omega,0) - \frac{1}{2} \partial_{\omega} G_{TT}^{zz,zz}(\omega,0) \right] = 0$, which is also satisfied by conformal relativistic hydrodynamics.¹⁹ Interestingly, this relations holds for all strengths of the magnetic field in the model studied in this work. One may wonder whether this relation between different bulk viscosities points to a more general property of (strongly interacting) MHD plasmas.

5.8 Dispersion relations of magnetosonic waves

In the magnetosonic channel, the polynomial equation in ω and k, which needs to be solved in order for us to find the dispersion relations $\omega(k)$ is a quartic equations in ω , which can be written in the following form:

$$\operatorname{Det}\left[-i\omega\mathbb{1} + \mathbb{M}\right] = 0, \qquad (5.151)$$

with 1 the 4×4 identity matrix and the non-zero components M_{ij} of the matrix M given by

¹⁹For a neutral relativistic fluid, one can show that $\operatorname{Im}\langle \delta T^{xx} \rangle + \operatorname{Im}\langle \delta T^{yy} \rangle = \omega \left(\frac{\eta}{3} + \zeta\right) \delta h_{aa} + \omega \left(\zeta - \frac{2}{3}\eta\right) \delta h_{zz}$ and $\operatorname{Im}\langle \delta T^{zz} \rangle = \frac{1}{2}\omega \left(\zeta - \frac{2}{3}\eta\right) \delta h_{aa} + \omega \left(\frac{2}{3}\eta + \frac{1}{2}\zeta\right) \delta h_{zz}$. In a conformal fluid with $\zeta = 0$, one find that $\lim_{\omega \to 0} \partial_{\omega} G_{TT}^{aa,aa}(\omega,0) = \lim_{\omega \to 0} \frac{1}{4}\partial_{\omega} G_{TT}^{zz,zz}(\omega,0) = -\eta/3$.

Chapter 5. Magnetohydrodynamic waves in a strongly interacting holographic plasma

$$\begin{split} M_{11} &= r_{\perp}k^{2}\sin^{2}\theta\mathcal{A}_{11} \,, \quad M_{12} = -r_{\perp}k^{2}\mathcal{A}_{12} \,, \qquad M_{13} = ik\sin\theta\mathcal{A}_{13} \,, \\ M_{14} &= ik\frac{s\cos\theta}{\chi_{11}} \, \qquad M_{21} = -r_{\perp}k^{2}\sin^{2}\theta\mathcal{A}_{21} \,, \quad M_{22} = r_{\perp}k^{2}\mathcal{A}_{22} \,, \\ M_{23} &= ik\rho\sin\theta \, \qquad M_{31} = ik\sin\theta\mathcal{A}_{31} \,, \qquad M_{32} = ik\mathcal{A}_{32} \,, \\ M_{33} &= \mathcal{A}_{33}k^{2} \,, \qquad \mathcal{M}_{34} = \eta_{\perp}k^{2} \, \qquad \mathcal{A}_{34}M_{41} = i\frac{k}{T}\cos\theta \,, \\ M_{43} &= \eta_{\perp}k^{2}\mathcal{A}_{43} \,, \qquad M_{44} = k^{2}\mathcal{A}_{44} \,. \end{split}$$

The coefficients A_{ij} are

$$\begin{aligned} \mathcal{A}_{11} &= \frac{1}{2T^{2}\chi_{11}} (\mu + T\chi_{12})(\mu - T\mu_{21}), \\ \mathcal{A}_{12} &= \frac{1}{2T\rho\chi_{11}} (\mu + T\chi_{12})(\mu\cos^{2}\theta + \rho\chi_{22}\sin^{2}\theta), \\ \mathcal{A}_{13} &= \frac{s - \rho\chi_{12}}{\chi_{11}}, \\ \mathcal{A}_{22} &= \frac{1}{2\rho} \left(\mu\cos^{2}\theta + \rho\chi_{22}\sin^{2}\theta \right), \\ \mathcal{A}_{32} &= \frac{2\rho}{(\varepsilon + p)\sin\theta} \mathcal{A}_{22}, \\ \mathcal{A}_{33} &= \left(\frac{\eta_{\parallel}\cos^{2}\theta + (\eta_{\perp} + \zeta_{\perp})\sin^{2}\theta}{\varepsilon + p} \right), \\ \mathcal{A}_{34} &= \frac{\cos\theta\sin\theta}{\varepsilon + p}, \\ \mathcal{A}_{44} &= \frac{2\zeta_{\parallel}\cos^{2}\theta + \eta_{\parallel}\sin^{2}\theta}{sT}. \end{aligned}$$

By computing the determinant in (5.151), the resulting quartic equation is

$$\omega^4 + c_3 \omega^3 + c_2 \omega^2 + c_1 \omega + c_0 = 0, \qquad (5.152)$$

where c_i are functions of thermodynamics quantities, transport coefficients, k and θ . The expressions for c_i in terms of A_{ij} in are

$$\begin{aligned} c_{3} &= ik^{2} \left(\mathcal{A}_{33} + \mathcal{A}_{44} + \mathcal{A}_{22}r_{\perp} + \mathcal{A}_{11}r_{\perp}\sin^{2}\theta \right), \\ c_{2} &= -\frac{k^{2}}{T\chi_{11}} \left(s\cos^{2}\theta + T\chi_{11}\sin\theta(\mathcal{A}_{32}\rho + \mathcal{A}_{13}\mathcal{A}_{31}\sin\theta) \right) \\ &- k^{4} \left[\mathcal{A}_{22}\mathcal{A}_{44}r_{\perp} + \mathcal{A}_{33}(\mathcal{A}_{44} + \mathcal{A}_{22}r_{\perp}) - \mathcal{A}_{34}\eta_{\perp}^{2} \\ &+ r_{\perp} \left(\mathcal{A}_{11}(\mathcal{A}_{33} + \mathcal{A}_{44}) + r_{\perp}\sin^{2}\theta(\mathcal{A}_{11}\mathcal{A}_{22} - \mathcal{A}_{12}\mathcal{A}_{21}) \right) \right], \\ c_{1} &= -i\frac{k^{4}}{T\chi_{11}} \left\{ s(r_{\perp}\mathcal{A}_{22} + \mathcal{A}_{33})\cos^{2}\theta - \eta_{\perp}\cos\theta\sin\theta(sT\mathcal{A}_{31} \\ &+ \chi_{11}\mathcal{A}_{13}\mathcal{A}_{34}) + \chi_{11}T\sin\theta \left[\rho\mathcal{A}_{32}\mathcal{A}_{44} + r_{\perp}\mathcal{A}_{32}\sin^{2}\theta(\mathcal{A}_{13}\mathcal{A}_{21} + \rho\mathcal{A}_{11}) \\ &+ \mathcal{A}_{31}\sin\theta(\mathcal{A}_{13}\mathcal{A}_{44} + r_{\perp}\mathcal{A}_{13}\mathcal{A}_{22} + r_{\perp}\rho\mathcal{A}_{12}) \right] \right\} \\ &- ir_{\perp}k^{6} \left\{ \mathcal{A}_{22}(\mathcal{A}_{33}\mathcal{A}_{44} - \mathcal{A}_{34}\eta_{\perp}^{2}) + \sin^{2}\theta \left[-r_{\perp}\mathcal{A}_{12}\mathcal{A}_{21}(\mathcal{A}_{33} + \mathcal{A}_{44}) \\ &+ \mathcal{A}_{11}\sin^{2}\theta\left(\mathcal{A}_{33}\mathcal{A}_{44} + r_{\perp}\mathcal{A}_{22}\mathcal{A}_{33} + r_{\perp}\mathcal{A}_{22}\mathcal{A}_{44} - \eta_{\perp}^{2}\mathcal{A}_{34} \right) \right] \right\}, \\ c_{0} &= \left(\frac{s\rho\mathcal{A}_{23}\cos^{2}\theta\sin^{2}\theta}{T\chi_{11}} \right) k^{4} + \frac{r_{\perp}k^{6}}{T\chi_{11}} \left\{ s\mathcal{A}_{22}\mathcal{A}_{33}\cos^{2}\theta \\ &+ \chi_{11}\mathcal{A}_{32}\mathcal{A}_{44}(\mathcal{A}_{13}\mathcal{A}_{21} + \rho\mathcal{A}_{11})\sin^{3}\theta + \chi_{11}\mathcal{A}_{13}\mathcal{A}_{31}\mathcal{A}_{44}(\mathcal{A}_{13}\mathcal{A}_{22} + \rho\mathcal{A}_{12}) \\ &+ \eta_{\perp}\cos\theta\sin\theta \left[sT\mathcal{A}_{22}\mathcal{A}_{31} + \chi_{11}\mathcal{A}_{13}\mathcal{A}_{22}\mathcal{A}_{34} \\ &+ \chi_{11}\rho\mathcal{A}_{12}\mathcal{A}_{34} + sT\mathcal{A}_{21}\mathcal{A}_{32}\sin\theta \right] \right\} \\ &+ r_{\perp}^{2}(\mathcal{A}_{12}\mathcal{A}_{21} - \mathcal{A}_{11}\mathcal{A}_{22})(\mathcal{A}_{33}\mathcal{A}_{34}\eta_{\perp}^{2})k^{8}\sin^{2}\theta \,. \end{aligned}$$
(5.153)

In principle, Eq. (5.152) gives a closed-form solution for the four $\omega(k)$. In practice, the solutions are extremely lengthy so it is more convenient to solve it

numerically (our equations of state and transport coefficients are in any case given numerically), or by using various expansions, e.g. small k/T or small $k/\sqrt{\mathcal{B}}$.

More generally, we should study Quantum Field Theory from many points of view because of its many applications (and) in order to understand it better. Hopefully, we can learn how to improve its presentation – reformulate it.

N. Seiberg



Conclusion and outlook

In this thesis, three different classes of strongly interacting quantum field theories with holographic duals, classified by their global symmetries, have been studied. This work has focused on the transport coefficients which govern the low energy excitations of these systems. The results in this thesis contribute to three broad aspects of strongly interacting quantum field theory. Let me point out what they are, what have we learned about them in the course of this thesis and outline possible future research directions.

Possible universal relations of the transport coefficients

Two types of possible universal relations are investigated in chapter 3 and 4 of this thesis. The first relation is the bound on the ratio of the shear viscosity to the entropy (the KSS bound), which is widely believed to hold for a large class of systems. In a theory without translational symmetry, however, it was observed that the supposed "shear viscosity" extracted from the 2-point corre-

lation function $\langle T^{xy}T^{xy}\rangle$ violates this bound. In chapter 3, we show that this quantity is not a shear viscosity due to the following reason. The holographic fluid which breaks translational symmetry is not a conventional relativistic fluid but a *forced fluid*, due to the presence of scalar fields which break translational symmetry. Thus, the constitutive relation and, consequently, the Kubo formula are modified. With the new constitutive relation, we can single out the *true shear viscosity* and show that it still violates the viscosity bound even at very weak momentum relaxation rate.

The second type of universal relations are relations between the anomalous conductivities and the anomaly coefficients. These are considered in chapter 4. We develop an independent proof for a large class of strongly interacting quantum field theories that the relations between the two quantities are exact, without relying on specific details of the Lagrangian. While, there is much evidences suggesting that the relations between anomalous conductivities and anomaly coefficients are exact, including a more conventional field theoretic proof, it is reassuring to find independent demonstration of the non-renomalisation nature of anomalous transports.

The result of chapter 3 is a clear demonstration that the relation between the shear viscosity and the entropy density is not a universal relation in theories without translational symmetry, such as those in condensed matter. It is an interesting question to investigate whether the KSS bound originates from something more fundamental, such as the quantum bound on the rate of entropy production which can still be applied in this scenario. There is a good amount of more recent works such as [42, 314–319] which explored this fundamental aspect and also its experimentally measurable consequences. In this regard, the anomalous transport is a promising area of research, strengthened by recent experimental realisations [320, 321]: the first direct measurements of anomaly coefficients! Since the materials in these experiments are described by weakly coupled QFT, it would be fascinating to see the same effect persist in strongly coupled systems to confirm the non-renormalisation nature of the anomalous conductivities and also to explore the consequences of quantum anomalies in a far-from-equilibrium behaviour.

Decoding hydrodynamic information from gravity

The results of chapter 3-5 can be obtained due to the fact that the transport coefficients are encoded in a theory of gravity. The most studied quantity is, of course, the shear viscosity. In many examples, it can be thought of as the absorption cross section of the low-energy graviton by the black hole. Moreover, it is encoded in the fluid description of the stretch horizon, thanks to the existence of radially conserved currents.

In chapters 4 and 5, we find new radially conserved currents associated to anomalous conductivities and MHD transport coefficients. Although, not all transport coefficients can be extracted from this method (namely the shear viscosity of the forced fluid in chapter 3 and the transverse resistivity and bulk viscosities in chapter 5), we present a simple way to extract these quantities despite the fact that they depend strongly on the bulk of AdS space. These formulae will be beneficial to future research as they greatly simplify the computation of the transport coefficients, thus allowing us to easily explore the landscape of strongly interacting quantum field theories.

From their relation to the radially conserved currents, we can categorise the transport coefficients into two kinds: those that only require the radially conserved currents and those that are not. With this information and the assumption that the radial coordinate of AdS corresponds to the energy scale, it is tempting to conjecture that the second kind of transport coefficients are controlled by specific details of renormalisation group flow while the first kind are encoded in some sort of "zero modes" which are independent of renormalisation group flow trajectories. Consequently, the transport coefficients of the first kind only depend on the IR fixed point and this could potentially lead us to new universal relations.

The fact that different parts of spacetime encode different QFT information is very intriguing. By decoding hydrodynamic data from certain regions of the spacetime, it is tempting to assume that every piece of spacetime contains information about quantum field theory. There are many recent works that try to come up with a unified framework to interpret any given region spacetime
to the information in quantum field theory (see e.g. [322–326]. The approach we take in this thesis, on the other hand, is much simpler, as we only interested in finding regions of spacetime that encode the transport coefficients. Perhaps further investigation in this direction could give us profound insight regarding the precise field theory interpretation of AdS radial coordinate and the field theory data encoded behind the black hole horizon.

This research direction ties together ideas from theory of gravity, renormalisation group and leads to possibility that the gravity is a phenomenon that emerges from quantum field theory. It is exciting that we are able to witness developments that could revolutionised the way we understand QFT.

New symmetry, new hydrrodynamics and new phenomena

On a less other-worldly note, I would like to emphasise the power of the hydrodynamic constructions, particularly in chapter 5. We are able to construct unique effective theories, characterised purely by equations of state and a few transport coefficients, only by considering the global symmetry. This type of effective theory is applicable not only to low-energy dynamics of quantum theories but also to classical theories with the same global symmetries. Thus, the classification of modified hydrodynamics is beneficial not only to the realm of strongly interacting quantum fluids considered in this thesis, but also to the other areas of physics which involve collective behaviour.

Magnetohydrodynamics, studied in chapter 5, is an effective theory of a strongly interacting matter and a dynamical electromagnetic field with the higherform symmetry as its guiding principle While the standard MHD formalism, which treats the matter sector and the electromagnetic field separately, is a very successful theory, it is only valid in the regime where the magnetic field is weak. The formalism in chapter 5 does not distinguish between the two sectors. Thus, it allows us to explore the scenario where both the magnetic field and the interaction strength within the matter sector are strong. We focus on a theory where the matter sector is a strongly interacting quantum field theory with a holographic dual and explore the entire range of the magnetic field strength. By doing this, we find several new qualitative features of waves in MHD as we vary the strength of the magnetic field and the angle between the wave propagation direction and magnetic field lines. Among many observations, we emphasise the role of the dissipative terms in turning the MHD sound modes into new diffusive modes as one varies the propagation direction. This is a phenomena which cannot be observed in an ideal magnetohydrodynamics, which is commonly applied in the phenomenology of plasma. Our results also reveal various qualitative features of MHD waves at strong magnetic field that are vastly different from those at weak field.

The role of holographic duality in this chapter is twofold. First, it can be used to create a state of plasma, which is thermodynamically stable throughout the entire range of the magnetic field strength. Second, the computation of transport coefficients in the gauge/gravity duality framework is much easier compared to the weakly coupled QFT computation. We expect that many quantitative features in chapter 5 also persist in the perturbative regime where the interaction within the matter sector is weak. It is one of our top priorities to confirm this conjecture. This would allow us to make easier contact with experiment (such as nuclear fusion experiment) or astronomical observations of phenomena which are believed to be described by magnetohydrodynamics (see e.g. Fig.6.1).

Last but not least, I would like to point out another promising aspect of global symmetry in hydrodynamics which emerges from attempts to combine dissipative effect with the Wilsonian effective action. It turns out that, as a consequence of fluctuation-dissipation relations, the effective action is invariant under an emergent supersymmetry (see e.g. [329] for early developments and [275, 330] for more recent ones). It would be very interesting to explore consequences of this large global symmetry group on observable effects, holographic duality and strongly interacting quantum field theory.

Let me conclude these adventures in the realm of strongly interacting QFT. (Global) Symmetry is a beautiful concept that guides us through this unknown land. It is a backbone of modern theoretical physics, including everything in this



Figure 6.1. One of the MHD modes, the Alfvén wave, can be thought of as an oscillation mode of the magnetic field lines as depicted in (LEFT). It was postulated to be responsible for heat transfer in the solar corona and was recently observed by Hinode spacecraft in [327]. The images are taken from [327, 328]

thesis. Regardless of how we change our view of QFT in the future, I believe that the global symmetry will always play a central role.

Samenvatting

In dit proefschrift wordt de niet-perturbatieve dualiteit tussen de zwaartekracht en ijktheorie gebruikt om de transporteigenschappen van een grote groep sterkwisselwerkende kwantumsystemen te onderzoeken. Deze methodiek, die ontstaan is vanuit de snaartheorie, stelt in staat om de kwantumveldentheorie te beschrijven via de zwaartekracht. Op deze manier kan een andere parameterruimte worden bestudeerd, die niet noodzakelijkerwijs kan worden onderzocht via conventionele methodes. Het blijkt dat de aangeslagen toestand bij lage energie in dergelijke theorieën kan worden beschreven via een symmetriegeleide, effectieve theorie: de hydrodynamica.

In hoofdstuk 3 construeren we een consistent hydrodynamisch model waar translatiesymmetrie wordt verbroken. Dit wordt tot stand gebracht via de koppeling van de sterk-wisselwerkende materie met externe, positie-afhankelijke scalaire velden. We definiëren nauwkeurig de schuifviscositeit in dit systeem en onderzoeken de geldigheid van de welbekende ondergrens op de ratio van deze viscositeit en de entropiedichtheid, in de hoop deze te kunnen toepassen op systemen in de fysica van gecondenseerde materie waar translatiesymmetrie is verbroken.

In hoofdstuk 4 onderzoeken we het voorgaande systeem met twee aanvullende, globale U(1)-symmetrieën, wanneer het wordt onderworpen aan een zwak extern, niet-dynamisch magnetisch veld. Een Noether-stroom wordt gebroken door een kwantumanomalie, die aanleiding geeft tot een nieuw transportfenomeen, genaamd anomaal transport. We ontdekken dat de geleidbaarheden die dit transport drijven, worden beschreven op de waarnemingshorizon

Samenvatting

van het zwarte gat in de zwaartekracht dualiteitstheorie. Via deze vergelijkingen voor de waarnemingshorizon tonen we aan dat - voor alle theorieën met een holografische, duale beschrijving bij eindige temperatuur - de functionele vorm van de anomale geleidbaarheden wordt vastgesteld door de anomaliecoëfficiënten, ongeacht de waarde van de koppelingsconstante, of de details van het renormalizatieproces.

In hoofdstuk 5 onderzoeken we een sterk-wisselwerkende theorie, gekoppeld met een dynamisch, extern U(1)-ijkveld. Een dergelijke situatie wordt dikwijls aangetroffen in de plasmafysica en magnetohydrodynamica. We gebruiken het nieuwe concept van gegeneraliseerde, globale symmetrieën, welke geassocieerd zijn met snaar-achtige objecten, om de daadwerkelijke globale symmetrie van dit systeem te identificeren. In combinatie met het nieuwe formalisme van de hydrodynamica met deze gegeneraliseerde, globale symmetrie, implementeren we dit nieuwe concept in de ijk/zwaartekracht-dualiteit. Het resultaat is dat we - voor het eerst - de dynamica van een dergelijk sterkwisselwerkend systeem bij lage energie kunnen onderzoeken, voor elke mogelijke waarde van de magnetische veldsterkte.

Bibliography

- http://www.claymath.org/millennium-problems/ yang-mills-and-mass-gap.
- [2] http://www.physics.adelaide.edu.au/theory/ staff/leinweber/VisualQCD/Nobel/.
- [3] J. Zaanen, (2010), arXiv:1012.5461 [cond-mat.supr-con].
- [4] J. Zaanen, Y. Liu, Y. Sun, and K. Schalm, *Holographic Duality in Condensed Matter Physics* (Cambridge University Press, 2015).
- [5] R. Shankar, Physica A Statistical Mechanics and its Applications 177, 530 (1991).
- [6] J. Polchinski, in *Theoretical Advanced Study Institute (Tasi 92): from Black Holes and Strings to Particles Boulder, Colorado, June 3-28, 1992* (1992) pp. 0235–276, arXiv:hep-th/9210046 [hep-th].
- [7] https://web.science.uu.nl/drstp/Postgr. courses/SPTCM/2015/adscmtreview.pdf.
- [8] M. Troyer and U.-J. Wiese, Phys. Rev. Lett. 94, 170201 (2005), arXiv:condmat/0408370 [cond-mat].
- [9] E. Y. Loh, J. E. Gubernatis, R. T. Scalettar, S. R. White, D. J. Scalapino, and R. L. Sugar, Phys. Rev. B 41, 9301 (1990).

- [10] C. Schmidt, Proceedings, 24Th International Symposium on Lattice Field Theory (Lattice 2006): Tucson, Usa, July 23-28, 2006, PoS LAT2006, 021 (2006), arXiv:hep-lat/0610116 [hep-lat].
- [11] J. Zaanen, F. Krüger, J. She, D. Sadri, and S. I. Mukhin, ArXiv e-prints (2008), arXiv:0802.2455.
- [12] http://www.claymath.org/millennium-problems/ p-vs-np-problem.
- [13] A. A. Belavin, A. M. Polyakov, and A. B. Zamolodchikov, Nucl. Phys. B241, 333 (1984).
- [14] R. Rattazzi, V. S. Rychkov, E. Tonni, and A. Vichi, JHEP 12, 031 (2008), arXiv:0807.0004 [hep-th].
- [15] N. Seiberg and E. Witten, Nucl. Phys. B426, 19 (1994), [Erratum: Nucl. Phys.B430,485(1994)], arXiv:hep-th/9407087 [hep-th].
- [16] F. Cachazo, P. Svrcek, and E. Witten, JHEP 09, 006 (2004), arXiv:hepth/0403047 [hep-th].
- [17] R. Britto, F. Cachazo, B. Feng, and E. Witten, Phys. Rev. Lett. 94, 181602 (2005), arXiv:hep-th/0501052 [hep-th].
- [18] N. Arkani-Hamed and J. Trnka, JHEP 10, 030 (2014), arXiv:1312.2007 [hep-th].
- [19] G. 't Hooft, in Salamfest 1993:0284-296 (1993) pp. 0284-296, arXiv:gr-qc/9310026 [gr-qc].
- [20] L. Susskind, J. Math. Phys. **36**, 6377 (1995), arXiv:hep-th/9409089 [hep-th]
- [21] A. Strominger and C. Vafa, Phys. Lett. B379, 99 (1996), arXiv:hepth/9601029 [hep-th].

- [22] J. M. Maldacena, Int. J. Theor. Phys. 38, 1113 (1999), [Adv. Theor. Math. Phys.2,231(1998)], arXiv:hep-th/9711200 [hep-th].
- [23] S. A. Hartnoll, P. K. Kovtun, M. Muller, and S. Sachdev, Phys. Rev. B76, 144502 (2007), arXiv:0706.3215 [cond-mat.str-el].
- [24] J. Crossno, J. K. Shi, K. Wang, X. Liu, A. Harzheim, A. Lucas, S. Sachdev, P. Kim, T. Taniguchi, K. Watanabe, *et al.*, Science **351**, 1058 (2016).
- [25] D. Bandurin, I. Torre, R. K. Kumar, M. B. Shalom, A. Tomadin, A. Principi, G. Auton, E. Khestanova, K. Novoselov, I. Grigorieva, *et al.*, Science **351**, 1055 (2016).
- [26] P. J. Moll, P. Kushwaha, N. Nandi, B. Schmidt, and A. P. Mackenzie, Science 351, 1061 (2016).
- [27] J. Zaanen, Science **351**, 1026 (2016).
- [28] J. Zaanen, Nature 430, 512 (2004).
- [29] S. Sachdev, *Quantum phase transitions* (Wiley Online Library, 2007).
- [30] P. Kovtun, D. T. Son, and A. O. Starinets, Phys. Rev. Lett. 94, 111601 (2005), arXiv:hep-th/0405231 [hep-th].
- [31] A. Adams, L. D. Carr, T. Schäfer, P. Steinberg, and J. E. Thomas, New J. Phys. 14, 115009 (2012), arXiv:1205.5180 [hep-th].
- [32] A. Buchel, Phys. Lett. B663, 286 (2008), arXiv:0708.3459 [hep-th].
- [33] M. Haack and A. Yarom, Nucl. Phys. B813, 140 (2009), arXiv:0811.1794 [hep-th].
- [34] A. Cherman, T. D. Cohen, and A. Nellore, Phys. Rev. D80, 066003 (2009), arXiv:0905.0903 [hep-th].
- [35] P. Kovtun and A. Ritz, Phys. Rev. D78, 066009 (2008), arXiv:0806.0110[hep-th].

- [36] S. A. Hartnoll, Nature Phys. 11, 54 (2015), arXiv:1405.3651 [cond-mat.strel].
- [37] J. Maldacena, S. H. Shenker, and D. Stanford, JHEP 08, 106 (2016), arXiv:1503.01409 [hep-th].
- [38] S. Grozdanov, A. Lucas, S. Sachdev, and K. Schalm, Phys. Rev. Lett. 115, 221601 (2015), arXiv:1507.00003 [hep-th].
- [39] S. Grozdanov, A. Lucas, and K. Schalm, Phys. Rev. D93, 061901 (2016), arXiv:1511.05970 [hep-th].
- [40] S. A. Hartnoll, D. M. Ramirez, and J. E. Santos, JHEP 03, 170 (2016), arXiv:1601.02757 [hep-th].
- [41] P. Kleinert and J. Probst, JHEP 12, 091 (2016), arXiv:1610.01081 [hep-th].
- [42] A. Lucas and S. A. Hartnoll, (2017), arXiv:1704.07384 [cond-mat.str-el].
- [43] P. Burikham and N. Poovuttikul, Phys. Rev. D94, 106001 (2016), arXiv:1601.04624 [hep-th].
- [44] S. Grozdanov and N. Poovuttikul, JHEP 09, 046 (2016), arXiv:1603.08770[hep-th].
- [45] S. Grozdanov and N. Poovuttikul, To appear .
- [46] D. Gaiotto, A. Kapustin, N. Seiberg, and B. Willett, JHEP 02, 172 (2015), arXiv:1412.5148 [hep-th].
- [47] S. Grozdanov, D. M. Hofman, and N. Iqbal, Phys. Rev. D95, 096003 (2017), arXiv:1610.07392 [hep-th].
- [48] L. Landau and E. Lifshitz, *Fluid Mechanics*, v. 6 (Elsevier Science, 2013).
- [49] L. Rezzolla and O. Zanotti, *Relativistic Hydrodynamics*, EBSCO ebook academic collection (OUP Oxford, 2013).

- [50] R. Baier, P. Romatschke, D. T. Son, A. O. Starinets, and M. A. Stephanov, JHEP 04, 100 (2008), arXiv:0712.2451 [hep-th].
- [51] P. Kovtun, Int Summer School on Applications of String Theory Seattle, Washington, Usa, July 18-29, 2011, J. Phys. A45, 473001 (2012), arXiv:1205.5040 [hep-th].
- [52] O. Aharony, S. S. Gubser, J. M. Maldacena, H. Ooguri, and Y. Oz, Phys. Rept. 323, 183 (2000), arXiv:hep-th/9905111 [hep-th].
- [53] E. D'Hoker and D. Z. Freedman, in *Strings, Branes and Extra Dimensions: Tasi 2001: Proceedings* (2002) pp. 3–158, arXiv:hep-th/0201253 [hep-th].
- [54] S. A. Hartnoll, Strings, Supergravity and Gauge Theories. Proceedings, Cern Winter School, Cern, Geneva, Switzerland, February 9-13 2009, Class.
 Quant. Grav. 26, 224002 (2009), arXiv:0903.3246 [hep-th].
- [55] J. McGreevy, Adv. High Energy Phys. 2010, 723105 (2010), arXiv:0909.0518 [hep-th].
- [56] S. Sachdev, Ann. Rev. Condensed Matter Phys. 3, 9 (2012), arXiv:1108.1197 [cond-mat.str-el].
- [57] S. A. Hartnoll, A. Lucas, and S. Sachdev, (2016), arXiv:1612.07324 [hepth].
- [58] J. Casalderrey-Solana, H. Liu, D. Mateos, K. Rajagopal, and U. Wiedemann, *Gauge/String Duality, Hot QCD and Heavy Ion Collisions* (Cambridge University Press, 2014).
- [59] M. Ammon and J. Erdmenger, Gauge/gravity Duality: Foundations and Applications (Cambridge University Press, 2015).
- [60] H. Năstase, Introduction to the AdS/CFT Correspondence (Cambridge University Press, 2015).
- [61] K. Jensen, M. Kaminski, P. Kovtun, R. Meyer, A. Ritz, and A. Yarom, Phys. Rev. Lett. 109, 101601 (2012), arXiv:1203.3556 [hep-th].

- [62] N. Banerjee, J. Bhattacharya, S. Bhattacharyya, S. Jain, S. Minwalla, and T. Sharma, JHEP 09, 046 (2012), arXiv:1203.3544 [hep-th].
- [63] P. Glorioso and H. Liu, (2016), arXiv:1612.07705 [hep-th].
- [64] P. Romatschke, Class. Quant. Grav. 27, 025006 (2010), arXiv:0906.4787 [hep-th].
- [65] R. A. Davison and B. Goutéraux, JHEP 01, 039 (2015), arXiv:1411.1062[hep-th].
- [66] R. A. Davison, L. V. Delacrétaz, B. Goutéraux, and S. A. Hartnoll, Phys. Rev. B94, 054502 (2016), arXiv:1602.08171 [cond-mat.supr-con].
- [67] L. V. Delacrétaz, B. Goutéraux, S. A. Hartnoll, and A. Karlsson, (2016), arXiv:1612.04381 [cond-mat.str-el].
- [68] L. V. Delacrétaz, B. Goutéraux, S. A. Hartnoll, and A. Karlsson, (2017), arXiv:1702.05104 [cond-mat.str-el].
- [69] A. Lucas, New J. Phys. 17, 113007 (2015), arXiv:1506.02662 [hep-th].
- [70] A. Lucas, J. Crossno, K. C. Fong, P. Kim, and S. Sachdev, Phys. Rev. B93, 075426 (2016), arXiv:1510.01738 [cond-mat.str-el].
- [71] V. Scopelliti, K. Schalm, and A. Lucas, (2017), arXiv:1705.04325 [condmat.str-el].
- [72] R. A. Bertlmann, Anomalies in Quantum Field Theory (1996).
- [73] J. A. Harvey (2005) arXiv:hep-th/0509097 [hep-th].
- [74] A. Bilal, (2008), arXiv:0802.0634 [hep-th].
- [75] M. Nakahara, *Geometry, Topology and Physics, Second Edition*, Graduate student series in physics (Taylor & Francis, 2003).
- [76] S. L. Adler, Phys. Rev. 177, 2426 (1969).

- [77] J. S. Bell and R. Jackiw, Nuovo Cim. A60, 47 (1969).
- [78] S. L. Adler and W. A. Bardeen, Phys. Rev. 182, 1517 (1969).
- [79] L. Álvarez-Gaumé and P. H. Ginsparg, Annals Phys. 161, 423 (1985), [Erratum: Annals Phys.171,233(1986)].
- [80] K. Fukushima, D. E. Kharzeev, and H. J. Warringa, Phys. Rev. D78, 074033 (2008), arXiv:0808.3382 [hep-ph].
- [81] A. Y. Alekseev, V. V. Cheianov, and J. Fröhlich, Phys. Rev. Lett. 81, 3503 (1998).
- [82] M. A. Metlitski and A. R. Zhitnitsky, Phys. Rev. D72, 045011 (2005), arXiv:hep-ph/0505072 [hep-ph].
- [83] G. M. Newman and D. T. Son, Phys. Rev. D73, 045006 (2006), arXiv:hepph/0510049 [hep-ph].
- [84] D. E. Kharzeev and H. J. Warringa, Phys. Rev. D80, 034028 (2009), arXiv:0907.5007 [hep-ph].
- [85] K. Jensen, Phys. Rev. D85, 125017 (2012), arXiv:1203.3599 [hep-th].
- [86] R. Loganayagam and P. Surowka, JHEP 04, 097 (2012), arXiv:1201.2812 [hep-th].
- [87] P. V. Buividovich, Nucl. Phys. **A925**, 218 (2014), arXiv:1312.1843 [hep-lat]
- [88] U. Gürsoy and A. Jansen, JHEP **1410**, 92 (2014), arXiv:1407.3282 [hep-th]
- [89] K. Landsteiner, Proceedings, 56Th Cracow School of Theoretical Physics : a Panorama of Holography: Zakopane, Poland, May 24-June 1, 2016, Acta Phys. Polon. B47, 2617 (2016), arXiv:1610.04413 [hep-th].
- [90] M. Kalb and P. Ramond, Phys. Rev. **D9**, 2273 (1974).

- [91] Z. Nussinov and G. Ortiz, Annals Phys. 324, 977 (2009).
- [92] A. Kapustin and R. Thorngren, (2013), arXiv:1309.4721 [hep-th].
- [93] B. Horn, A. Nicolis, and R. Penco, JHEP 10, 153 (2015), arXiv:1507.05635 [hep-th].
- [94] A. Esposito, R. Krichevsky, and A. Nicolis, (2017), arXiv:1704.08267 [hep-th] .
- [95] A. Nicolis and R. Penco, (2017), arXiv:1705.08914 [hep-th].
- [96] A. J. Beekman, J. Nissinen, K. Wu, and J. Zaanen, (2017), arXiv:1703.03157 [cond-mat.str-el].
- [97] H. Kleinert, Path Integrals in Quantum Mechanics, Statistics, Polymer Physics, and Financial Markets, EBL-Schweitzer (World Scientific, 2009).
- [98] K.-c. Chou, Z.-b. Su, B.-l. Hao, and L. Yu, Phys. Rept. 118, 1 (1985).
- [99] G. D. Moore and K. A. Sohrabi, Phys. Rev. Lett. 106, 122302 (2011), arXiv:1007.5333 [hep-ph].
- [100] S. S. Gubser, I. R. Klebanov, and A. M. Polyakov, Phys. Lett. B428, 105 (1998), arXiv:hep-th/9802109 [hep-th].
- [101] E. Witten, Adv. Theor. Math. Phys. 2, 253 (1998), arXiv:hep-th/9802150[hep-th].
- [102] S. de Haro, S. N. Solodukhin, and K. Skenderis, Commun. Math. Phys.
 217, 595 (2001), arXiv:hep-th/0002230 [hep-th].
- [103] G. Policastro, D. T. Son, and A. O. Starinets, Phys. Rev. Lett. 87, 081601 (2001), arXiv:hep-th/0104066 [hep-th].
- [104] G. Policastro, D. T. Son, and A. O. Starinets, JHEP 09, 043 (2002), arXiv:hep-th/0205052 [hep-th].

- [105] G. Policastro, D. T. Son, and A. O. Starinets, JHEP 12, 054 (2002), arXiv:hep-th/0210220 [hep-th].
- [106] C. P. Herzog, JHEP 12, 026 (2002), arXiv:hep-th/0210126 [hep-th].
- [107] A. Buchel, Nucl. Phys. **B708**, 451 (2005), arXiv:hep-th/0406200 [hep-th].
- [108] P. Benincasa, A. Buchel, and A. O. Starinets, Nucl. Phys. B733, 160 (2006), arXiv:hep-th/0507026 [hep-th].
- [109] A. Parnachev and A. Starinets, JHEP 10, 027 (2005), arXiv:hep-th/0506144[hep-th].
- [110] S. Bhattacharyya, V. E. Hubeny, S. Minwalla, and M. Rangamani, JHEP
 02, 045 (2008), arXiv:0712.2456 [hep-th].
- [111] S. Bhattacharyya, R. Loganayagam, I. Mandal, S. Minwalla, and A. Sharma, JHEP 12, 116 (2008), arXiv:0809.4272 [hep-th].
- [112] N. Banerjee, J. Bhattacharya, S. Bhattacharyya, S. Dutta, R. Loganayagam, and P. Surowka, JHEP 01, 094 (2011), arXiv:0809.2596 [hepth].
- [113] J. Erdmenger, M. Haack, M. Kaminski, and A. Yarom, JHEP 01, 055 (2009), arXiv:0809.2488 [hep-th].
- [114] L. Susskind and E. Witten, (1998), arXiv:hep-th/9805114 [hep-th].
- [115] A. W. Peet and J. Polchinski, Phys. Rev. D59, 065011 (1999), arXiv:hepth/9809022 [hep-th].
- [116] T. Faulkner, H. Liu, and M. Rangamani, JHEP 08, 051 (2011), arXiv:1010.4036 [hep-th].
- [117] I. Heemskerk and J. Polchinski, JHEP 06, 031 (2011), arXiv:1010.1264[hep-th].
- [118] S. Grozdanov, JHEP 06, 079 (2012), arXiv:1112.3356 [hep-th].

- [119] D. Z. Freedman, S. S. Gubser, K. Pilch, and N. P. Warner, Adv. Theor. Math. Phys. 3, 363 (1999), arXiv:hep-th/9904017 [hep-th].
- [120] K. Skenderis and P. K. Townsend, Phys. Lett. B468, 46 (1999), arXiv:hepth/9909070 [hep-th].
- [121] C. Charmousis, B. Gouteraux, B. S. Kim, E. Kiritsis, and R. Meyer, JHEP
 11, 151 (2010), arXiv:1005.4690 [hep-th].
- [122] B. Gouteraux and E. Kiritsis, JHEP 12, 036 (2011), arXiv:1107.2116 [hepth].
- [123] R. Meyer, B. Gouteraux, and B. S. Kim, String Theory 2010. Proceedings, 16Th European Workshop, Madrid, Spain, June 14-18, 2010, Fortsch. Phys. 59, 741 (2011), arXiv:1102.4433 [hep-th].
- [124] M. Blake and A. Donos, Phys. Rev. Lett. 114, 021601 (2015), arXiv:1406.1659 [hep-th].
- [125] K. S. Thorne, R. H. Price, and D. A. Macdonald, eds., *Black Holes: the Membrane Paradigm* (1986).
- [126] N. Iqbal and H. Liu, Phys. Rev. D79, 025023 (2009), arXiv:0809.3808 [hepth].
- [127] D. Nickel and D. T. Son, New J. Phys. 13, 075010 (2011), arXiv:1009.3094 [hep-th].
- [128] I. Bredberg, C. Keeler, V. Lysov, and A. Strominger, JHEP 03, 141 (2011), arXiv:1006.1902 [hep-th].
- [129] E. Banks, A. Donos, and J. P. Gauntlett, JHEP 10, 103 (2015), arXiv:1507.00234 [hep-th].
- [130] E. Banks, A. Donos, J. P. Gauntlett, T. Griffin, and L. Melgar, Class. Quant. Grav. 34, 045001 (2017), arXiv:1609.08912 [hep-th].

- [131] F. Denef, S. A. Hartnoll, and S. Sachdev, Phys. Rev. D80, 126016 (2009), arXiv:0908.1788 [hep-th].
- [132] F. Denef, S. A. Hartnoll, and S. Sachdev, Class. Quant. Grav. 27, 125001 (2010), arXiv:0908.2657 [hep-th].
- [133] S. Caron-Huot and O. Saremi, JHEP 11, 013 (2010), arXiv:0909.4525 [hepth].
- [134] P. Arnold, P. Szepietowski, and D. Vaman, JHEP 07, 032 (2016), arXiv:1603.08994 [hep-th].
- [135] M. T. Grisaru and D. Zanon, Phys. Lett. **B177**, 347 (1986).
- [136] M. D. Freeman, C. N. Pope, M. F. Sohnius, and K. S. Stelle, Phys. Lett. B178, 199 (1986).
- [137] D. J. Gross and E. Witten, Nucl. Phys. B277, 1 (1986).
- [138] J. Pawelczyk and S. Theisen, JHEP 09, 010 (1998), arXiv:hep-th/9808126[hep-th].
- [139] S. S. Gubser, I. R. Klebanov, and A. A. Tseytlin, Nucl. Phys. B534, 202 (1998), arXiv:hep-th/9805156 [hep-th].
- [140] A. Buchel, J. T. Liu, and A. O. Starinets, Nucl. Phys. B707, 56 (2005), arXiv:hep-th/0406264 [hep-th].
- [141] A. Buchel, R. C. Myers, M. F. Paulos, and A. Sinha, Phys.Lett. B669, 364 (2008), arXiv:0808.1837 [hep-th].
- [142] J. de Boer, M. Kulaxizi, and A. Parnachev, JHEP 1003, 087 (2010), arXiv:0910.5347 [hep-th].
- [143] A. Buchel, J. Escobedo, R. C. Myers, M. F. Paulos, A. Sinha, et al., JHEP 1003, 111 (2010), arXiv:0911.4257 [hep-th].
- [144] X. O. Camanho and J. D. Edelstein, JHEP 1006, 099 (2010), arXiv:0912.1944 [hep-th].

- [145] K. Skenderis, M. Taylor, and B. C. van Rees, JHEP 09, 045 (2009), arXiv:0906.4926 [hep-th].
- [146] X. O. Camanho, J. D. Edelstein, J. Maldacena, and A. Zhiboedov, (2014), arXiv:1407.5597 [hep-th].
- [147] R. C. Myers, S. Sachdev, and A. Singh, Phys. Rev. D83, 066017 (2011), arXiv:1010.0443 [hep-th].
- [148] P. M. Hohler and M. A. Stephanov, Phys. Rev. D80, 066002 (2009), arXiv:0905.0900 [hep-th].
- [149] S. Grozdanov and A. O. Starinets, JHEP 03, 007 (2015), arXiv:1412.5685 [hep-th].
- [150] F. M. Haehl, R. Loganayagam, and M. Rangamani, JHEP 05, 060 (2015), arXiv:1502.00636 [hep-th].
- [151] S. A. Hartnoll and C. P. Herzog, Phys. Rev. D77, 106009 (2008), arXiv:0801.1693 [hep-th].
- [152] M. Müller, J. Schmalian, and L. Fritz, Physical review letters 103, 025301 (2009).
- [153] G. T. Horowitz, J. E. Santos, and D. Tong, JHEP 07, 168 (2012), arXiv:1204.0519 [hep-th].
- [154] P. Chesler, A. Lucas, and S. Sachdev, Phys. Rev. D89, 026005 (2014), arXiv:1308.0329 [hep-th].
- [155] Y. Ling, C. Niu, J.-P. Wu, and Z.-Y. Xian, JHEP 11, 006 (2013), arXiv:1309.4580 [hep-th].
- [156] M. Rangamani, M. Rozali, and D. Smyth, JHEP 07, 024 (2015), arXiv:1505.05171 [hep-th].
- [157] A. Donos and J. P. Gauntlett, JHEP 01, 035 (2015), arXiv:1409.6875 [hepth].

- [158] A. Donos and J. P. Gauntlett, JHEP 06, 007 (2014), arXiv:1401.5077 [hepth].
- [159] D. K. O'Keeffe and A. W. Peet, Phys. Rev. D92, 046004 (2015), arXiv:1504.03288 [hep-th].
- [160] A. Lucas and S. Sachdev, Nucl. Phys. B892, 239 (2015), arXiv:1411.3331 [hep-th].
- [161] A. Lucas, S. Sachdev, and K. Schalm, Phys. Rev. D89, 066018 (2014), arXiv:1401.7993 [hep-th].
- [162] M. Blake, D. Tong, and D. Vegh, Phys. Rev. Lett. 112, 071602 (2014), arXiv:1310.3832 [hep-th].
- [163] D. Vegh, (2013), arXiv:1301.0537 [hep-th].
- [164] R. A. Davison, K. Schalm, and J. Zaanen, Phys. Rev. B89, 245116 (2014), arXiv:1311.2451 [hep-th].
- [165] K.-Y. Kim, K. K. Kim, Y. Seo, and S.-J. Sin, JHEP 12, 170 (2014), arXiv:1409.8346 [hep-th].
- [166] B. Goutéraux, JHEP 04, 181 (2014), arXiv:1401.5436 [hep-th].
- [167] A. Amoretti and D. Musso, JHEP 09, 094 (2015), arXiv:1502.02631 [hepth].
- [168] A. Donos and J. P. Gauntlett, JHEP 04, 040 (2014), arXiv:1311.3292 [hepth].
- [169] A. Amoretti, A. Braggio, N. Maggiore, N. Magnoli, and D. Musso, Phys. Rev. D91, 025002 (2015), arXiv:1407.0306 [hep-th].
- [170] A. Amoretti, A. Braggio, N. Maggiore, N. Magnoli, and D. Musso, JHEP
 09, 160 (2014), arXiv:1406.4134 [hep-th].
- [171] M. Blake and D. Tong, Phys. Rev. D88, 106004 (2013), arXiv:1308.4970[hep-th].

- [172] R. A. Davison and B. Goutéraux, JHEP 09, 090 (2015), arXiv:1505.05092 [hep-th].
- [173] L. Alberte, M. Baggioli, A. Khmelnitsky, and O. Pujolas, (2015), arXiv:1510.09089 [hep-th].
- [174] S. A. Hartnoll, D. M. Ramirez, and J. E. Santos, JHEP 04, 022 (2016), arXiv:1508.04435 [hep-th].
- [175] A. Donos and J. P. Gauntlett, Phys. Rev. D92, 121901 (2015), arXiv:1506.01360 [hep-th].
- [176] A. Donos, J. P. Gauntlett, T. Griffin, and L. Melgar, (2015), arXiv:1511.00713 [hep-th].
- [177] M. Blake, JHEP 09, 010 (2015), arXiv:1505.06992 [hep-th].
- [178] M. Blake, JHEP 10, 078 (2015), arXiv:1507.04870 [hep-th].
- [179] Y. Bardoux, M. M. Caldarelli, and C. Charmousis, JHEP 05, 054 (2012), arXiv:1202.4458 [hep-th].
- [180] T. Andrade and B. Withers, JHEP 05, 101 (2014), arXiv:1311.5157 [hep-th]
- [181] M. Baggioli and O. Pujolas, Phys. Rev. Lett. 114, 251602 (2015), arXiv:1411.1003 [hep-th].
- [182] S. Bhattacharyya, R. Loganayagam, S. Minwalla, S. Nampuri, S. P. Trivedi, and S. R. Wadia, JHEP 02, 018 (2009), arXiv:0806.0006 [hep-th].
- [183] M. Rangamani, Strings, Supergravity and Gauge Theories. Proceedings, Cern Winter School, Cern, Geneva, Switzerland, February 9-13 2009, Class.
 Quant. Grav. 26, 224003 (2009), arXiv:0905.4352 [hep-th].
- [184] S. Grozdanov and N. Kaplis, Phys. Rev. D93, 066012 (2016), arXiv:1507.02461 [hep-th].

- [185] L. Alberte, M. Baggioli, and O. Pujolas, (2016), arXiv:1601.03384 [hep-th]
- [186] S. Bhattacharyya, JHEP 07, 104 (2012), arXiv:1201.4654 [hep-th].
- [187] R. Loganayagam, JHEP 05, 087 (2008), arXiv:0801.3701 [hep-th].
- [188] P. Arnold, D. Vaman, C. Wu, and W. Xiao, JHEP 10, 033 (2011), arXiv:1105.4645 [hep-th].
- [189] S. Weinberg, The Quantum Theory of Fields. Vol. 2: Modern Applications (Cambridge University Press, 2013).
- [190] G. 't Hooft, Cargese Summer Institute: Recent Developments in Gauge Theories Cargese, France, August 26-September 8, 1979, NATO Sci. Ser. B 59, 135 (1980).
- [191] V. P. Nair, R. Ray, and S. Roy, Phys. Rev. D86, 025012 (2012), arXiv:1112.4022 [hep-th].
- [192] A. V. Sadofyev and M. V. Isachenkov, Phys. Lett. B697, 404 (2011), arXiv:1010.1550 [hep-th].
- [193] A. V. Sadofyev, V. I. Shevchenko, and V. I. Zakharov, Phys. Rev. D83, 105025 (2011), arXiv:1012.1958 [hep-th].
- [194] K. Jensen, P. Kovtun, and A. Ritz, JHEP 10, 186 (2013), arXiv:1307.3234 [hep-th].
- [195] D. T. Son and P. Surowka, Phys. Rev. Lett. 103, 191601 (2009), arXiv:0906.5044 [hep-th].
- [196] Y. Neiman and Y. Oz, JHEP 03, 023 (2011), arXiv:1011.5107 [hep-th].
- [197] J. Polonyi, Annals Phys. 342, 239 (2014), arXiv:1302.3864 [hep-th].
- [198] J. Polonyi, Symmetry 8, 25 (2016), arXiv:1503.08500 [cond-mat.statmech].

- [199] K. Jensen, R. Loganayagam, and A. Yarom, JHEP 02, 088 (2013), arXiv:1207.5824 [hep-th].
- [200] D. Kharzeev and A. Zhitnitsky, Nucl. Phys. A797, 67 (2007), arXiv:0706.1026 [hep-ph].
- [201] D. E. Kharzeev and D. T. Son, Phys. Rev. Lett. 106, 062301 (2011), arXiv:1010.0038 [hep-ph].
- [202] R. Loganayagam, JHEP 11, 205 (2013), arXiv:1211.3850 [hep-th].
- [203] K. Jensen, R. Loganayagam, and A. Yarom, JHEP 05, 110 (2014), arXiv:1311.2935 [hep-th].
- [204] T. Azeyanagi, R. Loganayagam, G. S. Ng, and M. J. Rodriguez, JHEP 08, 040 (2014), arXiv:1311.2940 [hep-th].
- [205] M. Torabian and H.-U. Yee, JHEP 08, 020 (2009), arXiv:0903.4894 [hep-th]
- [206] K. Landsteiner, E. Megias, L. Melgar, and F. Pena-Benitez, JHEP 09, 121 (2011), arXiv:1107.0368 [hep-th].
- [207] K. Landsteiner, E. Megias, and F. Pena-Benitez, Phys. Rev. Lett. 107, 021601 (2011), arXiv:1103.5006 [hep-ph].
- [208] S. Golkar and D. T. Son, JHEP 02, 169 (2015), arXiv:1207.5806 [hep-th].
- [209] D.-F. Hou, H. Liu, and H.-c. Ren, Phys. Rev. D86, 121703 (2012), arXiv:1210.0969 [hep-th].
- [210] E. V. Gorbar, V. A. Miransky, I. A. Shovkovy, and X. Wang, Phys. Rev. D88, 025025 (2013), arXiv:1304.4606 [hep-ph].
- [211] K. Huang, Y. Koike, and J. Polonyi, Int. J. Mod. Phys. A6, 1267 (1991).
- [212] A. Yamamoto, Phys. Rev. D84, 114504 (2011), arXiv:1111.4681 [hep-lat].

- [213] A. Yamamoto, Phys. Rev. Lett. 107, 031601 (2011), arXiv:1105.0385 [heplat].
- [214] K. Fukushima and M. Ruggieri, Phys. Rev. D82, 054001 (2010), arXiv:1004.2769 [hep-ph].
- [215] A. Gynther, K. Landsteiner, F. Pena-Benitez, and A. Rebhan, JHEP 1102, 110 (2011), arXiv:1005.2587 [hep-th].
- [216] I. Amado, K. Landsteiner, and F. Pena-Benitez, JHEP 05, 081 (2011), arXiv:1102.4577 [hep-th].
- [217] U. Gursoy and J. Tarrio, JHEP 10, 058 (2015), arXiv:1410.1306 [hep-th].
- [218] M. T. Grisaru, A. E. M. van de Ven, and D. Zanon, Phys. Lett. B173, 423 (1986).
- [219] D. J. Gross and J. H. Sloan, Nucl. Phys. B291, 41 (1987).
- [220] Y. Kats and P. Petrov, JHEP 0901, 044 (2009), arXiv:0712.0743 [hep-th].
- [221] M. Brigante, H. Liu, R. C. Myers, S. Shenker, and S. Yaida, Phys.Rev. D77, 126006 (2008), arXiv:0712.0805 [hep-th].
- [222] A. Buchel, Phys. Lett. B665, 298 (2008), arXiv:0804.3161 [hep-th].
- [223] R. C. Myers, M. F. Paulos, and A. Sinha, JHEP 06, 006 (2009), arXiv:0903.2834 [hep-th].
- [224] E. Shaverin and A. Yarom, JHEP 04, 013 (2013), arXiv:1211.1979 [hep-th]
- [225] S. Grozdanov and A. O. Starinets, Theor. Math. Phys. 182, 61 (2015), [Teor. Mat. Fiz.182,no.1,76(2014)].
- [226] E. Shaverin, (2015), arXiv:1509.05418 [hep-th].
- [227] M. Ostrogradsky, (Mem. Ac. St. Petersbourg VI 4 (1850) 385).

- [228] R. P. Woodard, Scholarpedia **10**, 32243 (2015), arXiv:1506.02210 [hep-th]
- [229] K. Landsteiner, E. Megias, and F. Pena-Benitez, Workshop on QCD in Strong Magnetic Fields Trento, Italy, November 12-16, 2012, Lect. Notes Phys. 871, 433 (2013), arXiv:1207.5808 [hep-th].
- [230] D. Anninos and G. Pastras, JHEP **07**, 030 (2009), arXiv:0807.3478 [hep-th]
- [231] Y. Kats, L. Motl, and M. Padi, JHEP 0712, 068 (2007), arXiv:hepth/0606100 [hep-th].
- [232] S. Grozdanov and A. O. Starinets, JHEP 03, 166 (2017), arXiv:1611.07053 [hep-th].
- [233] C. Csaki, H. Ooguri, Y. Oz, and J. Terning, JHEP 01, 017 (1999), arXiv:hepth/9806021 [hep-th].
- [234] A. Karch, E. Katz, D. T. Son, and M. A. Stephanov, Phys. Rev. D74, 015005 (2006), arXiv:hep-ph/0602229 [hep-ph].
- [235] U. Gursoy, E. Kiritsis, and F. Nitti, JHEP 02, 019 (2008), arXiv:0707.1349 [hep-th].
- [236] B. Batell and T. Gherghetta, Phys. Rev. D78, 026002 (2008), arXiv:0801.4383 [hep-ph].
- [237] B. Batell, T. Gherghetta, and D. Sword, Phys. Rev. D78, 116011 (2008), arXiv:0808.3977 [hep-ph].
- [238] C. P. Herzog, Phys. Rev. Lett. 98, 091601 (2007), arXiv:hep-th/0608151[hep-th].
- [239] K. Copsey and R. Mann, JHEP 04, 079 (2013), arXiv:1210.1231 [hep-th].
- [240] S. S. Gubser and F. D. Rocha, Phys. Rev. D81, 046001 (2010), arXiv:0911.2898 [hep-th].

- [241] L. Huijse, S. Sachdev, and B. Swingle, Phys. Rev. B85, 035121 (2012), arXiv:1112.0573 [cond-mat.str-el].
- [242] M. Cvetič, M. J. Duff, P. Hoxha, J. T. Liu, H. Lu, J. X. Lu, R. Martínez-Acosta, C. N. Pope, H. Sati, and T. A. Tran, Nucl. Phys. B558, 96 (1999), arXiv:hep-th/9903214 [hep-th].
- [243] I. R. Klebanov, P. Ouyang, and E. Witten, Phys. Rev. D65, 105007 (2002), arXiv:hep-th/0202056 [hep-th].
- [244] C. Hoyos, T. Nishioka, and A. O'Bannon, JHEP 1110, 084 (2011), arXiv:1106.4030 [hep-th].
- [245] S. S. Gubser, Adv. Theor. Math. Phys. 4, 679 (2000), arXiv:hep-th/0002160 [hep-th].
- [246] R. Casero, E. Kiritsis, and A. Paredes, Nucl. Phys. B787, 98 (2007), arXiv:hep-th/0702155 [HEP-TH].
- [247] A. Jimenez-Alba, K. Landsteiner, and L. Melgar, Phys. Rev. D90, 126004 (2014), arXiv:1407.8162 [hep-th].
- [248] A. Donos and J. P. Gauntlett, JHEP 11, 081 (2014), arXiv:1406.4742 [hepth].
- [249] S. Jain, N. Kundu, K. Sen, A. Sinha, and S. P. Trivedi, JHEP 01, 005 (2015), arXiv:1406.4874 [hep-th].
- [250] S. Jain, R. Samanta, and S. P. Trivedi, (2015), arXiv:1506.01899 [hep-th].
- [251] K. Jensen, M. Kaminski, P. Kovtun, R. Meyer, A. Ritz, and A. Yarom, JHEP 05, 102 (2012), arXiv:1112.4498 [hep-th].
- [252] J. E. Avron, R. Seiler, and P. G. Zograf, Phys. Rev. Lett. 75, 697 (1995).
- [253] J. E. Avron, ArXiv Physics e-prints (1997), physics/9712050.
- [254] C. Hoyos, Int. J. Mod. Phys. B28, 1430007 (2014), arXiv:1403.4739 [condmat.mes-hall].

- [255] O. Saremi and D. T. Son, JHEP 04, 091 (2012), arXiv:1103.4851 [hep-th].
- [256] A. Castro, S. Detournay, N. Iqbal, and E. Perlmutter, JHEP 07, 114 (2014), arXiv:1405.2792 [hep-th].
- [257] N. Iqbal and A. C. Wall, (2015), arXiv:1509.04325 [hep-th].
- [258] T. Nishioka and A. Yarom, (2015), arXiv:1509.04288 [hep-th].
- [259] T. Azeyanagi, R. Loganayagam, and G. S. Ng, (2015), arXiv:1507.02298 [hep-th].
- [260] A. Belin, A. Castro, and L.-Y. Hung, JHEP 11, 145 (2015), arXiv:1508.01201 [hep-th].
- [261] P. M. Bellan, *Fundamentals of plasma physics* (Cambridge University Press, 2008).
- [262] J. P. Freidberg, *Ideal MHD*: (Cambridge University Press, Cambridge, 2014).
- [263] S. Dubovsky, L. Hui, A. Nicolis, and D. T. Son, Phys. Rev. D85, 085029 (2012), arXiv:1107.0731 [hep-th].
- [264] S. Endlich, A. Nicolis, R. A. Porto, and J. Wang, Phys. Rev. D88, 105001 (2013), arXiv:1211.6461 [hep-th].
- [265] S. Grozdanov and J. Polonyi, Phys. Rev. D91, 105031 (2015), arXiv:1305.3670 [hep-th].
- [266] A. Nicolis, R. Penco, and R. A. Rosen, Phys. Rev. D89, 045002 (2014), arXiv:1307.0517 [hep-th].
- [267] P. Kovtun, G. D. Moore, and P. Romatschke, JHEP 07, 123 (2014), arXiv:1405.3967 [hep-ph].
- [268] M. Harder, P. Kovtun, and A. Ritz, JHEP **07**, 025 (2015), arXiv:1502.03076 [hep-th].

- [269] S. Grozdanov and J. Polonyi, Phys. Rev. D92, 065009 (2015), arXiv:1501.06620 [hep-th].
- [270] M. Crossley, P. Glorioso, and H. Liu, (2015), arXiv:1511.03646 [hep-th].
- [271] P. Glorioso, M. Crossley, and H. Liu, (2017), arXiv:1701.07817 [hep-th].
- [272] F. M. Haehl, R. Loganayagam, and M. Rangamani, JHEP 01, 184 (2016), arXiv:1510.02494 [hep-th].
- [273] F. M. Haehl, R. Loganayagam, and M. Rangamani, JHEP 04, 039 (2016), arXiv:1511.07809 [hep-th].
- [274] D. Montenegro and G. Torrieri, Phys. Rev. D94, 065042 (2016), arXiv:1604.05291 [hep-th].
- [275] P. Gao and H. Liu, (2017), arXiv:1701.07445 [hep-th].
- [276] K. Jensen, N. Pinzani-Fokeeva, and A. Yarom, (2017), arXiv:1701.07436 [hep-th].
- [277] D. Schubring, Phys. Rev. D91, 043518 (2015), arXiv:1412.3135 [hep-th].
- [278] J. Hernandez and P. Kovtun, (2017), arXiv:1703.08757 [hep-th].
- [279] X.-G. Huang, A. Sedrakian, and D. H. Rischke, Annals Phys. 326, 3075 (2011), arXiv:1108.0602 [astro-ph.HE].
- [280] R. Critelli, S. I. Finazzo, M. Zaniboni, and J. Noronha, Phys. Rev. D90, 066006 (2014), arXiv:1406.6019 [hep-th].
- [281] S. I. Finazzo, R. Critelli, R. Rougemont, and J. Noronha, Phys. Rev. D94, 054020 (2016), arXiv:1605.06061 [hep-ph].
- [282] P. Kovtun, JHEP 07, 028 (2016), arXiv:1606.01226 [hep-th].
- [283] D. Montenegro, L. Tinti, and G. Torrieri, (2017), arXiv:1701.08263 [hepth].

- [284] J. F. Fuini and L. G. Yaffe, JHEP **07**, 116 (2015), arXiv:1503.07148 [hep-th]
- [285] S. Weinberg, *The Quantum Theory of Fields. Vol. 1: Foundations* (Cambridge University Press, 2005).
- [286] M. E. Peskin and D. V. Schroeder, An Introduction to quantum field theory (1995).
- [287] D. Yamada and L. G. Yaffe, JHEP 09, 027 (2006), arXiv:hep-th/0602074 [hep-th].
- [288] A. Cherman, S. Grozdanov, and E. Hardy, JHEP **06**, 046 (2014), arXiv:1308.0335 [hep-th].
- [289] D. Z. Freedman, S. D. Mathur, A. Matusis, and L. Rastelli, Nucl. Phys.
 B546, 96 (1999), arXiv:hep-th/9804058 [hep-th].
- [290] D. Anselmi, J. Erlich, D. Z. Freedman, and A. A. Johansen, Phys. Rev. D57, 7570 (1998), arXiv:hep-th/9711035 [hep-th].
- [291] E. D'Hoker and P. Kraus, JHEP 10, 088 (2009), arXiv:0908.3875 [hep-th].
- [292] S. Janiszewski and M. Kaminski, Phys. Rev. D93, 025006 (2016), arXiv:1508.06993 [hep-th].
- [293] M. Ammon, M. Kaminski, R. Koirala, J. Leiber, and J. Wu, JHEP 04, 067 (2017), arXiv:1701.05565 [hep-th].
- [294] D. Marolf and S. F. Ross, JHEP 0611, 085 (2006), arXiv:hep-th/0606113 [hep-th].
- [295] N. Jokela, G. Lifschytz, and M. Lippert, JHEP 10, 014 (2013), arXiv:1307.6336 [hep-th].
- [296] E. Pomoni and L. Rastelli, JHEP 04, 020 (2009), arXiv:0805.2261 [hep-th]

- [297] T. Faulkner and N. Iqbal, JHEP 07, 060 (2013), arXiv:1207.4208 [hep-th].
- [298] S. S. Gubser, Phys. Rev. D63, 084017 (2001), arXiv:hep-th/9912001 [hep-th].
- [299] E. D'Hoker and P. Kraus, JHEP **03**, 095 (2010), arXiv:0911.4518 [hep-th].
- [300] S. S. Gubser, S. S. Pufu, and F. D. Rocha, JHEP **08**, 085 (2008), arXiv:0806.0407 [hep-th].
- [301] R. A. Davison, B. Goutéraux, and S. A. Hartnoll, JHEP 10, 112 (2015), arXiv:1507.07137 [hep-th].
- [302] M. Taylor, (2000), arXiv:hep-th/0002125 [hep-th].
- [303] J. M. Martín-García, "xAct: Efficient Tensor Computer Algebra, http: //www.xact.es/,".
- [304] S. A. Stricker, Eur. Phys. J. C74, 2727 (2014), arXiv:1307.2736 [hep-th].
- [305] S. Waeber, A. Schaefer, A. Vuorinen, and L. G. Yaffe, JHEP 11, 087 (2015), arXiv:1509.02983 [hep-th].
- [306] S. Grozdanov, N. Kaplis, and A. O. Starinets, JHEP 07, 151 (2016), arXiv:1605.02173 [hep-th].
- [307] S. Grozdanov and W. van der Schee, (2016), arXiv:1610.08976 [hep-th].
- [308] A. Lucas, JHEP 03, 071 (2015), arXiv:1501.05656 [hep-th].
- [309] D. T. Son and A. O. Starinets, JHEP 09, 042 (2002), arXiv:hep-th/0205051 [hep-th].
- [310] C. P. Herzog and D. T. Son, JHEP 03, 046 (2003), arXiv:hep-th/0212072 [hep-th].
- [311] V. E. Fortov and V. B. Mintsev, Phys. Rev. Lett. **111**, 125004 (2013).
- [312] L. P. Kadanoff and P. C. Martin, Annals of Physics 24, 419 (1963).

- [313] M. Edalati, J. I. Jottar, and R. G. Leigh, JHEP 10, 058 (2010), arXiv:1005.4075 [hep-th].
- [314] M. Blake, Phys. Rev. Lett. 117, 091601 (2016), arXiv:1603.08510 [hep-th].
- [315] A. Lucas and J. Steinberg, JHEP 10, 143 (2016), arXiv:1608.03286 [hep-th]
- [316] Y. Ling, Z. Xian, and Z. Zhou, Chin. Phys. C41, 023104 (2017), arXiv:1610.08823 [hep-th].
- [317] S.-F. Wu, B. Wang, X.-H. Ge, and Y. Tian, (2017), arXiv:1702.08803 [hep-th].
- [318] M. Blake, R. A. Davison, and S. Sachdev, (2017), arXiv:1705.07896 [hep-th].
- [319] T. Hartman, S. A. Hartnoll, and R. Mahajan, (2017), arXiv:1706.00019 [hep-th].
- [320] Q. Li, D. E. Kharzeev, C. Zhang, Y. Huang, I. Pletikosic, et al., (2014), arXiv:1412.6543 [cond-mat.str-el].
- [321] J. Gooth et al., (2017), arXiv:1703.10682 [cond-mat.mtrl-sci].
- [322] A. Hamilton, D. N. Kabat, G. Lifschytz, and D. A. Lowe, Phys. Rev. D74, 066009 (2006), arXiv:hep-th/0606141 [hep-th].
- [323] V. Balasubramanian, B. D. Chowdhury, B. Czech, J. de Boer, and M. P. Heller, Phys. Rev. D89, 086004 (2014), arXiv:1310.4204 [hep-th].
- [324] B. Czech, X. Dong, and J. Sully, JHEP 11, 015 (2014), arXiv:1406.4889 [hep-th].
- [325] E. Hijano, P. Kraus, E. Perlmutter, and R. Snively, JHEP 01, 146 (2016), arXiv:1508.00501 [hep-th].
- [326] F. Pastawski, B. Yoshida, D. Harlow, and J. Preskill, JHEP 06, 149 (2015), arXiv:1503.06237 [hep-th].

- [327] S. W. McIntosh, B. De Pontieu, M. Carlsson, V. Hansteen, P. Boerner, and M. Goossens, Nature 475, 477 (2011).
- [328] P. Cargill and I. De Moortel, Nature 475, 463 (2011).
- [329] J. Zinn-Justin, *Quantum Field Theory and Critical Phenomena*, International series of monographs on physics (Clarendon, 2002).
- [330] F. M. Haehl, R. Loganayagam, and M. Rangamani, (2017), arXiv:1701.07896 [hep-th].

Curriculum Vitæ

Nick Poovuttikul was born in Bangkok, Thailand on May 1990. He stayed in Bangkok until he graduated from high school in 2008. During that time he participated in a few physics competitions and got a scholarship from the Thai government to study physics. At Imperial College London, he obtained a bachelor degree in Physics with Theoretical Physics and a master degree in Quantum Fields and Fundamental Forces. He worked on a condensed matter problem for his final year project with Dr. Derek K.K. Lee and and wrote a literature review on fermions in gauge/gravity duality for his master degree dissertation under the guidance of Prof. Dr. Toby Wiseman. While in London, he was a bassist in a Thai pop rock band at Thai Square Putney; a member of the editorial board for an annual journal of "Samaggi Samagom", the Thai association in the UK, from 2011-2012; and Vice president of Imperial College's Thai society in 2011. He also did a few volunteering jobs for education in Thailand during the summer breaks.

He moved to Leiden, the Netherlands, to start his PhD under the supervision of Prof. Dr. Jan Zaanen in autumn 2013. He publised his work in peerreviewed journals, volunteered to organise a holography journal club while grading homeworks for master students attending the theory of general relativity course at Leiden. He tried to make friends and promote his work by giving talks and poster presentations and attending several schools, meetings, workshops and conferences in the Netherlands, Brazil, Belgium, Germany, Portugal, Denmark, Spain, Greece, England, Sweden and Thailand. After his PhD, he will be a postdoctoral researcher in the high energy theoretical physics group at the University of Iceland. Eventually, bound by his so-called "scholarship" contract, he will go back to do research in one of the universities in Thailand.

List of Publications

- A. Parnachev, N. Poovuttikul, "Topological entanglement entropy, ground state degeneracy and holography," JHEP **1510** (2015) 092 [arXiv:1504.08244 [hep-th]].
- P. Burikham, N. Poovuttikul, "Shear viscosity in holography and effective theory of transport without translational symmetry," Phys.Rev. D 94 (2016) 106001 [arXiv:1601.04624 [hep-th]].
- S. Grozdanov, N. Poovuttikul, "Universality of anomalous conductivities in theories with higher-derivative holographic duals," JHEP 1609 (2016) 046 [arXiv:1603.08770 [hep-th]].
- 4. S. Grozdanov, N. Poovuttikul, "Magnetohydrodynamic waves in a strongly coupled holographic plasma," [arXiv:1707.04182 [hep-th]].
- 5. T. Andrade, M. Baggioli, A. Krikun, N. Poovuttikul, "Pinning of longitudinal phonons in holographic helical crystals," [arXiv:1708.08306 [hep-th]].

Acknowledgement

First of all, I would like to thank Jan Zaanen for having accepted me into the quantum matter group at Leiden, for his honest and supportive supervision, for providing the stimulating working environment. I am particularly thankful for the freedom he provided, which allowed me to find the topics I am interested in, and for constantly supporting my independent work. I would also like to thank Koenraad Schalm for the time we spent to discuss many papers, ideas and his insightful comments on all work I have been doing in the past few years. My PhD and the way I think about physics would be very different without Andrei Parnachev and Sašo Grozdanov. I would like to wholeheartedly thank them for their guidance, for suggesting interesting and well-defined physics problems and for being an inspiration throughout my PhD.

My journey into the world of hydrodynamics would never have happened without Richard Davison, Andrea Amoretti, Nikos Kaplis and, especially, Sašo Grozdanov. I am deeply grateful for everything I have learned from all of you. I would like to give a special thanks to Sašo Grozdanov for all enjoyable collaborations and many inspirational ideas during my PhD. It is very much fun to work with you and I hope we will be able to continue working together on many other interesting things.

Furthermore, I would like to thanks all my collaborators whom I haven't mentioned yet: Misha Goykhman, Ann-Kathrin Straub, Steffen Klug, Matteo Baggioli, Sasha Krikun, Tomas Andrade, Juan Pedraza, Bartek Benenowski and Piyabut Burikham for letting me collaborate on many interesting projects and for all they taught me. I am particularly grateful to Steffen, Nikos and Sasha for their patience and for helping me improving my numerical skills.

I would like to thank my flatmates, Miggy and Jaakko, as well as my officemates, Marco, Misha and Petter for the pleasant time and many intriguing discussions about physics, politics and whatnot. My years at Leiden would be very dull without friends and fellow member of the quantum matter group: Miggy, Petter, Jaakko, Louk, Robert-Jan, Misha, Andrea, Richard, Steffen, Nikos, Sašo, Balasz, Andrey, Ke, Bartek, Sasha, Vincenzo, Aurelio, Philippe, Simon and Koenraad. For all the enjoyable discussions and time we spent together, I thank you all.

I would like to thank Simon, Philippe and Miggy for proof reading and for their comments on this thesis and Hiddo Algera for translating the samenvatting. I am also grateful to the secretaries, Fran, Marianne and Trudy, for their help in many aspects.

Lastly, I would like to thank my mother, sister and Toon for their love and unconditional support. Having all three of you as a family is the best thing that could happen in my life.

Stellingen

behorende bij het proefschrift Transport coefficients and low energy excitations of a strongly interacting holographic fluid

1. In the situation where additional degrees of freedom are added to a hydrodynamical system, the shear viscosity is defined through the constitutive relation, not the Kubo formula derived for a relativisic fluid.

Chapter 3

2. Before declaring results from holography to be "universal", one has to make sure that these do not rely on the fine-tuning of the equation of state or the coupling constants.

Chapter 4

3. Magnetohydrodynamics is a theory of dissipative strings, not of charged particles and photons.

Chapter 5

 Gauge/gravity duality can be useful for studying macroscopic properties but is useless for extracting microscopic information unless one is interested in supersymmetric quantum field theories.

Chapter 1

5. Despite its widespread usage, the ideal fluid is unstable: it is highly sensitive to initial conditions. For example, in the simplest model of the Earth's atmosphere, the deviations grow by a factor of 10^5 in 2 months.

V.I. Arnold & B.A. Khesin, Topological Methods in Hydrodynamics, Springer 1998
6. Hydrodynamical fluctuations, such as long-time tails, cannot be observed in a field theory with a classical gravity dual. Instead, they are intriguingly connected to quantum gravity corrections. This connection has largely been ignored in the literature.

> e.g. P. Kovtun & L. Yaffe, Phys.Rev. D 68 02 (2003) [arxiv:hep-th/0303010] and S. Caron-Hout & O. Saremi, JHEP 11 013 (2010) [arxiv:0909.4525]

7. From the condensed matter point of view, one of the most interesting aspects of entanglement entropy is topological entanglement entropy. In the holographic literature, a topological term is often mentioned in the introductions as a motivation to study the entanglement entropy but has almost never been the main focus.

A. Parnachev & A. Pakman JHEP **07** 097 (2008) [arxiv:0805.1891] and A. Parnachev & N. Poovuttikul, JHEP **10** 092 (2015) [arxiv:1504.08244]

8. A popular toy model for weak momentum relaxation, where only the conservation of momentum is modified (taking the form $\partial_{\mu}T^{\mu i} = -\Gamma T^{ti}$), cannot reliably capture dissipative properties. This is because it ignores, by construction, the correction from momentum relaxation to the constitutive relation.

S.Hartnoll, P.Kovtun, M.Muller & S.Sachdev, Phys.Rev. B **76** 144502 (2007) [arXiv:0901.0924]

9. Despite being trained to be logical, scientists are still highly susceptible to arguments from authority.

Nick Poovuttikul Leiden, 16 November 2017